# University of Strathclyde Department of Electronic and Electrical Engineering

Grid Integration of Renewable Power Generation

by

Syahrul Ashikin Azmi B. Eng., M.Sc.

A thesis presented in fulfilment of the requirements for the degree of Doctor of Philosophy

March 2014

This thesis is the result of the author's original research. It has been composed by the author and has not been previously submitted for examination which has led to the award of degree.

The copyright of this thesis belongs to the author under the terms of the United Kingdom Copyright Acts as qualified by University of Strathclyde Regulation 3.50. Due acknowledgement must always be made of the use of any material contained in, or derived from, this thesis.

Signed: **8. Azmi** 

Date: 23 March 2014

#### Acknowledgements

I would like to express my deep gratefulness to my supervisors Prof. Barry W. Williams, Prof. Stephen J. Finney and Dr. Grain P. Adam for their extensive guidance, continuous support, kind advice, helpful comments and capable supervision which helped a lot for this work to be in its present form.

I also would like to acknowledge staff and colleagues in the Department of Electronic and Electrical Engineering in University of Strathclyde for their assistance and support during the years of research.

A gracious thankful is made to my sponsors, Ministry of Higher Education Malaysia and Universiti Malaysia Perlis.

Finally, a heartfelt gratitude goes to my beloved parents, my husband and my children for all the sacrifices they had to make to see this through to the end and for their prayers and encouragement.

Syahrul Ashikin Azmi

November 2013

#### Abstract

This thesis considers the use of three-phase voltage and current source inverters as interfacing units for renewable power, specifically photovoltaic (PV) into the ac grid. This thesis presented two modulation strategies that offer the possibility of operating PV inverters in grid and islanding modes, with reduced switching losses. The first modulation strategy is for the voltage source inverter (VSI), and exploits 3<sup>rd</sup> harmonic injection with selective harmonic elimination (SHE) to improve performance at low and high modulation indices, where the traditional SHE implementation experiences difficulties due to pulse dropping. The simulations and experimentation presented show that the proposed SHE allows grid PV inverters to be operated with less than a 1kHz effective switching frequency per device. This is vital in power generation, especially in medium and high power applications. Pulse dropping is avoided as the proposed modified SHE spreads the switching angles over 90°, in addition increasing the modulation index.

The second proposed modulation strategy, called direct regular sampled pulse width modulation (DRSPWM), is for the current source inverter (CSI). It exploits a combination of forced and natural commutation imposed by the co-existence of an insulated gate bipolar transistor in series with a diode in a three phase current source inverter, to determine device dwell times and switching sequence selection. The DRSPWM strategy reduces switching frequency per device in a CSI by suspending each phase for 60°, similar to VSI dead-band, thus low switching losses are expected. Other benefits include simple digital platform implementation and more flexible switching sequence selection and pulse placement than with space vector modulation. The validity of the DRSPWM is confirmed using simulations and experimentation.

This thesis also presents a new dc current offset compensation technique used to facilitate islanding or grid operation of inverter based distributed generation, with a reduced number of interfacing transformers. The proposed technique will enable transformerless operation of all inverters within the solar farm, and uses only one power transformer at the point of common coupling. The validity of the presented modulation

strategies and dc current offset compensation technique are substantiated using simulations and experimentation.

# List of Symbols

$C_{f}$ , $C_{F}$	Filter capacitance (F)
<i>f</i> <sub>cr</sub>	Carrier frequency (Hz)
$f_{sw}$	Switching frequency (Hz)
I <sub>dc</sub>	dc-link current (A)
$I_g$	Grid current vector (A)
$I_m$	Peak phase current (A)
$I_{pv}$	Photovoltaic output current (A)
$i_{ca}$ , $i_{cb}$ , and $i_{cc}$	Instantaneous capacitor currents (A)
$i_{ia}$ , $i_{ib}$ , and $i_{ic}$	Instantaneous inverter currents (A)
$i_{oa}$ , $i_{ob}$ , and $i_{oc}$	Instantaneous output currents (A)
k <sub>p</sub> , k <sub>i</sub>	Proportional and integral control parameters
$L_f$ , $L_F$	Filter inductance (H)
$m, m_a$	Modulation index
Ν	Transformer turns ratio
Р	Active power (W)
Q	Reactive power (VAr)
$R_f, R_F$	Filter resistor ( $\Omega$ )
$T_s$	Switching period (s)
$V_{cr}, \hat{V}_{cr}$	High frequency carrier voltage (V)
V <sub>dc</sub>	dc-link voltage (V)
$V_g$	Grid voltage vector (V)
$V_m, \hat{V}_m$	Peak fundamental voltage (V)
$V_{pv}$	Photovoltaic output voltage (V)
$v_{ca}$ , $v_{cb}$ , and $v_{cc}$	Instantaneous capacitor voltages (V)
V <sub>cr</sub>	High frequency carrier signal (V)
$v_{ma}$ , $v_{mb}$ , and $v_{mc}$	Modulating signals (V)
$v_{oa}$ , $v_{ob}$ , and $v_{oc}$	Instantaneous load voltages (V)
$\delta$	Load angle (rad)
$\phi$	Phase angle (rad)

# List of Abbreviations

CSI	Current source inverter
DG	Distributed generation
DRSPWM	Direct regular sampled pulse width modulation
IEC	International Electrotechnical Commission
IEEE	Institute of Electrical and Electronic Engineers
IGBT	Insulated gate bipolar transistor
MPPT	Maximum power point tracking
NPC	Neutral point clamping
PCC	Point of common coupling
PI	Proportional integral
PLL	Phase locked loop
PWM	Pulse width modulation
PR	Proportional resonance
PV	Photovoltaic
SHE	Selective harmonic elimination
SPWM	Sinusoidal pulse width modulation
STATCOM	Static synchronous compensator
SVM	Space vector modulation
THD	Total harmonic distortion
THIPWM	Third harmonic injection pulse width modulation
VSI	Voltage source inverter

# Table of Contents

Ackn	owledgement	i
Abstr	ract	ii
List o	f Symbols	iv
List o	f Abbreviation	v
Chap	ter One: Introduction	
1	Background and development of renewable energy systems	1
1.1	Development of renewable energy systems in the UK	2
1.2	Application of distributed generation based renewable energy sources	3
1.3	Motivation	4
1.4	Objectives	6
1.5	Methodology and scope of study	7
1.6	Thesis organization	7
Ref	Terences (Control of the control of	9
Chap	ter Two: Voltage Source Inverter Topologies	
2.	Introduction	11
2.1	Background	11
2.2	DC-AC power inverter topologies	13
2	2.2.1 Single-phase half-bridge inverter	13
2	2.2.2 Single-phase full-bridge inverter	14
2.3	Three-phase two-level voltage source inverter	15

2.4 Multilevel inverter		17
2.4.1	Neutral-point clamped inverter	17
2.4.2	Capacitor-clamped multilevel inverter	18
2.4.3	Cascaded multilevel inverter	20
2.4.4	Modular multilevel	21
2.5 Summary		24
References		24

### Chapter Three: Modulation Technique for the Voltage Source Inverter

	3	Introduction	27
	3.1	Background	27
	3.2	Carrier-based Sinusoidal Pulse Width Modulation	28
	3.3	Third Harmonic Injection Pulse Width Modulation (THIPWM)	31
	3.4	Space vector modulation (SVM)	32
	3.5	Selective harmonic elimination	34
	3.	5.1 Improved selective harmonic elimination	38
	3.6	Assessment of PWM modulation techniques	45
	3.7	Summary of major findings	49
Re	fere	nces	50

### **Chapter Four: Grid Integration of Voltage Source Inverter**

4	Introduction	53
4.1	Background	54
4.2	Operation of the VSI in island and grid modes	55
	4.2.1 Control system	56
4.3	Analysis of AC dynamic equations for the three-phase two-level voltage s	source

inverter in island mode584.3.1Simulation results of a three-phase VSI using PR control and PI control for

island mode

	4.3.2 Experimental verification of three-phase VSI using PR control and	PI
	control for island mode	66
4.4	Analysis of ac-side dynamic equations of grid connected VS proportional-resonance control in <i>abc</i> coordinates	SI using 68
4.5	Analysis of AC side dynamic equation of a grid-connected VS proportional-integral control in the $dq$ -frame	SI using 70
4.6	Simulation results for a VSI in a medium voltage photovoltaic ap	plication 78
4.7	Experimental results of a low-loss grid-connected VSI	88
4.8	Typical application of VSI for grid connection of PV arrays	90
4.9	Summary	93
Refe	rences	93

63

### Chapter Five: Current Source Inverter for Photovoltaic Application

5. Ir	ntroduction	98
5.1 Ba	ackground	98
5.2 Cu	urrent source inverter topologies	99
5.2.1	Single-phase H-bridge current source inverter	99
5.2.2	2 Three-phase current source inverter	100
5.2.3	3 Multilevel current source inverter	101
5.3 M	odulation strategies for current source inverter	102
5.3.1	Carrier-based sinusoidal pulse width modulation	102
5.3.2	2 Selective harmonic elimination	105
5.4 D	irect CSI space vector modulation	106
5.4.1	VSI-space vector modulation with logic mapping circuitry	110
5.5 Tv	welve-Step Direct Pulse Width Modulation (TS-PWM)	111
5.6 Pr	coposed direct-regular sampled pulse width modulation	112
5.6.1	Theoretical basis of the proposed DRSPWM	117

5.6.2	Open-loop validation of the proposed DRSPM	119
5.6.3	Simulation results	121
5.6.4	Experimental verification of the proposed DRSPWM	125
5.7 Ass	essment of PWM modulation techniques	127
5.8 Summary		129
References		129

### Chapter Six: DC Current Offset Compensation Technique for Grid Connected Inverters

6. In	troduction	134
6.1 Ba	ckground	134
6.2 Re	view of dc current compensation techniques	136
6.2.1	Passive methods	136
6.2.2	Active methods	138
6.3	Proposed dc current offset compensation technique for grid-converters	nnected 141
6.4 Sir	nulation result	147
6.4.1	Configuration I: Combination of VSIs	148
6.4.2	Configuration II: VSI and CSI combination	151
6.5 Ex	perimental verification	152
6.5.1	Experimental results: inverters 1 and 2 without dc offset injection	153
6.5.2	Experimental results: the proposed dc current compensation technique	ue with
	different levels of dc current injected into the output of inverter 1	155
6.6 Su	mmary	161
Reference	S	161

Chapter Seven: Grid Integration of a Current Source Inverter

7	Intr	oduction	164
7.1	Bac	kground	164
7.2	Con	trol system in island and grid modes	166
	7.2.1	Voltage control in island operation	167
	7.2.2	Simulation and experimental operational verification of the CSI in is	land
		mode	171
	7.3	Current control system for a grid-connected CSI	176
	7.3.1	Power control loop	177
	7.3.2	Current control loop	183
	7.4	Simulation results	183
	7.4.1	Unity power factor application	183
	7.4.2	Lagging power factor	185
	7.4.3	Leading power factor	186
	7.4.4	Zero power factor	187
7.	.5 Exp	perimental results of the proposed DRSPWM in a grid connected CSI	188
	7.5.1	Unity power factor application	190
	7.5.2	Lagging power factor	192
	7.5.3	Leading power factor	193
	7.5.4	Zero power factor	194
	7.5.5	Black-start capability with the proposed DRSPWM	195
7.	.6 Тур	ical application of a CSI for grid connection of PV arrays	197
7.7		immary	199
Refe	erences		199

# **Chapter Eight: Conclusion**

8	General conclusion	202
8.1	Author's contributions	205
8.2	Recommendations for future research	206

# Appendices

Appendix A	Real-timedSPACE1006 processor and I/O boards	207
Appendix B	Arbitrary custom pulse patterns code generation in dSPACE5101	210
Appendix C	List of Figures and Tables	215
Appendix D	Summary of relevant published work by the author	223

### **Chapter One**

### Introduction

#### 1 Background and development of renewable energy systems

In the last two decades, environmental organizations have encouraged the power industry to use renewable power such as wind, photovoltaic, and tidal instead of fossil fuels, as a means to combat climate change. After years of research and development of hydroelectric, wind and photovoltaic (PV) energy sources, the economical and technical feasibility of these forms of power has extended to a large-scale level, but they remain expensive compared to fossil fuels[1.1-4]. Rethinking of the whole power industry approach to renewable power is on-going.

Renewable energy production rose from 4225.2TWh to 4447.5TWh between 2010 and 2011, representing a 20.2% share of the total global electricity production. Table 1.1 lists the global fraction of electricity production from various renewable sources in 2011. China ranks as the highest renewable energy producer with 785.9TWh, followed by the United States (555.6TWh), Brazil (450.8TWh), Canada (403TWh), and Russia (166.7TWh). The United Kingdom is placed 19<sup>th</sup> among the global producers of renewable energy production, with 37.6TWh [1.5, 6].

Source	Electricity production (TWh)	Percentage relative to the total production (%)
Hydraulic	3579.5	80.5%
Wind	459.9	10.3%
Biomass	276.0	6.2%
Geothermal	69.9	1.6%
Solar	61.6	1.4%
Marine energies	0.5	0.01%
Total	4447.5	100%

Table 1.1: Total electricity production from renewable sources in 2011.

The use of renewable energy resources, rather than fossil fuel and other non-renewable energy sources, led to positive consequences and impact on the climate and environment. These include low greenhouse gas emissions and polluting gases [1.7]. Among these renewable power sources, wind farms have raised several social and technical concerns. Examples of social concerns are the generated noise from the wind turbines and an unpleasant scenery, such that these concerns complicate the process of obtaining permission to build new onshore wind farms[1.8, 9]. Unlike wind energy, a photovoltaic system is a static device and does not produce noise or pollution; therefore it is suitable and safe for use in residential areas or large-scale onshore PV farms.

#### 1.1 Development of renewable energy systems in the UK

Due to geographical location and unpredictable weather conditions, it is expected that photovoltaic systems are not suitable for the UK. However, with the advancement of the photovoltaic system, they can operate without the need of large amounts of sunlight. In 2007, the UK government agreed with the EU's target to increase the Union's portion of the total energy consumption from renewable energy sources to 15% by 2020. The Government's effort continues by introducing a system called Feed-In Tariff in April 2010 to support the usage of photovoltaic (roof-mounted or standalone), wind turbine (building mounted or free standing), hydroelectric, anaerobic digester, and micro combined heat and power (CHP). The system supports householders and communities in generating their own power with regular payment through their energy supplier. According to the Feed-In Tariff, there are approximately 174,988 photovoltaic sites in England with cumulative installed capacity of 573.3MW, 19,138 sites in Wales (46.3MW), 12,808 sites in Scotland (41.2MW), and 204 sites in Northern Ireland (0.7MW). As for onshore wind farms (dated construction finished by April 2013), Scotland holds the largest installed capacity of 1,845.9MW, followed by England (124.8MW), Wales (58.5MW) and North Ireland (54MW). Whilst, England is leading for operational offshore wind farms with a capacity of 2,969.2MW, this is followed by Scotland with 190MW and Wales with 150MW. The world's largest offshore wind farm is the London Array, located off-coast at Kent, England with a capacity of 630MW[1.10-12].

#### **1.2** Application of distributed generation based renewable energy sources

Distributed generation is a form of decentralised electricity generation that permits a large number of medium-scale generating units, with a high concentration of inverter connected renewable energy based generation. This generation type is expected to be connected to the distribution network rather than a high voltage transmission network, or can be connected to form a so called micro-grid. The development of distributed generation based renewable energy systems has an important role in long term environmental targets. The main aims of such systems are to increase the network connection capacity by allowing more consumers and producer customers connections without creating new reinforcement costs, to enhance the reliability of the system, and to improve the overall supply quality with modern voltage control [1.13].

There are two major application modes that can be beneficial when using distributed generation system based renewable energy sources: island and grid connected modes. In the island mode of operation, the unit operates as an independent voltage source, without connection to the utility grid. Since this system is the source of power, it must be able to specify and control the voltage and frequency of its output. The dedicated control system in this mode must provide sufficient active and reactive power to match the connected loads whilst delivering output voltage with high power quality. The island mode based distributed generation isolates the user from the grid either by choice or circumstance, as in remote applications. The units are parallel connected to supply all the local loads. Power converters are designed to share the local loads and also to be controlled from a remote supervisory controller that can set the load contribution of each unit [1.14].

The grid is an integrated network of generation, high-voltage transmission, substations, and local distribution. Grid connected renewable energy systems (wind and photovoltaic) are designed to operate in parallel with the utility grid. The features of a grid connected mode include active and reactive power control with sufficient output current quality, good parallel load sharing operation, and establishment of frequency and voltage levels [1.15]. As the power generated by medium-scale renewable based

distributed generation is much smaller than that of the utility grid, there is no need to worry about the power balance.

#### 1.3 Motivation

One of the main challenges in a distributed generation system is to connect hundreds or thousands of renewable energy systems to the utility/grid network. If the distributed energy sources are not properly connected, the grid can become unstable or even fail. This challenge is largely handled by power electronic converters.

Power electronic converters play an important role in distributed generation and in the integration of renewable energy sources into the electrical grid. They are widely used and rapidly expanding as these applications become more integrated with the grid-based systems. The main tasks of the power electronic converters are [1.16-18]:

- Maximum power transfer: on the input side (source), power electronics permit control and tracking of pre-defined power characteristic curves to maximize power extraction from the resources. Also, in the event of faults and sudden load variations, power electronics help to adjust the power extraction level and protects the sources.
- Active and reactive power control and power quality control: on the grid side, power converters must control active and reactive power injection into the grid, ensuring sinusoidal current is injected into grid, with low harmonic content, low electromagnetic interference, low leakage dc currents that meets grid codes and standard.

Although the power electronic converter is an efficient unit in the integration of distributed generation based renewable energy sources, it is the most susceptible unit to failure which may lead to major breakdown of a power plant. A statistic for wind power system failures, based on Swedish wind power plant in 2000 to 2004, shows that nearly 50% of the failures are caused by the power electronic converter[1.19]. Similarly in photovoltaic systems, for example, the Florida Solar Energy Center has recorded 213 failure events on 130 grid-connected PV systems from 1999 to 2003, where 65% of those failures are contributed by inverters[1.20, 21]. Thus, the issue of reliable and

efficient inverter design and inverter control systems are the main concerns for manufacturers, researchers and users when operating renewable energy systems.

Another issue related to massive penetration of distributed generation based renewable energy sources is the impact on grid stability [1.22, 23]. This issue has caused grid operators in many countries to upgrade their connection codes to ensure the safety and reliability of the network. In most countries, local regulations are imposed by the utility or the government. However, many organizations are working towards imposing grid requirement standards that can be adapted by most countries. The organizations are The Institute of Electrical and Electronic Engineers (IEEE) and International Electrotechnical Commission (IEC). Examples of IEEE and IEC standards, used as the main reference in this thesis, are IEEE-1547and IEC-61727. The IEEE-1547 standard deals with interconnecting distributed generation to the electric power system and the IEC-61727 standard is for Photovoltaic Systems-Characteristics of the Utility Interface [1.24, 25]. Table 1.2summarises some of the important issues listed in both the IEEE-1547 and IEC-61727 standards.

Issues	IEEE1574	IEC61727
Nominal power	30 kW	10 kW
Harmonic current (order-h) limits		(3-9) 4.0% (11-15) 2.0% (17-21) 1.5% (23-33) 0.6% ese ranges shall be less harmonic limits listed.
Maximum current THD	5.	0%
DC current injection	Less than 0.5% of rated output current	Less than 1.0% of rated output current
Frequency range for normal operation	59.3 Hz to 60.5 Hz	50±1 Hz

Table 1.2:Summary of some important issues in the IEEE1574 and IEC61727 standards.

Grid connected inverters have to meet strict codes imposed by IEEE and IEC standards [1.26]. For example in Table 1.2, the current harmonic injection into the grid must not exceed 5%, with limitations imposed on individual harmonics as a percentage of the fundamental component. One widely used strategy to control the quality of injected output current into the grid, is the pulse width modulation technique. The modulation

technique is at the heart of the majority power electronic inverters with the target to reduce harmonic distortion and increased output magnitudes for a given switching frequency. Other performance criteria include minimization of switching losses and simple implementation, which are some of the challenges in selecting a suitable modulation technique[1.27].

DC current injection issues in grid connected inverters have been largely overlooked, despite growing concerns regarding the effect of dc current injection on distribution system components. The most severe problems that can be caused are saturation of the distribution transformers which results in waveform distortion, excessive losses, overheating and reduced lifespan [1.28]. The IEEE and IEC standards have been imposed to minimize the impact of dc current injection into the ac grid. The allowable limits range from 0.5% and 1% of the rated output current, as listed in Table 1.2 for low-voltage applications. However, these regulations and standards do not guarantee that the accumulative effect of numerous power converters connected to the grid remains acceptable, with no harmful effects to electric systems sensitive to dc components. Various methods have been proposed to prevent or minimize dc current injection into the ac grid, and have attract considerable interest and attention.

#### 1.4 Objectives

In this thesis, three-phase voltage and current source inverters for grid integration are discussed, focusing on the following aspects: modulation strategies with the aim to ensure low inverter-semiconductor loss and guarantee acceptable power quality of the output voltage and current waveforms in accordance with the grid codes. In attempting to achieve these objectives, a number of new modulation strategies for low-loss, medium-voltage, three-phase current and voltage source inverters are proposed. These modulation strategies shown promising results, and other benefits such as simple implementation using digital platforms. Additionally, this thesis presents a new dc current offset compensation technique that may facilitate islanding or grid mode operation of distributed generation, with a reduced number of interfacing transformers. The proposed technique may have significant implication in the overall cost of medium-scale solar farms.

#### 1.5 Methodology and scope of study

This thesis uses simulation and experimentation as the two main tools to assess and validate the technical viability of each modulation strategy and dc compensation techniques presented. The theoretical basis of each proposed strategy or technique is provided and supported with quantitative and qualitative analysis. Despite control not forming the main core of this thesis, significant effort has been devoted to the presentation of the control systems needed to facilitate safe operation of the simulation and experimental setups. Systematic derivation of the control structures and designs is presented appropriately to assist readers to replicate the material presented in this thesis. The level of details presented on the core power electronic device related topics such as modulation and control designs are kept within the appropriate level for a wide range of readers.

#### 1.6 Thesis organization

The thesis is organized into eight chapters: Chapter One provides the general background to the drivers behind distributed generation in the context of grid and island modes, and highlights the challenges in grid interfacing of renewable power in generation, including PV systems. This chapter also presents some of the motivation, scope and methodology of this thesis.

Chapter Two provides a brief review and discussion of different voltage source inverter topologies, and articulates their suitability for PV systems.

As part of the contribution of thesis in the exploitation of converter modulation to reduce power loss and simplify digital implementation, Chapter Three discusses a number of existing pulse width modulation (PWM) techniques in general, with emphasis on the suitability of techniques for power generation in a harsh grid mode.

Chapter Four presents detailed discussion of the grid connected voltage source inverter with the implementation of existing modulation schemes and the proposed improved selective harmonic elimination (SHE) that exploits 3<sup>rd</sup> harmonic elimination. Using

simulation and experimentation, this chapter demonstrates that the three-phase voltage source inverter can be operated in the grid mode when the effective switching frequency is less than 1kHz, without compromising control of the active and reactive powers.

Chapter Five discusses possible current source inverter topologies that can be used as a grid interfacing unit. A new modulation strategy, called direct regular-sampled pulse width modulation (DRSPWM), is proposed that offers distinctive features compared to existing modulation strategies. A thorough comparison of the proposed modulation strategy and existing modulation strategies for the three-phase current source inverter are presented. Simulation and experimentation are used to substantiate the outcomes of the comparisons.

Chapter Six discusses several methods claimed to be appropriate to compensate dc current offset injected by grid connected inverters. This thesis presents a dc current offset injection compensation technique that exploits a simple resonance controller, with possible incorporation of a proportional-integral controller to reduce capital cost of medium-scale solar farms. The proposed technique permits transformerless inverter operation within the solar farm, and uses a single transformer at the point of common coupling to the grid.

Chapter Seven demonstrates the viability of the proposed DRSPWM discussed in chapter five, when a current source inverter is operated in grid and island modes. This chapter also presents possible control structures that can be used to control a current source inverter in the grid mode. It is shown that the current source inverter has the potential to match the voltage source inverter over the entire operating range, including lagging, leading, and zero power factor.

Chapter Eight draws general conclusions, author contributions and some recommendations for future research.

#### References

- [1.1] J. P. Benner and L. Kazmerski, "Photovoltaics gaining greater visibility," *Spectrum, IEEE*, vol. 36, pp. 34-42, 1999.
- [1.2] C. Zhe, J. M. Guerrero, and F. Blaabjerg, "A Review of the State of the Art of Power Electronics for Wind Turbines," *Power Electronics, IEEE Transactions on*, vol. 24, pp. 1859-1875, 2009.
- [1.3] Y. Bo, L. Wuhua, Z. Yi, and H. Xiangning, "Design and Analysis of a Grid-Connected Photovoltaic Power System," *Power Electronics, IEEE Transactions* on, vol. 25, pp. 992-1000, 2010.
- [1.4] E. Babaei and J. Ebrahimi, "A New Topology of Cascaded Multilevel Converters with Reduced Number of Components for High-Voltage Applications," *Power Electronics, IEEE Transactions on*, vol. PP, pp. 1-1, 2011.
- [1.5] IEA International Energy Agency. (2011, 16 September 2011). *Preliminary Trend Report 2010.*
- [1.6] K. Ardani and D. Feldman, "National Survey Report of PV Power Applications in the United States 2011," 2011.
- [1.7] S. B. Kjaer, J. K. Pedersen, and F. Blaabjerg, "A review of single-phase gridconnected inverters for photovoltaic modules," *Industry Applications, IEEE Transactions on*, vol. 41, pp. 1292-1306, 2005.
- [1.8] S. Krohn and S.Damborg, "On public attitudes towards wind power," *Renewable Energy*, vol. 16, pp. 954-960, 1999.
- [1.9] S. Iniyan, G.M.J. Herbert, E.Sreevalsan and S.Rajapandian, "A review of wind power technologies," *Renewable and Sustainable Energy Reviews*, vol. 11, pp. 1117-1145, 2007.
- [1.10] Department of Energy and Climate Change. UK. (2011). Renewable electricity in Scotland, Wales, Northern Ireland and the regions of England in 2011 (September 2012 edition of Energy Trends ed.). Available:https://www.gov.uk/government/publications/energy-trendsseptember-2012-special-feature-articles
- [1.11] The Microgeneration Certification Scheme. (2011). Available: <u>http://www.microgenerationcertification.org/</u>
- [1.12] M. I. Blanco, "The economics of wind energy," *Renewable and Sustainable Energy Reviews*, vol. 13, pp. 1372-1382, 2009.
- [1.13] S. R. Bull, "Renewable energy today and tomorrow," *Proceedings of the IEEE*, vol. 89, pp. 1216-1226, 2001.
- [1.14] "Distributed Generation: Securing America's Future with Reliable and Flexible Power," October 1999.
- [1.15] A. Bertani, C. Bossi, F. Fornari, S. Massucco, S. Spelta, and F. Tivegna, "A microturbine generation system for grid connected and islanding operation," in *Power Systems Conference and Exposition, 2004. IEEE PES*, 2004, pp. 360-365 vol.1., 2004.
- [1.16] M. Liserre, T. Sauter, and J. Y. Hung, "Future Energy Systems: Integrating Renewable Energy Sources into the Smart Power Grid Through Industrial Electronics," *Industrial Electronics Magazine, IEEE*, vol. 4, pp. 18-37, 2010.

- [1.17] F. Blaabjerg, C. Zhe, and S. B. Kjaer, "Power electronics as efficient interface in dispersed power generation systems," *Power Electronics, IEEE Transactions on*, vol. 19, pp. 1184-1194, 2004.
- [1.18] J. M. Carrasco, L. G. Franquelo, J. T. Bialasiewicz, E. Galvan, R. C. P. Guisado, M. A. M. Prats, J. I. Leon, and N. Moreno-Alfonso, "Power-Electronic Systems for the Grid Integration of Renewable Energy Sources: A Survey," *Industrial Electronics, IEEE Transactions on,* vol. 53, pp. 1002-1016, 2006.
- [1.19] R. Teodorescu and F. Blaabjerg, "Flexible control of small wind turbines with grid failure detection operating in stand-alone and grid-connected mode," *Power Electronics, IEEE Transactions on,* vol. 19, pp. 1323-1332, 2004.
- [1.20] G. Petrone, G. Spagnuolo, R. Teodorescu, M. Veerachary, and M. Vitelli, "Reliability Issues in Photovoltaic Power Processing Systems," *Industrial Electronics, IEEE Transactions on*, vol. 55, pp. 2569-2580, 2008.
- [1.21] Florida Solar Energy Center. [online]. Available: http://www.fsec.ucf.edu/en/
- [1.22] F. Blaabjerg, R. Teodorescu, M. Liserre, and A. V. Timbus, "Overview of Control and Grid Synchronization for Distributed Power Generation Systems," *Industrial Electronics, IEEE Transactions on*, vol. 53, pp. 1398-1409, 2006.
- [1.23] B. Sahan, Arau, x, S. V. jo, No, C. ding, and P. Zacharias, "Comparative Evaluation of Three-Phase Current Source Inverters for Grid Interfacing of Distributed and Renewable Energy Systems," *Power Electronics, IEEE Transactions on*, vol. 26, pp. 2304-2318, 2011.
- [1.24] "IEEE Application Guide for IEEE Std 1547, IEEE Standard for Interconnecting Distributed Resources with Electric Power Systems," *IEEE Std 1547.2-2008*, pp. 1-207, 2009.
- [1.25] S. Halpin, "Comparison of IEEE and IEC harmonic standards," in *Power Engineering Society General Meeting, IEEE*, pp. 2214-2216, 2005.
- [1.26] C. Yaow-Ming, W. Hsu-Chin, C. Yung-Chu, L. Kung-Yen, and S. Shian-Shing, "The AC Line Current Regulation Strategy for the Grid-Connected PV System," *Power Electronics, IEEE Transactions on*, vol. 25, pp. 209-218, 2010.
- [1.27] G. D. Holmes and T. A. Lipo, Pulse Width Modulation for Power Converters-Principle and Practice Piscataway, NJ: Wiley-IEEE Press-Series on Power Engineering, 2003.
- [1.28] University of Strathclyde, A technical report on "An Investigation on DC Injection into Low-voltage AC System," Project Steering Group Work Stream 4 Project WS4-P11, Contract number DG/CG/00002/00/00, DTI/OFGEM Distributed Generation Co-ordinating Group,June 2005.

### Chapter Two

### Voltage Source Inverter Topologies

#### 2.0. Introduction

This chapter discusses the different voltage source inverter (VSI) topologies commonly used in distributed generation systems, particularly in photovoltaic systems, and highlights the features of each topology. The suitability of each topology for gridconnected photovoltaic application is also discussed. Among the topologies discussed are: single-phase half and full bridge inverters, the three-phase two-level inverter, the neutral-point clamped (three-level diode clamped) inverter, the three-level flying capacitor inverter, the cascaded multilevel inverter with electrically isolated dc sources, and the modular multilevel inverter. However, of these topologies only the three-phase two-level VSI is deemed appropriate for subsequent chapters, which focus on grid integration of photovoltaic inverters.

#### 2.1. Background

The power electronic interface is an important part of distributed generation units as it influences the overall system efficiency and performance. Since photovoltaic (PV) cells generate low voltage dc, normally dc-ac power converters are used to convert the generated dc power into ac, with suitable ac voltage that matches the ac utility. To maximize energy capture from such renewable energy sources, the converters also track the PV maximum power point (MPP)[2.1, 2]. Such power tracking systems are desirable because they speed up the investment return. At present, single-stage and two-stage arrangements are two competing solutions used for grid interfacing of PV systems [2.3]. In a single-stage grid-connected system, the PV system utilizes a single conversion unit (normally dc-ac power inverter) to track the MPP and to interface the PV system to the grid, as illustrated in Figure 0.1(a). Using this concept, the PV maximum power is delivered into the grid, such a topology requires either a step-up

transformer or a PV array with high dc voltage output, which can be realized using a string of series connected PV modules [2.4]. The disadvantages of these single stage solutions can be summarised as follows:

- A PV array with a large dc voltage increases control complexity and excites high leakage current between the panels and the system ground through parasitic capacitances. Also, this approach exacerbates partial shading problems.
- Using PV panels with low dc voltage and a step-up transformer to match the inverter output to the grid, increases inverter losses. This solution is also limited to small-scale applications. In addition, a 50/60Hz matching transformer is bulky.

For single-stage arrangements, different types of dc-ac inverters, such as single-phase half-bridge and full-bridge inverters, three-phase two-level inverters, and multilevel inverters, can be used. From these, there are two inverter topologies that are commonly used as grid interfacing unit of photovoltaic system; voltage source inverter and current source inverter. The use of current source inverter topologies will be discussed further in Chapter 5.



Figure 0.1: General block diagram of (a) single-stage arrangement grid connected PV system and (b) twostage arrangement grid connected PV system.

For two-stage arrangements, a step-up dc-dc inverter can be used to boost the PV output voltage and perform maximum power tracking, while a dc-ac inverter is used to regulate the dc link voltage and provide a clean ac output current to the grid, as shown in Figure 0.1(b). The main advantage of this solution is the simplicity of the control structure since the control requirements are distributed between the two stages. Additionally, a PV array with a high dc voltage is not required, because the dc-dc converter boosts the voltage to the desired level. Since this topology consists of a two-stage arrangement, the system efficiency is expected to be lower and the cost is increased, compared to a single-stage arrangement [2.5].

#### 2.2. DC-AC power inverter topologies

#### 2.2.1. Single-phase half-bridge inverter

The topology of a half-bridge VSI is shown in Figure 0.2, where two dc capacitors are required to provide a return path for the load current, and ensure the inverter output voltage is modulated between  $\pm \frac{1}{2}V_{dc}$ . An additional benefit of having such capacitors is that the interfacing transformer no longer needs to be designed to block dc voltage stresses. Since the inverter output current does contain a significant dc offset, the voltage across these capacitors will remain virtually constant at  $\frac{1}{2}V_{dc}$  (for an input dc voltage  $V_{dc}$ ). Table 0.1 lists the two switch states and the corresponding output voltages of the single-phase half-bridge inverter. The switches S<sub>1</sub> and S<sub>2</sub> must not be turned on simultaneously to avoid a short circuit across the dc-link. Such a consideration is normally incorporated into the modulating technique in the form of dead time. This dead time guarantees complementary operation of switches S<sub>1</sub> and S<sub>2</sub>. The same principle is applicable to other VSI topologies.

The VSI uses self-commutated switching devices, which must be able to conduct current in both directions to ensure bi-direction power flow capability. For this reason, a VSI uses insulated gate bipolar transistors (IGBTs) with inverse anti-parallel diodes.

The single-phase half-bridge inverter in Figure 0.2 can be controlled using selective harmonic elimination (SHE) or high-frequency carrier based pulse-width modulation. However, SHE is not suitable for single-phase inverters because all the low-order odd harmonics needed to be eliminated, including the triplen. By doing so, the dc link

voltage utilization is reduced (corresponding modulation index  $m_a=1$ ), and the system may suffer from high switching losses. Therefore, carrier based pulse width modulation (PWM) is favoured [2.6]. The same principle is applicable to most VSIs, with some minor modifications.



Figure 0.2: Topology of half-bridge single-phase VSI.

Table 0.1: Switch states and the outputs.

Output voltage	Switch	n states	Power path		
	$\mathbf{S}_1$	$S_2$	$i_0 > 0$	<i>i</i> <sub>o</sub> <0	
$1/2 V_{dc}$	ON	OFF	$S_1C_1$	$D_1C_1$	
$-\frac{1}{2}V_{dc}$	OFF	ON	$D_2C_2$	$S_2C_2$	

#### 2.2.2 Single-phase full-bridge inverter

Figure 0.3 shows the schematic of a single-phase full-bridge inverter, which is also called an H-bridge, due to the arrangement of the power switches and the load. This inverter topology generates three voltage levels,  $\pm V_{dc}$ , and 0 at the inverter output. Table 2.2 summarises the switch combinations that can be utilized to generate the output voltage of the single-phase H-bridge inverter, assuming  $V_{dc}$  is the input dc voltage [2.6, 7].In single-phase inverter systems, the dc input current is pulsating, hence the power pulsates and a relatively large dc-link capacitor is required[2.8]. A pulsating input dc current may affect the lifetime of the PV arrays and the input filter capacitor. A single-phase topology is usually used with 50/60Hz transformer isolation as a grid-connected inverter for residential or small-scale PV systems, for powers between 1kW<sub>peak</sub> and  $6kW_{peak}[2.4, 9-12]$ .



Figure 0.3: Topology of H-bridge single-phase VSI.

 Table 0.2: Summary of the switch states of a single-phase H-bridge inverter that can be used to generate a three-level output voltage.

Output voltage		Switch	Power path			
Output voltage	<b>S</b> <sub>1</sub>	$S_2$	<b>S</b> <sub>3</sub>	$S_4$	$i_o > 0$	<i>i</i> <sub>o</sub> <0
V <sub>dc</sub>	ON	OFF	OFF	ON	$S_1S_4$	$D_1D_4$
0	ON	ON	OFF	OFF	$S_1D_2$	$S_2D_1$
	OFF	OFF	ON	ON	$S_4D_3$	$S_3D_4$
- <i>V</i> <sub>dc</sub>	OFF	ON	ON	OFF	$D_2D_3$	$S_2S_3$

#### 2.3. Three-phase two-level voltage source inverter

The topology in Figure 0.4 is basically an extension of the half-bridge version of the single-phase inverter, by combining three legs to form a three-phase two-level VSI. Each phase of a three-phase two-level inverter generates two voltage levels,  $\pm \frac{1}{2}V_{dc}$ , as with the half-bridge version. The control strategy is similar to that of the single-phase inverter, except the reference signals for the different phase-legs are shifted 120° from each other. The main advantages of this topology is that all triplen harmonics such as  $3^{rd}$ ,  $9^{th}$ ,  $15^{th}$ , etc. are eliminated in the line-to-line output voltage, so they will not appear in the currents in a three wire system [2.6, 13]. Since the sum of the currents in a balanced three-phase system is zero, the instantaneous ac output power is constant. Hence, dc input power and current are constant but also include some high switching frequency components that result from the modulation process. This means a relatively small dc capacitor is sufficient to filter these high switching frequency components. The constant dc input current is one of the features that make three-phase VSI an attractive interfacing option for PV systems (or any constant dc energy source)[2.14].

Three-phase two-level VSI has inherent features such as generation of leading and lagging reactive power, high power density (MVA per unit volume), fast dynamic response, and ability to operate under challenging conditions in medium and highvoltage applications[2.15]. These features make the three-phase two-level VSI the favourite candidate for medium and large scale solar farms [2.16-18]. The three-phase two-level inverter in such applications uses a relatively high switching frequency to minimize the ac filter size and to achieve a high dynamic response, but this can also be considered as its main disadvantage. This is because such a high switching frequency increases the dv/dt stresses on the interfacing transformers, hence imposing high insulation requirements to withstand switching of large voltage steps at high frequency (1kHz to 2kHz) [2.19]. Since galvanic isolation is a requirement imposed by many grid codes for medium and large scale grid connected inverters, a transformerless solution is a viable option to reduce cost but only at a power level permitted by the grid code. In PV applications, the three-phase two-level VSI is the most commonly used interface unit between the PV panels and the ac grid, due to its simplicity and availability [2.3, 16, 18, 20-23].

The three-phase two-level VSI in Figure 0.4 can be controlled using most known modulation techniques, such as carrier based modulation techniques, as with the single-phase inverter and its derivatives (sinusoidal with some form for of  $3^{rd}$  harmonic injection), space vector modulation, and selective harmonic elimination. Details of these modulation strategies will be discussed in the subsequent chapters.



Figure 0.4: Topology of three phase two-level VSI.

#### 2.4. Multilevel inverter

Recently, multilevel inverters have attracted attention because of their ability to overcome three-phase two-level VSI problems such as high voltage stress on the power switches and interfacing transformer, and high power losses due to a high switching frequency. Additionally, they offer lower THD (total harmonic distortion) due to a phase output voltage formation using small voltage steps with reduced switching frequency per device, low dv/dt, and a low common mode voltage [2.19, 24-26]. However, these advantages are achieved at the expense of increased power circuit complexity. Also a bulky transformer and an electrolytic capacitor are still required to meet galvanic isolation in a grid mode. A few multilevel inverter topologies are used in PV systems, namely the neutral-point clamped inverter, cascaded with electrically isolated inverter, and modular multilevel, which are briefly discussed in the following sections.

#### 2.4.1. Neutral-point clamped inverter

Figure 0.5 show one phase of the neutral-point clamped inverter (also known as a threelevel diode clamped inverter) that can generate three output levels  $\pm \frac{1}{2}V_{dc}$  and 0 between phase 'a' and reference point '0'. Table 0.3 lists the switch states of neutral-point clamped inverter and their corresponding output voltages. The voltage across each dclink capacitor must be maintained at  $\frac{1}{2}V_{dc}$  in order to ensure the voltage stress on each individual device is limited to  $\frac{1}{2}V_{dc}$  [2.19, 27]. The use of high frequency PWM to control the switches of the neutral-point clamped inverter will suppress the voltage harmonics around and beyond the switching frequency components, and therefore allows the use of small ac output filters to attenuate these high frequency harmonics.

In the past there was expectation of extending the diode clamped inverter to generate more than three voltage levels. However, this did not materialise as dc link capacitance balancing proved to be impossible over the entire inverter operating range when the number of levels exceeded three. For this reason most of the practical applications of the diode clamped inverter are limited to the three-level inverter (NPC) and its derivative, the active neutral-point clamped (ANPC) inverter. The NPC or ANPC converter preserve the advantages of the two-level three-phase inverter, offer additional benefits such as reduced switching losses and low dv/dt, and permit the use of relatively small ac filters [2.28, 29].



Figure 0.5: Topology of one-phase of neutral-point clamped inverter.

Output voltage level		Switch	states		Current polarity	Current path	
V <sub>a0</sub>	S <sub>a1</sub>	S <sub>a2</sub>	$S_{a3}$	S <sub>a4</sub>	Current polarity	Current path	
+1/V	ON	ON	OFF	OFF	$i_a > 0$	$S_{a1}, S_{a2}$	
$+\frac{1}{2}V_{dc}$	ON	UN	OFF		<i>i</i> <sub>a</sub> <0	D <sub>a1</sub> ,D <sub>a2</sub>	
0	OFF	ON	ON	OFF	$i_a > 0$	D <sub>a5</sub> ,S <sub>a2</sub>	
					$i_a < 0$	$S_{a3}, D_{a6}$	
$-\frac{1}{2}V_{dc}$	$-\frac{1}{2}V_{d_{2}}$ OFF O		ON	ON	$i_a > 0$	D <sub>a3</sub> ,D <sub>a4</sub>	
$-72V_{dc}$		OFF OFF			$i_a < 0$	S <sub>a3</sub> ,S <sub>a4</sub>	

Table 0.3: Summary of the switch state combinations and their corresponding output voltage.

#### 2.4.2. Capacitor-clamped multilevel inverter

Figure 0.6 shows one phase of the three-phase flying capacitor multilevel inverter. The voltage across each capacitor must be maintained at  $\frac{1}{2}V_{dc}$ , therefore the maximum voltage stress on any switching device is limited to one capacitor voltage [2.24, 25, 30]. The inverter in Figure 0.6 provides a three-level output between 'a' and '0' as listed in Table 2.4, which summarizes the switch states and corresponding inverter output voltages, in addition to the current paths. In this topology, the switch states that produce

the same phase voltage level are referred to as redundant switch states. This topology has several disadvantages which restrict its use. Converter initialization is a problem where the clamping capacitors of the flying capacitor multilevel inverter must be fully pre-charged before being modulated. This complicates the modulation process and hinders converter performance under ride-through conditions. Another problem concerns clamping capacitor ratings. Since these are rated at a large percentage of dc-bus voltage, these dc capacitor voltage ratings must be large compared to those of the diode-clamped inverter. Due to these limitations, there has been limited development of this topology, and it is not deemed suitable for PV applications [2.13]. Moreover, both the diode and flying capacitor clamped inverters exacerbate the problem of partial shading as they require a large number of series connected PV cells in order to build-up a relatively high dc voltage appropriate for medium-voltage applications [2.25, 31].



Figure 0.6: One phase of three-level flying capacitor (also known as capacitor clamped) inverter.

Output voltage level	Switch states				Current polarity	Current noth	
$V_{a0}$	S <sub>a1</sub>	S <sub>a2</sub>	S <sub>a3</sub>	S <sub>a4</sub>	Current polarity	Current path	
$+\frac{1}{2}V_{dc}$	ON	ON	OFF	F OFF	$i_a > 0$	$S_{a1}, S_{a2}$	
1/2V dc	UN	UN	UIT		$i_a < 0$	D <sub>a1</sub> ,D <sub>a2</sub>	
	ON	ON OFF	ON	OFF	$i_a > 0$	$S_{a1}, C_1, D_{a3}$	
0	ON		UN		$i_a < 0$	$S_{a3}, C_1, D_{a1}$	
U U	OFF	ON	OFF	ON	$i_a > 0$	$D_{a4}, C_1, S_{a2}$	
					$i_a < 0$	$S_{a4}, C_1, D_{a2}$	
1/ V	$V_{dc}$ OFF OFF O	OFF	ON	ON	$i_a > 0$	D <sub>a3</sub> ,D <sub>a4</sub>	
-/2V dc		UN	UN	<i>i</i> <sub>a</sub> <0	$S_{a3}, S_{a4}$		

Table 0.4: Summary of the switch state combinations and their corresponding output voltages.

#### 2.4.3. Cascaded multilevel inverter

The cascaded multilevel inverter has received attention due to its simplicity and power stage modularity, which makes it scalable to medium voltage applications [2.32]. The cascaded multilevel inverter exploits series connection of single-phase H-bridge inverters with separate dc sources to build-up high output voltage using low voltage rated switching devices. Figure 0.7 shows the topology for one phase leg of a nine-level inverter with four cells in each phase [2.24, 33]. Exploiting bipolar voltage generation ability of each single-phase H-bridge cell, different voltage levels in the phase output voltage of the cascaded multilevel inverter are synthesized using the available switch combinations provided by the H-bridge cells. Each single-phase H-bridge inverter generates three voltage levels at its output:  $\pm V_{dc}$  and 0, as stated in section 0. Thus, for example, a cascaded multilevel inverter with four cells per phase leg can generate a maximum of nine voltage levels that vary between  $\pm 4V_{dc}$  and 0. This allows the production of a good sine-wave pre-filtered output voltage with staircase or pulse width modulation.

The cascaded inverter topology is attractive in PV applications because of its relative low voltage isolated dc sources, which can be replaced by PV arrays. These dc sources need not be of the same magnitude, and such a low voltage requirement of the H-bridge cells can be provided by fewer PV cells in series, hence the problem of partial shading is avoided. Since a PV array can be considered as an active dc source, capacitor voltage balancing is no longer required. Capacitive PV array currents to ground will be problematic because the various arrays are not grounded and experience small progressive dv/dt steps.



Figure 0.7: Power circuit of one phase leg of nine-level cascaded multilevel inverter.

#### 2.4.4. Modular multilevel

Figure 0.8 shows one-phase of a three-level modular inverter which is composed of two groups of switches: the main switches are  $S_{a1}$ ,  $S_{a2}$ ,  $S_{a3}$  and  $S_{a4}$  and the auxiliary switches are  $S_{x1}$ ,  $S_{x2}$ ,  $S_{x3}$  and  $S_{x4}$ . There are four complementary switch pairs per phase ( $S_{a1}$ ,  $S_{x1}$ ), ( $S_{a2}$ ,  $S_{x2}$ ), ( $S_{a3}$ ,  $S_{x3}$ ) and ( $S_{a4}$ ,  $S_{x4}$ ) such that  $S_{ai}+S_{xi}=1$  where *i*=1 to 4. Each capacitor in the circuit is maintained at  $\frac{1}{2}V_{dc}$  and the voltage stress on each switching device is limited to one capacitor voltage [2.30, 34]. With reference to the supply midpoint 0 as in Figure 0.8, the three-level waveform between output phase 'a' and 0 can be synthesized as follows:

- For  $V_{a0}=\frac{1}{2}V_{dc}$ , the upper main switches (S<sub>a1</sub>and S<sub>a2</sub>) and lower auxiliary switches (S<sub>x3</sub> and S<sub>x4</sub>) are all turned on.
- For  $V_{a0}=0$ , there are four different switch combinations:
  - i.  $S_{a1}$ ,  $S_{a3}$ ,  $S_{x2}$  and  $S_{x4}$  are turned on.
  - ii.  $S_{a2}$ ,  $S_{a3}$ ,  $S_{x1}$  and  $S_{x4}$  are turned on.
  - iii.  $S_{a2}$ ,  $S_{a4}$ ,  $S_{x1}$  and  $S_{x3}$  are turned on.
  - iv.  $S_{a1}$ ,  $S_{a4}$ ,  $S_{x2}$  and  $S_{x3}$  are turned on.
- For  $V_{a0}$ =- $\frac{1}{2}V_{dc}$ , the upper auxiliary switches (S<sub>x1</sub> and S<sub>x2</sub>) and lower main switches (S<sub>a3</sub> and S<sub>a4</sub>) are all turned on.

Table 0.5 summarizes the voltage levels and their corresponding switch states [2.26]. In the modular multilevel inverter, high frequency carrier based PWM is favoured over space vector modulation because it reduces the capacitor voltage balancing method complexity, and improves the capacitor voltage balancing method's robustness to different operating conditions, including unbalanced operating conditions and asymmetrical faults.

In a two-stage grid connected PV system, a modular multilevel inverter regulates the boost converter output voltage to be fixed and regulates the injected current into the grid to achieve unity power factor operation [2.34]. The modular multilevel inverter alleviates the disadvantages of conventional multilevel inverters such as the difficulty of capacitor voltage balancing with higher order levels, the need for ac output filtering and an interfacing transformer. Another advantage of the modular multilevel inverter is that it requires only one dc source and is able to synthesize intermediate voltage levels using its cell capacitors. Furthermore, it utilizes phase redundancy to balance the dc link capacitors regardless of number of levels, independent of power factor and modulation index. Also, it provides capability for fault management and fault ride-through [2.25, 27].

In order to realise these advantages, however, modular multilevel converters require a large number of active and passive components, and a robust capacitor voltage balancing strategy to control the voltage stress across each switching device. These requirements increase both the circuit and control complexity.



Figure 0.8: Power circuit of one phase leg of three-level modular inverter.

Output voltage	ut voltage Switch states								
$V_{a0}$	S <sub>a1</sub>	S <sub>a2</sub>	S <sub>a3</sub>	S <sub>a4</sub>	S <sub>x1</sub>	S <sub>x2</sub>	S <sub>x3</sub>	S <sub>x4</sub>	
1/2V <sub>dc</sub>	ON	ON	OFF	OFF	OFF	OFF	ON	ON	
0	ON	OFF	ON	OFF	OFF	ON	OFF	ON	(i)
	OFF	ON	ON	OFF	ON	OFF	OFF	ON	(ii)
	OFF	ON	OFF	ON	ON	OFF	ON	OFF	(iii)
	ON	OFF	OFF	ON	OFF	ON	ON	OFF	(iv)
$-\frac{1}{2}V_{dc}$	OFF	OFF	ON	ON	ON	ON	OFF	OFF	

Table 0.5: Switch combinations for a three-level modular inverter.
# 2.5. Summary

This chapter presented single-stage and two-stage converter arrangements as possible integration options for a grid-connected photovoltaic system. In both arrangements, the dc-to-ac inverter configurations, including single-phase, three-phase and multilevel, are discussed, and suitability and practicality of each topology for PV applications are highlighted. For simplicity and other features previously highlighted, the three-phase two-level VSI is selected as a suitable interfacing option to be used for the grid integration of the PV system in this thesis, which will be further discussed in subsequent chapters.

# References

- [2.1] F. Blaabjerg, R. Teodorescu, M. Liserre, and A. V. Timbus, "Overview of Control and Grid Synchronization for Distributed Power Generation Systems," *Industrial Electronics, IEEE Transactions on*, vol. 53, pp. 1398-1409, 2006.
- [2.2] F. Blaabjerg, C. Zhe, and S. B. Kjaer, "Power electronics as efficient interface in dispersed power generation systems," *Power Electronics, IEEE Transactions on*, vol. 19, pp. 1184-1194, 2004.
- [2.3] W. Tsai-Fu, C. Chih-Hao, L. Li-Chiun, and K. Chia-Ling, "Power Loss Comparison of Single- and Two-Stage Grid-Connected Photovoltaic Systems," *Energy Conversion, IEEE Transactions on*, vol. 26, pp. 707-715, 2011.
- [2.4] S. Jain and V. Agarwal, "A Single-Stage Grid Connected Inverter Topology for Solar PV Systems With Maximum Power Point Tracking," *Power Electronics, IEEE Transactions on*, vol. 22, pp. 1928-1940, 2007.
- [2.5] S. B. Kjaer, J. K. Pedersen, and F. Blaabjerg, "A review of single-phase gridconnected inverters for photovoltaic modules," *Industry Applications, IEEE Transactions on*, vol. 41, pp. 1292-1306, 2005.
- [2.6] M. H.Rashid. (2007). Power Electronic Handbook (2nd edition ed.).
- [2.7] B. Wu. (2006). *High Power Converters and AC Drives*.
- [2.8] G. Petrone, G. Spagnuolo, R. Teodorescu, M. Veerachary, and M. Vitelli, "Reliability Issues in Photovoltaic Power Processing Systems," *Industrial Electronics, IEEE Transactions on*, vol. 55, pp. 2569-2580, 2008.
- [2.9] N. Kasa and T. Iida, "Flyback type inverter for small scale photovoltaic power system," in *IECON 02 [Industrial Electronics Society, IEEE 2002 28th Annual Conference of the]*, 2002, pp. 1089-1094 vol.2.
- [2.10] C.-M. Wang, "A novel single-stage full-bridge buck-boost inverter," in *Applied Power Electronics Conference and Exposition, 2003. APEC'03. Eighteenth Annual IEEE*, 2003, pp. 51-57.
- [2.11] H. Yi, S. Miaosen, F. Z. Peng, and W. Jin, "Z-Source Inverter for Residential Photovoltaic Systems," *Power Electronics, IEEE Transactions on*, vol. 21, pp. 1776-1782, 2006.

- [2.12] T. Kerekes, R. Teodorescu, P. Rodriguez, G. Vazquez, and E. Aldabas, "A New High-Efficiency Single-Phase Transformerless PV Inverter Topology," *Industrial Electronics, IEEE Transactions on*, vol. 58, pp. 184-191, 2011.
- [2.13] D. G. Holmes and T.A.Lipo, Pulse Width Modulation for Power Converter: Principle and Practice. United States: IEEE Press Series on Power Engineering: John Wiley & Sons, 2003.
- [2.14] N. A. Rahim and S. Mekhilef, "Implementation of three-phase grid connected inverter for photovoltaic solar power generation system," in *Power System Technology, 2002. Proceedings. PowerCon 2002. International Conference on*, 2002, pp. 570-573 vol.1.
- [2.15] Y. Zhang, G. P. Adam, T. C. Lim, S. J. Finney, and B. W. Williams, "Voltage source converter in high voltage applications: Multilevel versus two-level converters," in AC and DC Power Transmission, 2010. ACDC. 9th IET International Conference on, 2010, pp. 1-5.
- [2.16] L. Wu, Z. Zhao, and J. Liu, "A Single-Stage Three-Phase Grid-Connected Photovoltaic System With Modified MPPT Method and Reactive Power Compensation," *Energy Conversion, IEEE Transactions on*, vol. 22, pp. 881-886, 2007.
- [2.17] J. M. Carrasco, L. G. Franquelo, J. T. Bialasiewicz, E. Galvan, R. C. P. Guisado, M. A. M. Prats, J. I. Leon, and N. Moreno-Alfonso, "Power-Electronic Systems for the Grid Integration of Renewable Energy Sources: A Survey," *Industrial Electronics, IEEE Transactions on,* vol. 53, pp. 1002-1016, 2006.
- [2.18] T. Kerekes, M. Liserre, R. Teodorescu, C. Klumpner, and M. Sumner, "Evaluation of Three-Phase Transformerless Photovoltaic Inverter Topologies," *Power Electronics, IEEE Transactions on*, vol. 24, pp. 2202-2211, 2009.
- [2.19] L. Jih-Sheng and P. Fang Zheng, "Multilevel converters-a new breed of power converters," *Industry Applications, IEEE Transactions on*, vol. 32, pp. 509-517, 1996.
- [2.20] S. A. Azmi, G. P. Adam, K. H. Ahmed, S. J. Finney, and B. W. Williams, "Grid Interfacing of Multimegawatt Photovoltaic Inverters," *Power Electronics, IEEE Transactions on*, vol. 28, pp. 2770-2784, 2013.
- [2.21] L. Wu, Z. Zhao, J. Liu, S. Liu, and L. Yuan, "Modified MPPT strategy applied in single-stage grid-connected photovoltaic system," in *Electrical Machines and Systems, 2005. ICEMS 2005. Proceedings of the Eighth International Conference on, 2005, pp. 1027-1030 Vol. 2.*
- [2.22] M. C. Cavalcanti, K. C. de Oliveira, A. M. de Farias, F. A. S. Neves, G. M. S. Azevedo, and F. C. Camboim, "Modulation Techniques to Eliminate Leakage Currents in Transformerless Three-Phase Photovoltaic Systems," *Industrial Electronics, IEEE Transactions on*, vol. 57, pp. 1360-1368, 2010.
- [2.23] C. C. Hou, C. C. Shih, P. T. Cheng, and A. M. Hava, "Common-Mode Voltage Reduction Pulsewidth Modulation Techniques for Three-Phase Grid-Connected Converters," *Power Electronics, IEEE Transactions on*, vol. 28, pp. 1971-1979, 2013.
- [2.24] J. Rodriguez, L. Jih-Sheng, and P. Fang Zheng, "Multilevel inverters: a survey of topologies, controls, and applications," *Industrial Electronics, IEEE Transactions on*, vol. 49, pp. 724-738, 2002.

- [2.25] G. P. Adam, B. Alajmi, K. H. Ahmed, S. J. Finney, and B. W. Williams, "New flying capacitor multilevel converter," in *Industrial Electronics (ISIE)*, 2011 *IEEE International Symposium on*, 2011, pp. 335-339.
- [2.26] O. A. Lara. G.P.Adam, G.Burt, D.Telford, B.W. Williams and J.R. McDonald, "Modular multilevel inverter: pulse width modulation and capacitor balancing technique," *IET Power Electronics*, vol. 3, pp. 702-715, 2010.
- [2.27] O. A. Lara. G.P.Adam, G.Burt, S.J.Finney and B.W.Williams, "Comparison between Two VSC-HVDC Transmission Systems Technologies: Modular and Neutral Point Clamped Multilevel Converter," presented at the 13th Annual Conference of Industrial Electronic Society IECON2009, Porto-portugal, November 2009.
- [2.28] X. Huafeng and X. Shaojun, "Transformerless Split-Inductor Neutral Point Clamped Three-Level PV Grid-Connected Inverter," *Power Electronics, IEEE Transactions on*, vol. 27, pp. 1799-1808, 2012.
- [2.29] E. Pouresmaeil, D. Montesinos-Miracle, and O. Gomis-Bellmunt, "Control Scheme of Three-Level NPC Inverter for Integration of Renewable Energy Resources Into AC Grid," *Systems Journal, IEEE*, vol. 6, pp. 242-253, 2012.
- [2.30] O. A. Lara. G.P.Adam, G.Burt, S.J.Finney and B.W.Williams, "Comparison between flying capacitor and modular multilevel inverter," presented at the 13th Annual Conference of Industrial Electronic Society IECON2009, Portugal, November 2009.
- [2.31] G. P. Adam, S. J. Finney, A. M. Massoud, and B. W. Williams, "Capacitor Balance Issues of the Diode-Clamped Multilevel Inverter Operated in a Quasi Two-State Mode," *Industrial Electronics, IEEE Transactions on*, vol. 55, pp. 3088-3099, 2008.
- [2.32] G. Grandi, C. Rossi, D. Ostojic, and D. Casadei, "A New Multilevel Conversion Structure for Grid-Connected PV Applications," *Industrial Electronics, IEEE Transactions on*, vol. 56, pp. 4416-4426, 2009.
- [2.33] M. Calais, V. G. Agelidis, and M. S. Dymond, "A cascaded inverter for transformerless single-phase grid-connected photovoltaic systems," *Renewable Energy*, vol. 22, pp. 255-262, 2001.
- [2.34] B. N. Alajmi, K. H. Ahmed, G. P. Adam, S. J. Finney, and B. W. Williams, "Modular multilevel inverter with maximum power point tracking for grid connected photovoltaic application," in *Industrial Electronics (ISIE), 2011 IEEE International Symposium on*, 2011, pp. 2057-2062.

# **Chapter Three**

# Modulation Techniques for the Voltage Source Inverter

#### 3.0. Introduction

This chapter discusses a number of pulse width modulation (PWM) techniques used to generate gating signals for a three-phase two-level voltage source inverter (VSI). These modulation techniques, namely carrier-based sinusoidal pulse width modulation, third harmonic injection pulse width modulation, space vector modulation and selective harmonic elimination, are addressed and assessed in this chapter, and the features of each modulation technique are presented. A new implementation of the selective harmonic elimination (SHE) technique that exploits the 3<sup>rd</sup> harmonic to improve performance at low and high modulation indices, is presented, and validated by simulation and experimentation.

## 3.1. Background

In principle, all two-level three-phase inverter modulation techniques create a train of pulses that have the same fundamental voltage as a target reference voltage. However, these pulse trains contain unwanted harmonic components which should be minimised. The duty cycles for the pulses must be calculated, and the best way to place these pulses within each switching cycle must be determined, taking into consideration minimisation of unwanted harmonics and other objectives such as low switching loss. A number of PWM techniques may be used to generate gating signals for the two-level three-phase VSI. In the literature, the most commonly discussed PWM techniques used to control a three-phase two-level VSI are carrier-based sinusoidal pulse width modulation (SPWM) [3.1-5], third harmonic injection pulse width modulation (THIPWM) [3.1, 6,7], space vector modulation (SVM) [3.7-12] and selective harmonic elimination (SHE) [3.13-23]. Since SHE is known for better harmonic performance than SVM and THIPWM but

poorer dc voltage utilization, like SPWM, this thesis proposes an improved SHE implementation with adjustment of third harmonic injection to preserve the advantage of standard SHE but with improved dc link utilization. The initial objective of third harmonic adjustment is to extend the maximum modulation index as the switching angle increases. The adjustment of the 3<sup>rd</sup> harmonic in SHE does not result in decoupling of the modulation index from the number of switching angles, however, it remains useful at high modulation indices where standard SHE tends to produce closely located switching angles (which cannot be realised practically). Thus, the proposed SHE results in wider separation of switching angles for the same modulation index as standard SHE, which leads to more viable practical realisation.

## 3.2. Carrier-based Sinusoidal Pulse Width Modulation

A circuit diagram of a three-phase two-level VSI that comprises six active switches,  $S_1$  to  $S_6$  with a free-wheeling diode in parallel with each switch, is shown in Figure 0.1.



Figure 0.1: Topology of three phase two-level voltage source inverter.

Carrier-based sinusoidal pulse width modulation (SPWM) refers to the generation of PWM gating signals by comparing sinusoidal modulating signals ( $v_{ma}$ ,  $v_{mb}$  and  $v_{mc}$ ) to a high frequency carriers ( $v_{cr}$ ) to generate the on and off periods of the PWM output signals, as shown in Figure 0.2(a) [3.24].

The fundamental-frequency component in the inverter output voltage can be determined by the amplitude modulation index,  $m_a$ 

$$m_a = \frac{\dot{V}_m}{\dot{V}_{cr}} \tag{3.1}$$

where  $\hat{V}_m$  and  $\hat{V}_{cr}$  are the peak values of the modulating and carrier signals respectively. The amplitude modulation index  $m_a$  is adjusted by varying  $\hat{V}_m$  while  $\hat{V}_{cr}$  is fixed. The frequency modulation index is defined by

$$m_f = \frac{f_{cr}}{f_m} \tag{3.2}$$

where  $f_m$  and  $f_{cr}$  are the frequencies of the modulating and carrier signals respectively. For instance, when  $v_{ma} \ge v_{cr}$ , the upper switch S<sub>1</sub> in leg 'a' is turned on while the complimentary switch S<sub>4</sub> in the same leg is turned off, leading to the inverter voltage  $v_{AN}$  being equal to  $+\frac{1}{2}V_{dc}$ . When  $v_{ma} \le v_{cr}$ , S<sub>4</sub> is on and S<sub>1</sub> is off leading to  $v_{AN}$  equalling to  $-\frac{1}{2}V_{dc}$ , as shown in Figure 0.2(a). These voltage signals ( $v_{AN}$  and  $v_{BN}$ ) are defined with respect to the midpoint of the dc bus. The inverter line-to-line voltage,  $v_{AB}$  is determined by  $v_{AB} = v_{AN} - v_{BN}$ . The peak amplitude of the modulating signals,  $V_m$ , determines the amplitude modulation index  $m_a$  and in turn controls the rms value of the output line-to-line voltage,  $v_{AB}$ . The inverter switching frequency,  $f_{SW}$  is determined by the carrier frequency,  $f_{cr}$ . Figure 0.2(b) to (d) illustrates the experimental waveforms of the two-level three phase VSI line-to-line voltage,  $v_{AB}$ , load voltage, and current respectively. The main advantage of SPWM is its simple implementation, but it is unable to fully utilize the dc link voltage, therefore does not maximise the power density [3.25, 26].



Figure 0.2: Modulation of the three phase VSI using SPWM and sample of open-loop experimental waveforms ( $V_{dc}$ =100V,  $m_a$ =0.8 and  $f_{sw}$ =2000Hz).

#### 3.3. Third Harmonic Injection Pulse Width Modulation (THIPWM)

The modulating signal of THIPWM is generated by injecting selected harmonics into the reference signal of each leg. This technique provides a 15.5% higher fundamental amplitude than SPWM, which enables full utilisation of dc-link voltage. The modulating signal,  $v_{mx}$  is defined as [3.24]:

$$v_{mx} = m_a \left( \sin \left( \omega t + \phi \right) + \frac{1}{6} \sin 3 \omega t \right)$$
(3.3)

where  $m_a$  is the amplitude modulation index,  $\phi = 0, \frac{2\pi}{3}, \frac{4\pi}{3}$ , and x = a, b or *c*.

Figure 0.3(a) shows the modulating signals ( $v_{ma}$ ,  $v_{mb}$  and  $v_{mc}$ ) composed of the fundamental and third harmonic components, and highlights the flattened top of the modulating signals. As a result, the peak fundamental signal,  $V_m$  can be higher than the peak of the carrier signal,  $V_{cr}$  which boosts the fundamental line-to-line voltage,  $v_{AB}$ . A sample simulation result of the line-to-line voltage waveform  $v_{AB}$  with modulation index,  $m_a$ =0.8 and switching frequency,  $f_{sw}$ =2kHz, is illustrated in Figure 0.3(b).

Third harmonic injection does not affect the quality of the output voltage because the three-phase output does not contain triple harmonics [3.1]. The main advantage of this technique is simplicity in hardware and implementation. However, the disadvantage is that it is only applicable to a three-phase three-wire system.



(b) Line-to-line voltage,  $v_{ab}$  waveform with repsect to midpoint of the dc bus

Figure 0.3: Reference modulating and output voltage waveforms when the VSI is controlled with third harmonic injection PWM.

#### **3.4.** Space vector modulation (SVM)

Space vector modulation is based on eight possible switch state combinations for a three-phase VSI. A switch constraint for the VSI is that each leg contains two series switching devices operating in a complementary manner [3.7]. If the upper switch is turned on, the corresponding lower switch is off, and vice versa. The 'on' switch is represented by 1 and the 'off' switch is represented by 0, as stated in Table 0.1. The switched output voltage can be determined by the states of three upper switches  $S_1$ ,  $S_3$  and  $S_5$  of phase 'a', 'b' and 'c' respectively, as in Figure 0.1. The resultant output phase and line-to-line voltages are produced with respect to the corresponding 'on' and 'off' states of three upper switches.

$S_1$	<b>S</b> <sub>3</sub>	$S_5$	V <sub>an</sub>	$V_{bn}$	V <sub>cn</sub>	$V_{ab}$	$V_{bc}$	V <sub>ca</sub>
0	0	0	0	0	0	0	0	0
1	0	0	$\frac{2}{3}V_{dc}$	$-\frac{1}{3}V_{dc}$	$-\frac{1}{3}V_{dc}$	$V_{dc}$	0	$-V_{dc}$
1	1	0	$\frac{1}{3}V_{dc}$	$\frac{1}{3}V_{dc}$	$-\frac{2}{3}V_{dc}$	0	$V_{dc}$	$-V_{dc}$
0	1	0	$-\frac{1}{3}V_{dc}$	$\frac{2}{3}V_{dc}$	$-\frac{1}{3}V_{dc}$	$-V_{dc}$	$V_{dc}$	0
0	1	1	$-\frac{2}{3}V_{dc}$	$\frac{1}{3}V_{dc}$	$\frac{1}{3}V_{dc}$	$-V_{dc}$	0	$V_{dc}$
0	0	1	$-\frac{1}{3}V_{dc}$	$-\frac{1}{3}V_{dc}$	$\frac{2}{3}V_{dc}$	0	$-V_{dc}$	$V_{dc}$
1	0	1	$\frac{V_3}{V_{dc}}$	$-\frac{2}{3}V_{dc}$	$\frac{1}{3}V_{dc}$	$V_{dc}$	$-V_{dc}$	0
1	1	1	$V_{dc}$	$V_{dc}$	$V_{dc}$	0	0	0

Table 0.1: Eight possible switch state combinations of a three-phase VSI.

The relationship between the leg voltage, phase voltage and line-to-line voltage in a two-level three-phase VSI can be expressed as:

$$\begin{bmatrix} V_{an} \\ V_{bn} \\ V_{cn} \end{bmatrix} = \frac{1}{3} \begin{bmatrix} 2 & -1 & -1 \\ -1 & 2 & -1 \\ -1 & -1 & 2 \end{bmatrix} \begin{bmatrix} V_{ao} \\ V_{bo} \\ V_{co} \end{bmatrix} \text{ and } \begin{bmatrix} V_{ab} \\ V_{bc} \\ V_{ca} \end{bmatrix} = \begin{bmatrix} 1 & -1 & 0 \\ 0 & 1 & -1 \\ -1 & 0 & 1 \end{bmatrix} \begin{bmatrix} V_{ao} \\ V_{bo} \\ V_{co} \end{bmatrix}$$
(3.4)

The eight switch state combinations in Table 0.1can be represented in the stationary reference frame  $\alpha$ - $\beta$  by transforming the phase voltages from phase variables *a*, *b* and *c* to  $\alpha$  and  $\beta$  using:

$$\begin{bmatrix} V_{\alpha} \\ V_{\beta} \end{bmatrix} = \begin{bmatrix} 1 & -\frac{1}{2} & -\frac{1}{2} \\ 0 & \frac{\sqrt{3}}{2} & -\frac{\sqrt{3}}{2} \end{bmatrix} \begin{bmatrix} V_{ao} \\ V_{bo} \\ V_{co} \end{bmatrix}$$
(3.5)

where  $V_{ref} = \sqrt{V_{\alpha}^2 + V_{\beta}^2}$  and  $\theta = \tan^{-1} \frac{V_{\beta}}{V_{\alpha}}$ 

Based on this concept, the space vector diagram for a two-level inverter is shown in Figure 0.4. There are six stationery active vectors, namely  $\overline{V_1}$  to  $\overline{V_6}$ , zero vector  $\overline{V_0}$  and the reference vector,  $\overline{V_{ref}}$  that rotate in space at an angular velocity of the fundamental frequency of the inverter voltage [3.26].



Figure 0.4: Space vector diagram for a two-level inverter.

The dwell times for the stationery vectors represents the duty-cycle time (on-state or off-state time) of the chosen switches during a sampling period  $T_s$  for all six sectors that can be calculated as follows:

$$T_{1} = m_{a}T_{s}\sin\left(\frac{n\pi}{3} - \theta\right)$$

$$T_{2} = m_{a}T_{s}\sin\left(\theta - \frac{(n-1)\pi}{3}\right)$$

$$T_{0} = T_{s} - T_{1} - T_{2}$$
(3.6)

where  $m_a = \frac{2}{\sqrt{3}} \frac{V_{ref}}{V_{dc}}$  is amplitude modulation index and *n* is the sector number.

There are many switching sequences proposed for three-phase two-level VSI. Most are aimed at minimizing switching losses by reducing the number of commutations per switching cycle and use the redundant zero vectors in every sequence. Space vector modulation has the ability to transfer harmonic energy into the outer sideband harmonics and, to some extent, to the sidebands of the second carrier group [3.24]. It also offers flexibility in terms of pulse placement in any of the dedicated sectors and switching sequences. Furthermore, SVM increases dc link voltage by 15.5% and semiconductor utilization, improving the power density compared to sinusoidal PWM. It is not best suited for operating in medium-voltage distributed generation systems where the possibility of unbalanced operation and ac faults are high[3.27]. Digital implementation of these modulation strategies with grid connected PV inverters necessitates the use of relatively high switching frequencies for control purposes (the control period is equal to the reciprocal of switching frequency). Figure 0.5shows a simulation result of the line-to-line voltage,  $v_{AB}$  waveform with modulation index,  $m_a=0.8$  and switching frequency,  $f_{SW}=2kHz$ .



Figure 0.5: Line-to-line voltage,  $v_{AB}$  waveform relative to midpoint of the dc bus when the VSI is controlled by SVM.

#### **3.5.** Selective harmonic elimination

The selective harmonic elimination (SHE) technique, based on the idea of Fourier decomposition, is used to eliminate specific lower order harmonics in a switching waveform. SHE is considered useful for low baseband distortion in inverters with a low switching-to-fundamental frequency ratio[3.20].

The Fourier series expansion of the SHE-PWM waveform is given by:

$$V(\omega t) = \sum_{n=1}^{\infty} \left( a_n \cos(n\omega t) + b_n \sin(n\omega t) \right)$$
(3.7)

Owing to the PWM waveform characteristics of odd functions and quarter-wave symmetry,  $a_n = 0$ , for all *n*, and the output voltage reduces to:

$$V(\omega t) = \sum_{n=1}^{\infty} b_n \sin(n\omega t)$$
(3.8)

where  $\omega$  is the radian frequency of the output voltage [3.16, 19].

There are two SHE solutions proposed based on the trajectories of chopping or switching angles over a  $\frac{1}{2}\pi$  period. The switching angles for solution 1 are located between 0 and 60° while for solution 2, the angles are located between 0 and 90°. A generalized expression of  $b_n$  for any number of switching angles for both solutions is given by:

$$b_n = \frac{2V_{dc}}{n\pi} \left[ 2\sum_{k=1}^{N} (-1)^{k-1} \cos n\alpha_k - 1 \right], \text{ in the case of the first arrangement that produces}$$

solution 1, and

$$b_n = \frac{2V_{dc}}{n\pi} \left[ 2\sum_{k=1}^{N} (-1)^k \cos n\alpha_k + 1 \right]$$
in case of the second arrangement that produces

solution 2, for odd values of *n*[3.20, 22, 23].

Figure 0.6(a) and (b) show the normalized(by  $\frac{1}{2}V_{dc}$ ) output voltage of two-level VSI relative to the supply mid-point, when SHE is used to control the inverter, with three notches ( $\alpha_1$ ,  $\alpha_2$  and  $\alpha_3$ ) for fundamental voltage adjustment and elimination of the 5<sup>th</sup> and 7<sup>th</sup> harmonics respectively. As previously mentioned, all switching angles for solution 1 in Figure 0.6(a) are located between 0 and 60° as shown in Figure 0.6(c) (solid line). The maximum attainable modulation index for solution 1 is 1.1884. This maximum modulation index is achieved by solving an optimization problem that maximizes the modulation index. The solution uses equations for the elimination of the 5<sup>th</sup> and 7<sup>th</sup>harmonics as nonlinear constraints, and  $\alpha_1 < \alpha_2$ ,  $\alpha_2 < \alpha_3$  and  $\alpha_3 \le \frac{1}{2}\pi$  as linear

inequalities that maintain output voltage quarter wave symmetry [3.14-16]. The waveform of the line-to-line voltage  $v_{AB}$ , using solution 1, is shown in Figure 0.6(d).

Solution 2 (dotted line) in Figure 0.6(c) produces switching angles distributed over 90°, with  $\alpha_2 < \alpha_3$  located between 60° and 90°. The maximum modulation index with solution 2 is 1.16, which is lower than the modulation index produced by solution 1.

Additionally, it can be observed that solution 2 produces a fundamental voltage 180° out of phase with that produced by solution 1, for the same fundamental voltage. This can be observed in Figure 0.6(b) as during most of the first half cycle, the inverter output phase is connected to the negative bus. This observation may be critical for SHE implementation in grid connected applications. The arrangement for solution 2 always produces a smaller modulation index and hence the fundamental voltage. Therefore, the choice between the arrangements may require careful attention to the fundamental voltage polarity.



(a) Waveforms illustrate phase voltage relative to supply mid-point when the 5<sup>th</sup> and 7<sup>th</sup> harmonics are eliminated and the modulation index is  $m_a$ =1.1 (solution 1)



(b) Waveforms illustrate phase voltage relative to supply mid-point when the  $5^{th}$  and  $7^{th}$ 







(solid lines represent switching angles for solution 1 and dashed line for solution 2)



(d) Line-to-line voltage,  $v_{AB}$  waveform

Figure 0.6: SHE implementation using solution 1 and 2 when 5<sup>th</sup> and 7<sup>th</sup> harmonic components are eliminated.

## 3.5.1. Improved selective harmonic elimination

Based on the previous discussion, with the intention of extending the modulation index as the switching angle increases, this thesis investigates adjustment of the third harmonic to decouple the maximum modulation index achievable from a given number of switching angles. When a fourth angle is added for  $3^{rd}$  harmonic adjustment, the maximum modulation index using solution 1 remains at 1.1884 as shown in Figure 0.8(a). This is obtained when the  $3^{rd}$  harmonic voltage magnitude changes from  $\frac{1}{6}$  to  $\frac{1}{5}$  of the fundamental voltage. As illustrated in Figure 0.7(a) and (b), control of  $3^{rd}$ harmonic injection results in the spreading of switching angles over one quarter of the fundamental cycle, i.e.  $90^{\circ}$ .

Adjustment of the 3<sup>rd</sup> harmonic in SHE does not result in decoupling of the modulation index from the number of switching angles, however it remains useful at high modulation indices where standard SHE tends to produce switching angles in a narrow range close to zero (which cannot be realized practically). Injection of the 3<sup>rd</sup> harmonic into arrangements 1 and 2 forces their switching angles to converge, for the same modulation index, thus practical realisation is now possible(see Figure 0.8(b))[3.27].

For further confirmation of the potential usefulness of  $3^{rd}$  harmonic adjustment, the case with five notches where the  $5^{th}$ ,  $7^{th}$ , and  $11^{th}$  harmonics are eliminated, is presented in Figure 0.9(a). With adjustment of the  $3^{rd}$  harmonic, arrangements 1 and 2 produce one unique solution that satisfies quarter wave symmetry in the low modulation index range,  $m_a < 0.6$ . Above this modulation index, the switching angles produced by solving the equations for arrangement 1 diverge from those produced by arrangement 2 (see Figure 0.9(a)). When the  $5^{th}$ ,  $7^{th}$ ,  $11^{th}$ , and  $13^{th}$  harmonics are eliminated, the switching angles associated with the  $7^{th}$ ,  $11^{th}$  and  $13^{th}$  remain the same for both solutions over the entire modulation index range, while those associated with  $1^{st}$ ,  $3^{rd}$  and  $5^{th}$  diverge as modulation index increases, see Figure 0.9(b). The addition of the  $3^{rd}$  harmonic equation to the set of nonlinear transcendental equations contributes to simplification of the solution process as it is straightforward to find one set of initial guesses that can be used to produce the solutions for the entire modulation index linear range. A switching angle must be designated to control the third harmonic magnitude.

The modulation indices for the fundamental and 3<sup>rd</sup> harmonic voltages are defined as:

 $m_a = \frac{b_1}{\frac{1}{2}V_{dc}} = \frac{V_1}{\frac{1}{2}V_{dc}}$  and  $m_3 = \frac{b_3}{\frac{1}{2}V_{dc}} = \frac{V_3}{\frac{1}{2}V_{dc}} = \frac{\gamma V_1}{\frac{1}{2}V_{dc}}$ , where  $\gamma$  is a positive fraction used to describe

the 3<sup>rd</sup> harmonic voltage component in terms of the fundamental.



(b) Solution 2 (0 to  $90^{\circ}$ )

Figure 0.7: Waveform illustrate phase voltage relative to supply mid-point when the 5<sup>th</sup> and 7<sup>th</sup> harmonic are eliminated,  $m_a=1.1$  and 3<sup>rd</sup> harmonic is  $m_3=m/5$  for both SHE solutions.



(a) The dotted and solid lines represent solution 1 with and without the addition of 3<sup>rd</sup> harmonic component respectively. The maximum attainable modulation index is 1.1884.



(b) The dashed and solid lines represent solutions 1 and 2 with the addition of 3<sup>rd</sup> harmonic component, maximum attainable modulation index for both solutions is 1.185.

Figure 0.8: Switching angle trajectories for solution 1 and 2 when 5<sup>th</sup> and 7<sup>th</sup> harmonics are eliminated.



(a) Maximum attainable modulation index for solutions 1 and 2 when 5<sup>th</sup>, 7<sup>th</sup> and 11<sup>th</sup> harmonics are eliminatedare 1.1773 and 1.092 respectively.



(b) Maximum attainable modulation index for solutions 1 and 2 when5<sup>th</sup> , 7<sup>th</sup> , 11<sup>th</sup> , and 13<sup>th</sup>

harmonics are eliminated re 1.170 and 1.088 respectively.

Figure 0.9: Switching angle trajectories for solution 1 and 2 with control of the 3<sup>rd</sup> harmonic component. Solid and dashed lines represent solution 1 and 2 respectively.

# *A. Open-loop validation of the proposed SHE*

Figure 0.10 and 3.11 displays the two-level inverter output voltage and phase voltage waveforms obtained with SHE for solutions 1 and 2, with and without  $3^{rd}$  harmonic injection, using the DSP-hardware platform dSPACE1006. For simplicity of illustration, only two harmonics are eliminated ( $5^{th}$  and  $7^{th}$ ), with a modulation index  $m_a$ =1.1. Figure 0.10(b) shows that solution 2 fundamental voltage is 180° out of phase compared to solution 1 (see Figure 0.10(a)), for the same modulation index. This occurs because the switching angles for solution 2 are located in positions that result in the inverter output phase being connected to the negative bus during most of the first fundamental half cycle. Figure 0.11(a) to (b) display the output voltage waveforms for solutions 1 and 2 with adjustment of the  $3^{rd}$  harmonic ( $m_3 = \frac{1}{5}m$ ). The switching instants in both waveforms are approximately identical, but the fundamental voltage produce by solution 2 remains 180° out of phase as compared to solution 1. The adjustment of the  $3^{rd}$  harmonic (switching angles) that can be used with both arrangements to achieve the desired fundamental voltage and eliminate the  $5^{th}$  and  $7^{th}$ , with both satisfying the quarterwave symmetry constraint.



(a) Phase output voltage relative to supply mid-point (solution 1)





Figure 0.10: Experimental waveforms illustrating two-level inverter output voltage normalized by  $\frac{1}{2}V_{dc}$  for solutions 1 and 2, when the 5<sup>th</sup> and 7<sup>th</sup> harmonics are eliminated without the adjustment of 3<sup>rd</sup> harmonic component.



(a) Phase output voltage relative to supply mid-point (solution 1)



(b) Phase output voltage relative to dc supply mid-point (solution 2)

Figure 0.11: Experimental waveforms illustrating two-level inverter output voltage normalized by  $\frac{1}{2}V_{dc}$  for solutions 1 and 2, when the 5<sup>th</sup> and 7<sup>th</sup> harmonics are eliminated, and including adjustment of the 3<sup>rd</sup> harmonic component.

To further illustrate the usefulness of 3<sup>rd</sup> harmonic adjustment, Figure 0.12 shows the dSPACE1006 experimental results for voltage and current with elimination of the 5<sup>th</sup>, 7<sup>th</sup>, 11<sup>th</sup> and 13<sup>th</sup> harmonics using solution 1. These results are compared to standard SHE without 3<sup>rd</sup> harmonic component control. As more switching angles are added for both arrangements (standard and the proposed SHE), the quality of the line-to-line voltage and load current in Figure 0.12(f) and (h) are improved, compared to the waveforms in Figure 0.12(e) and (g) respectively. With a high modulation index of 1.1, the switching angles are increasingly wider using the proposed SHE (see Figure 0.12(b) and (d)), compared to standard SHE, as in Figure 0.12(a) and (c). A practical realisation is now possible without affecting the output quality.



Figure 0.12: Experimental voltage and current waveforms for two-level VSI using SHE solution 1, with and without adjustment of the 3<sup>rd</sup> harmonic component and 5<sup>th</sup>, 7<sup>th</sup>, 11<sup>th</sup> and 13<sup>th</sup> harmonics eliminated

 $(m_a=1.1).$ 

# *B. Open-loop validation of SHE with 9 angles*

Figure 0.13(a) and (b) illustrate the switching angle trajectories with nine notches per quarter cycle and phase voltage(normalized by  $\frac{1}{2}V_{dc}$ ) and its fundamental, for modulation  $m_a=1.1$ . The experimental waveforms of the voltage across switch S<sub>1</sub>, line-to-line voltage  $v_{AB}$ , and three-phase load current are presented in Figure 0.14(a) to (c), which further illustrate the suitability of using SHE for a grid connected VSI.

As the switching angle increases, SHE approaches high frequency carrier based PWM and SVM in terms of output voltage quality, but with a reduced number of switching transitions compared to the counterparts. This is significant in multi-megawatt mediumvoltage applications as nine notches per quarter cycle is equivalent to a 950Hz switching frequency, with relatively good output voltage and current waveform quality that cannot be achieved with carrier or space vector PWM with a similar switching frequency. The potential benefits of such an approach can be observed in terms of converter efficiency, which may not be significant in low power applications.



Figure 0.13: (a) Switching angle trajectories with nine notches per quarter fundamental cycle and (b) normalized PWM voltage waveform with reference to midpoint of dc bus.



 $(V_{dc}=100V, m_a=1.1, f_{sw}=950Hz, L_F=3mH \text{ and } C_F=40\mu\text{F}).$ 

#### **3.6.** Assessment of PWM modulation techniques

Figure 0.15 presents the pre-filter voltage spectra for all the modulation techniques discussed in this chapter. All simulation results use  $f_{sw}$ =2kHz and a modulation index of 0.9. From these results, SPWM is able to eliminate all harmonics less than or equal to  $2n\pm 1$ , where n is the number of pulses per half cycle of the sine wave. The inverter output voltage harmonics are pushed to the range around the carrier frequency and its odd multiples as shown in Figure 0.15(a). There are no low-order harmonics in the spectra. Both THIPWM and SVM contain similar harmonic distributions, as shown in Figure 0.15(b) and (c). The only exception is that SVM contains even-order harmonics such as the 2<sup>nd</sup>, 4<sup>th</sup>, 8<sup>th</sup> and 10<sup>th</sup> along with low-order odd-order harmonic contents. These even-order harmonics are not eliminated as they have very low magnitudes in the spectra, as illustrated in Figure 0.15(c). Most of the carrier harmonics and their sidebands in the phase voltage are suppressed around and beyond the second carrier component. Also THIPWM does not present low-order harmonics and the switching frequency component in the spectrum. Figure 0.15(d) shows SHE successfully eliminates the specified harmonics from the baseband (5<sup>th</sup>, 7<sup>th</sup>, 11<sup>th</sup>, 13<sup>th</sup>, 17<sup>th</sup>, 19<sup>th</sup>, 23<sup>rd</sup> and 25<sup>th</sup>). Whilst most of the harmonic energy is placed at the 29<sup>th</sup> and 31<sup>st</sup>, with the magnitude of the 29<sup>th</sup> harmonic being 74% of the fundamental. Although SHE is the best choice in eliminating specific harmonic components, it has the highest percentage THD of all presented modulation techniques. Figure 0.15(e) shows that the adjustment of  $3^{rd}$  harmonic injection in SHE improves the fundamental phase voltage by 15.5%, compared to traditional SHE. This leads to similar dc utilization as with THIPWM and SVM. The low-order harmonics are present in the spectrum since only the 5<sup>th</sup>, 7<sup>th</sup>, 11<sup>th</sup> and 13<sup>th</sup> harmonics are eliminated, as presented in Figure 0.15(e).

Table 3.2 and 3.3 list the output voltage and current harmonic contents with modulation indices of 0.9 and 0.5 for all the discussed modulation techniques.

Figure 0.16(a) and (b) present the relationships between modulation index and attainable fundamental phase voltage (pre-filter), and THD for SPWM, THIPWM, SVM, SHE and the improved SHE. The improved SHE is able to almost match SVM and THIPWM over the entire modulation index range, where all three modulation techniques achieve a maximum fundamental phase voltage equal to  $0.575V_{dc}$ . SPWM and SHE with nine angles, are able to reach a maximum fundamental voltage of only  $0.5V_{dc}$ , see Figure 0.16(a). The plot of THD versus modulation index in Figure 0.16(b) shows that SVM and THIPWM have the lowest THD, while the proposed SHE almost matches the plot. This is mainly because only four harmonic components are eliminated, therefore the THD is slightly higher for SVM and THIPWM. Standard SHE with nine angles has the highest THD of all the techniques.



Figure 0.15: Theoretical phase voltage harmonic spectrum of all PWM techniques.

		Harmonics (% of fundamental)										
Order of the harmonics		$1^{st}$	5 <sup>th</sup>	7 <sup>th</sup>	$11^{\text{th}}$	13 <sup>th</sup>	$17^{\text{th}}$	19 <sup>th</sup>	23 <sup>rd</sup>	25 <sup>th</sup>	27 <sup>th</sup>	29 <sup>th</sup>
	SPWM	100	0.09	0.26	0.12	0.04	0.19	0.11	0.25	0.15	0.18	0.21
	THIPWM	100	0.26	0.37	0.07	0.16	0.14	0.38	0.03	0.33	0.39	0.03
Voltage	SVM	100	0.25	0.32	0.22	0.15	0.53	0.14	0.44	0.17	0.39	0.36
	SHE	100	0.25	0.15	0.16	0.28	0.18	0.18	0.06	0.39	0.25	74.42
	Proposed SHE	100	0.31	0.22	0.12	0.28						

Table 0.2 (a) Harmonic content in voltage for all discussed modulation techniques ( $m_a=0.9$ ).

(b) Harmonic content in voltage for all discussed modulation techniques ( $m_a=0.5$ )

		Harmonics (% of fundamental)										
Order of the harmonics		$1^{st}$	5 <sup>th</sup>	7 <sup>th</sup>	$11^{\text{th}}$	13 <sup>th</sup>	$17^{\text{th}}$	19 <sup>th</sup>	23 <sup>rd</sup>	25 <sup>th</sup>	27 <sup>th</sup>	29 <sup>th</sup>
	SPWM	100	0.3	0.5	0.18	0.71	0.32	0.39	0.14	0.34	0.28	0.8
	THIPWM	100	0.09	0.25	0.37	0.77	0.46	0.13	0.9	0.14	0.02	0.41
Voltage	SVM	100	0.33	0.23	0.39	0.34	0.28	0.63	0.29	0.44	0.09	1.06
	SHE	100	0.02	0	0.62	0.57	0.09	0.05	1.15	0.8	0.56	111.91
	Proposed SHE	100	0.05	0.41	0.43	0.06						

Table 0.3(a) Harmonic content in current for all discussed modulation techniques ( $m_a=0.9$ )

		Harmonics (% of fundamental)							1			
Order of the harmonics		$1^{st}$	5 <sup>th</sup>	7 <sup>th</sup>	$11^{\text{th}}$	$13^{\text{th}}$	$17^{\text{th}}$	19 <sup>th</sup>	23 <sup>rd</sup>	25 <sup>th</sup>	27 <sup>th</sup>	29 <sup>th</sup>
	SPWM	100	0.05	0.44	0.05	0.01	0.05	0.02	0.05	0.03	0.03	0.03
	THIPWM	100	0.12	0.57	0.03	0.05	0.03	0.08	0.00	0.05	0.06	0.00
Current	SVM	100	0.12	0.5	0.1	0.05	0.13	0.03	0.08	0.03	0.06	0.05
	SHE	100	0.12	0.24	0.08	0.1	0.04	0.04	0.01	0.06	0.04	10.08
	Proposed SHE	100	0.56	0.19	0.05	0.09						

(b) Harmonic content in current for all discussed modulation techniques ( $m_a=0.5$ )

			Harmonics (% of fundamental)										
Order of the harmonics		1 <sup>st</sup>	5 <sup>th</sup>	7 <sup>th</sup>	11 <sup>th</sup>	13 <sup>th</sup>	17 <sup>th</sup>	19 <sup>th</sup>	23 <sup>rd</sup>	25 <sup>th</sup>	27 <sup>th</sup>	29 <sup>th</sup>	
	SPWM	100	0.14	0.81	0.09	0.25	0.08	0.08	0.02	0.05	0.04	0.11	
	THIPWM	100	0.04	0.4	0.17	0.27	0.11	0.03	0.16	0.02	0	0.06	
Current	SVM	100	0.09	0.35	0.09	0.06	0.17	0.18	0.09	0.07	0.04	0.11	
	SHE	100	0.01	0.01	0.29	0.2	0.02	0.01	0.2	0.13	0.08	15.16	
	Proposed SHE	100	0.08	0.36	0.17	0.05							



Figure 0.16: Comparison of PWM strategies with varying modulation index for (a) per-unit fundamental pre-filter inverter current and (b) total harmonic distortion (%THD).

	SPWM	THIPWM	SVM	SHE	Proposed SHE
DC voltage utilization, $V_{phase}/V_{dc}$	0.5	0.575	0.575	0.5	0.575
Harmonics	Medium	Medium	Worst	Low	Low
Digital implementation	Real-time	Real-time	Real-time	Look-up table	Look-up table

Table 0.4: Comparison summary of all presented modulation techniques.

# 3.7. Summary of major findings

This chapter discussed a number of modulation techniques that can be used to generate a train of pulses that have the same fundamental voltage as the reference voltage for a three phase two-level VSI. Table 0.4 summarises a comparison between these modulation techniques, with reference to several important features, namely dc rail utilization, switching losses, harmonic performance, and digital implementation.

The use of sinusoidal PWM does not fully utilize dc link voltage as stated in Table 0.4, thus lowering the power density and is discarded for future investigation in subsequent chapters. The disadvantage of SPWM however can be reduced by using THIPWM, which enables a 15.5% dc utilization improvement and can be easily implemented in real-time. Similarly, SVM also increase the dc link voltage and semiconductor utilization and improves the power density, compared to sinusoidal PWM, but may have difficulties operating in medium-voltage distribution systems where the possibility of unbalanced operation and ac faults is high. Digital implementation of these modulation strategies with grid connected PV inverters necessitates the use of a relatively high

switching frequencies for control purposes (the control period must be equal to the reciprocal of switching frequency). For medium voltage applications, strict grid codes and the use of a suitable modulation technique are the main priorities, therefore it is sometimes favourable to use traditional modulation approaches such as SHE and carrier based PWM over space vector modulation.

The possibility of using SHE was investigated to control high-power PV inverters in an attempt to suppress switching losses (by adopting switching frequencies of less than 1kHz) and improve the dc link voltage utilization, despite the requirement of an offline look-up table. To achieve these objectives, a new SHE implementation with adjustment of the 3<sup>rd</sup> harmonic magnitude in the phase voltage was introduced. The control of 3<sup>rd</sup> harmonic results in spreading of switching angles over the quarter of fundamental cycle. This feature can be used to simplify pratical realization of SHE at high modulation indices when a large number of harmonics are eliminated.

# References

- [3.1] A. M. Hava, R. J. Kerkman, and T. A. Lipo, "Simple analytical and graphical methods for carrier-based PWM-VSI drives," *Power Electronics, IEEE Transactions on*, vol. 14, pp. 49-61, 1999.
- [3.2] A. Iqbal and S. Moinuddin, "Comprehensive Relationship Between Carrier-Based PWM and Space Vector PWM in a Five-Phase VSI," *Power Electronics, IEEE Transactions on*, vol. 24, pp. 2379-2390, 2009.
- [3.3] J. Pou, J. Zaragoza, S. Ceballos, M. Saeedifard, and D. Boroyevich, "A Carrier-Based PWM Strategy With Zero-Sequence Voltage Injection for a Three-Level Neutral-Point-Clamped Converter," *Power Electronics, IEEE Transactions on*, vol. 27, pp. 642-651, 2012.
- [3.4] S. R. Bowes and D. Holliday, "Optimal Regular-Sampled PWM Inverter Control Techniques," *Industrial Electronics, IEEE Transactions on*, vol. 54, pp. 1547-1559, 2007.
- [3.5] A. Y. E. Lesan, M. L. Doumbia, and P. Sicard, "DSP-based sinusoidal PWM signal generation algorithm for three phase inverters," in *Electrical Power & Energy Conference (EPEC), 2009 IEEE*, pp. 1-6, 2009.
- [3.6] A. Iqbal, E. Levi, M. Jones, and S. N. Vukosavic, "Generalised Sinusoidal PWM with Harmonic Injection for Multi-Phase VSIs," in *Power Electronics Specialists Conference, 2006. PESC '06. 37th IEEE*, pp. 1-7, 2006.
- [3.7] Z. Keliang and W. Danwei, "Relationship between space-vector modulation and three-phase carrier-based PWM: a comprehensive analysis [three-phase inverters]," *Industrial Electronics, IEEE Transactions on,* vol. 49, pp. 186-196, 2002.

- [3.8] A. Iqbal, A. Lamine, I. Ashraf, and Mohibullah, "Matlab/Simulink Model of Space Vector PWM for Three-Phase Voltage Source Inverter," in Universities Power Engineering Conference, 2006. UPEC '06. Proceedings of the 41st International, pp. 1096-1100, 2006.
- [3.9] Y. Wenxi, H. Haibing, and L. Zhengyu, "Comparisons of Space-Vector Modulation and Carrier-Based Modulation of Multilevel Inverter," *Power Electronics, IEEE Transactions on,* vol. 23, pp. 45-51, 2008.
- [3.10] D. G. Holmes, "The significance of zero space vector placement for carrierbased PWM schemes," *Industry Applications, IEEE Transactions on*, vol. 32, pp. 1122-1129, 1996.
- [3.11] G. Narayanan, V. T. Ranganathan, D. Zhao, H. K. Krishnamurthy, and R. Ayyanar, "Space Vector Based Hybrid PWM Techniques for Reduced Current Ripple," *Industrial Electronics, IEEE Transactions on*, vol. 55, pp. 1614-1627, 2008.
- [3.12] Z. Di, V. S. S. P. K. Hari, G. Narayanan, and R. Ayyanar, "Space-Vector-Based Hybrid Pulsewidth Modulation Techniques for Reduced Harmonic Distortion and Switching Loss," *Power Electronics, IEEE Transactions on*, vol. 25, pp. 760-774, 2010.
- [3.13] D. Ahmadi, Z. Ke, L. Cong, H. Yi, and W. Jin, "A Universal Selective Harmonic Elimination Method for High-Power Inverters," *Power Electronics, IEEE Transactions on*, vol. 26, pp. 2743-2752, 2011.
- [3.14] V. G. Agelidis, A. I. Balouktsis, and C. Cossar, "On Attaining the Multiple Solutions of Selective Harmonic Elimination PWM Three-Level Waveforms Through Function Minimization," *Industrial Electronics, IEEE Transactions on*, vol. 55, pp. 996-1004, 2008.
- [3.15] M. S. A. Dahidah, V. G. Agelidis, and M. V. Rao, "On Abolishing Symmetry Requirements in the Formulation of a Five-Level Selective Harmonic Elimination Pulse-Width Modulation Technique," *Power Electronics, IEEE Transactions on*, vol. 21, pp. 1833-1837, 2006.
- [3.16] V. G. Agelidis, A. Balouktsis, I. Balouktsis, and C. Cossar, "Multiple sets of solutions for harmonic elimination PWM bipolar waveforms: analysis and experimental verification," *Power Electronics, IEEE Transactions on*, vol. 21, pp. 415-421, 2006.
- [3.17] P. N. Enjeti, P. D. Ziogas, and J. F. Lindsay, "Programmed PWM techniques to eliminate harmonics: a critical evaluation," *Industry Applications, IEEE Transactions on*, vol. 26, pp. 302-316, 1990.
- [3.18] F. Wanmin, D. Xiaoli, and W. Bin, "A Generalized Half-Wave Symmetry SHE-PWM Formulation for Multilevel Voltage Inverters," *Industrial Electronics*, *IEEE Transactions on*, vol. 57, pp. 3030-3038, 2010.
- [3.19] F. Wanmin, R. Xinbo, and W. Bin, "A Generalized Formulation of Quarter-Wave Symmetry SHE-PWM Problems for Multilevel Inverters," *Power Electronics, IEEE Transactions on*, vol. 24, pp. 1758-1766, 2009.
- [3.20] J. R. Espinoza, G. Joos, J. I. Guzman, L. A. Moran, and R. P. Burgos, "Selective harmonic elimination and current/voltage control in current/voltage-source topologies: a unified approach," *Industrial Electronics, IEEE Transactions on*, vol. 48, pp. 71-81, 2001.

- [3.21] J. I. Guzman, J. R. Espinoza, L. A. Moran, and G. Joos, "Selective Harmonic Elimination in Multimodule Three-Phase Current-Source Converters," *Power Electronics, IEEE Transactions on*, vol. 25, pp. 44-53, 2010.
- [3.22] V. Blasko, "A Novel Method for Selective Harmonic Elimination in Power Electronic Equipment," *Power Electronics, IEEE Transactions on*, vol. 22, pp. 223-228, 2007.
- [3.23] J. R. Wells, X. Geng, P. L. Chapman, P. T. Krein, and B. M. Nee, "Modulation-Based Harmonic Elimination," *Power Electronics, IEEE Transactions on*, vol. 22, pp. 336-340, 2007.
- [3.24] G. D. H. a. T. A. Lipo, *Pulse Width Modulation for Power Converters-Principle and Practice* Piscataway, NJ: Wiley-IEEE Press-Series on Power Engineering, 2003.
- [3.25] B.W.Williams. (1992). Power Electronics: Devices, Drivers, Application and Passive Components. 2nd edition.
- [3.26] B. Wu. (2006). *High Power Converters and AC Drives*.
- [3.27] S. Azmi, G. Adam, K. Ahmed, S. Finney, and B. Williams, "Grid Interfacing of Multi-Megawatt Photovoltaic Inverters," *Power Electronics, IEEE Transactions* on, vol. PP, pp. 1-1, 2012.

# **Chapter Four**

# Grid Interfacing of Voltage Source Inverter

#### 4.0. Introduction

A low loss multi-megawatt medium-voltage voltage source inverter (VSI) for a grid connected photovoltaic system is presented in this chapter. The suitability of using selective harmonic elimination (SHE) as the modulation technique and ability of the proposed system to meet utilities regulations, IEEE and IEC standards, are investigated. In an attempt to illustrate the potential superiority of SHE over carrier based or space vector pulse width modulation (PWM), this chapter demonstrates SHE allows a grid connected inverter to be controlled using a switching frequency of less than 1kHz, while the inverter is able to provide other necessary operation features such as independent control of active and reactive powers and operation control simplicity. For system validation, experimental results with SHE are compared to a inverter controlled using third harmonic injection PWM, with a 2kHz switching frequency. Furthermore, this thesis proposes a new implementation for SHE that utilizes the 3<sup>rd</sup> harmonics to spread the switching angles over 90° instead of them being located in a narrow range as generated when using conventional techniques, along with increased linear modulation index. The advantages of the proposed technique include implementation simplicity and flexibility in PWM waveforms. Simulations and experimentation to demonstrate the viability of SHE for grid connected inverters, are presented and discussed.

#### 4.1. Background

The use of voltage source inverters (VSIs) as interfacing units for grid integration of photovoltaic (PV) systems has been investigated intensively in the last decade. This includes the use of two-level and multilevel VSIs[4.1-3]. A multilevel VSI approach seems unsuitable for photovoltaic grid integration due to the following reason: multilevel inverters with a common dc link bus, such as diode and flying capacitor clamped inverters, exacerbate the problem of partial shading as they require a large number of series connected PV cells in order to build a relatively high dc voltage appropriate for medium-voltage applications[4.4-10]. Although the modularity of a cascaded multilevel inverter is attractive for PV systems [4.11, 12], it exposes the PV cells to a low frequency ac current component, and hence reduces PV lifetime [4.12-15]. The two-level VSI is widely used for grid connected PV systems, regardless of disadvantages such as high switching losses and relatively low quality output voltage waveforms[4.16]. Most grid connected PV systems presented in the literature are inappropriate for high-power application due to the use of a high switching frequency which leads to high switching losses and increased voltage stresses (dv/dt) on phase interfacing reactors and the transformer.

The current source option for PV grid interfacing has been advocated recently as a means of improving the effectiveness of maximum power point tracking (MPPT), as the number of series connected PV in each string is reduced[4.17, 18]. This reduces the impact of partial shading, and increases system efficiency by using one conversion stage instead of two stages, therein benefiting from the voltage boost capability of the current source inverter (lower dc link voltage and higher ac side voltage)[4.19-24]. This approach may not be viable for high-power medium-voltage PV systems, since current source inverters (CSI) usually have higher semiconductor losses than voltage. This is because the CSI can have twice the semiconductors in the conduction path at any instant, compared to the VSI[4.25-27].

With regard to the modulation strategies discussed in chapter three and grid codes and standards requirements, medium-voltage applications tend to favour the use of traditional modulation approaches such as selective harmonic elimination (SHE) and

carrier based pulse width modulation, over space vector modulation [4.28-31]. This chapter explores the possibility of using SHE to control high-power PV inverters in an attempt to suppress switching losses (by adopting switching frequencies less than 1kHz), improved dc link voltage utilization, and reduced phase interfacing reactor dv/dt [4.29, 30, 32, 33]. To facilitate the inverter control process, the low frequency output voltage switching waveforms produced using SHE are constructed using a relatively high sampling frequency, which is exclusively determined by the time required to execute the control algorithm in a DSP. The outcome of this investigation is substantiated using simulation and experimental results are compared to the case where the inverter is controlled using third harmonic injection PWM (THIPWM) that permits increased dc link voltage utilization [4.34].

All the grid connected inverter experiments discussed in this chapter use nine switching angles in an attempt to meet harmonic requirements at the point-of-common-coupling. Eight angles are used to eliminate non-triplen harmonics up to the 25<sup>th</sup> harmonic. An attractive feature that led to the adoption of SHE in this chapter is that the magnitude of non-triplen harmonics eliminated remain insignificant over the full modulation index linear range (0 to 1). Such a feature is not available in space-vector and carrier based PWM.

# 4.2. Operation of the VSI in island and grid modes

A standalone ac system may be described as one in which the entire ac power is delivered to the system through inverters. In island mode, there are no synchronous alternators present to provide a reference for the system frequency and voltage. All the system inverters need to be operated to provide a stable frequency and voltage in the presence of arbitrarily varying loads. Output voltage quality is the main factor to be considered for power quality.

On the other hand, in the grid connected mode, the system must control and handle disturbances from the network such as unbalanced or distorted grid voltages. The inverter module supplies the grid with the required controlled active and reactive powers. The power quality is determined by the current quantity as the voltage is not controlled by the PV unit.

# 4.2.1 Control system

The main target of the control system in island mode operation is to maintain the voltage and frequency within the standard specification limits. The control system must provide sufficient active and reactive power to match the connected load, therein providing high power quality.

The main target for grid-connected mode is to achieve good tracking of the controlled variables with minimal phase error. The tracking accuracy is constrained by the maximum allowable inverter switching frequency, the dc link voltage and the converter output inductance. Current control for grid connected inverters controls both the harmonic content and the power injected into the grid. Three main voltage and current control algorithms for the three-phase two-level VSI are briefly introduced.

# A. Stationary proportional-integral (PI) controller

Conventional PI control is simple and easy to implement. The transfer function is:

$$G(s) = K_P + \frac{K_I}{s} \tag{4.1}$$

The control variable is forced to track a time domain reference. The PI controller can achieve fast response and good harmonic rejection, when combined with a high inverter switching frequency. A compromise between reference tracking and stability is required in applications with a low switching frequency. However, the disadvantages of this controller are an inability to track a sinusoidal reference with zero steady-state error and poor disturbance rejection capability.

# B. Synchronous proportional-integral (PI) controller

Synchronous PI control is only applicable to three (multi) phase systems. The threephase ac quantities are transformed into dc components in the synchronous reference frame to ensure zero steady-state error, unlike in the stationery reference frame. The control transfer function is also given by equation (4.1). In the synchronous frame, stability is improved at the expense of response time and harmonic rejection. The technique must be extended to achieve good performance under distorted or unbalanced mains conditions. The disadvantage of this controller is the requirement for a phase looked loop (PLL) for transformations, hence the performance is dependent on the PLL characteristics.

## C. Proportional resonance (PR) controller

Proportional resonance (PR) control introduces an infinite gain at the fundamental frequency, and hence achieves zero steady-state error. The system is in the stationary frame with independence between phases. It can be used in single and three phase systems and the controller transfer function is:

$$G(s) = K_P + \frac{K_I s}{s^2 + \omega_o^2}$$
(4.2)

An infinite gain at the resonant frequency is not achievable in a digital system. Also, any slight displacement between the reference signal frequency and the measured signal resonant frequency leads to performance degradation. A modified PR controller is used to solve this problem, with a transfer function given by:

$$G(s) = K_P + \frac{2K_I\omega_c s}{s^2 + 2\omega_c s + \omega_o^2}$$
(4.3)

where  $\omega_c$  is the integrator low frequency cut-off.

For digital implementation, a discrete time transfer characteristic is obtained via a Tustin transformation. As with the synchronous controller, control is limited to the fundamental frequency component. Unlike synchronous frame control, the phases are individually controlled, which gives more robust performance in the presence of imbalance.

# 4.3. Analysis of AC dynamic equations for the three-phase two-level voltage source inverter in island mode

Figure 0.1 show the circuit diagram of three-phase two-level voltage source inverter operating in island mode. A proportional-resonance (PR) voltage regulator control loop is shown in Figure 0.2.



Figure 0.1: Circuit diagram of VSI in island mode.

A. Analysis of ac dynamic equations using proportional-resonance (PR) as a voltage regulator

Based on Figure 0.1, the dynamic equations that describe the VSI in the *abc* frame for island mode operation are:

$$i_{iabc} = i_{cabc} + i_{oabc}$$

$$\frac{dv_{oabc}}{dt} = \frac{1}{C_f} \left[ i_{iabc} - \frac{v_{oabc}}{R_o} \right]$$

$$(4.4)$$

$$(4.5)$$

where  $v_{oabc}$  and  $i_{iabc} = \gamma$  represent the output voltages and inverter currents, while  $L_f$ ,  $C_f$  and  $R_o$  are the ac power filters and load respectively.

The proportional-resonance (PR) controller transfer function used in this chapter is expressed as:

$$y = k_{pv}e + k_{iv}\frac{es}{s^2 + \omega_o^2}$$
(4.6)

where  $k_{pv}$  is the proportional gain,  $k_{iv}$  is the integral gain,  $e = v_{oabc}^* - v_{oabc}$  is the fundamental ac current error, and  $\omega_o$  is the fundamental angular frequency.

The transfer function of the PR controller is digitized for DSP implementation using the following derivations.

Let

$$z = k_{iv} \frac{es}{s^2 + \omega_o^2} \tag{4.7}$$

Rearranging (4.7) to:

$$sz + \frac{\omega_o^2}{s} z = k_{iv} e \tag{4.8}$$

and

$$w = \frac{\omega_o^2}{s} z \tag{4.9}$$

By taking the derivative of (4.8) and (4.9):

$$\frac{dw}{dt} = \omega_o^2 z \tag{4.10}$$

$$\frac{dz}{dt} = k_{iv}e - w \tag{4.11}$$

where z and w are auxiliary control variables used to facilitate the control design. From (4.6), (4.10) and (4.11), the output of the PR controller can be rewritten as:

$$y = k_{pv}e + \int \left(k_{iv}e - \int \omega_o^2 z dt\right) dt$$
(4.12)

Replacing  $i_{iabc}$  in (4.4) with  $\gamma$  and from (4.5), including the output voltage error into the PR controller, the value of  $\gamma$  is obtained from (4.12):

$$\gamma = k_{pv} \left( v_{oabc}^* - v_{oabc} \right) + \int \left( k_{iv} \left( v_{oabc}^* - v_{oabc} \right) - \int \omega_o^2 z_v dt \right) dt$$
(4.13)

Substituting (4.13) into (4.5)

$$\frac{dv_{oabc}}{dt} = -\frac{\left(1 + R_o k_p\right)}{R_o C_f} v_{oabc} + \frac{k_p v_{oabc}^*}{C_f} + \frac{z_v}{C_f}$$
(4.14)

where

$$z_{v} = \int \left( k_{iv} \left( v_{abco}^{*} - v_{abco} \right) - w_{v} \right) dt$$
(4.15)

and

$$w_v = \int \omega_o^2 z_v dt \tag{4.16}$$
By taking the derivatives of (4.15) and (4.16)

$$\frac{dz_{v}}{dt} = k_{iv} \left( v_{abco}^{*} - v_{abco} \right) - w_{v}$$

$$\frac{dw_{v}}{dt} = \omega_{o}^{2} z_{v}$$

$$(4.17)$$

From (4.14), (4.17) and (4.18), the state-space model of the voltage loop controller can be written in the form:  $\frac{dx(t)}{dt} = Ax(t) + Bu(t)$ 

$$\frac{d}{dt}\begin{bmatrix} v_{oabc} \\ z_{v} \\ w_{v} \end{bmatrix} = \begin{bmatrix} -\frac{\left(1+R_{o}k_{pv}\right)}{R_{o}C_{f}} & \frac{1}{C_{f}} & 0 \\ -k_{iv} & 0 & -1 \\ 0 & \omega_{o}^{2} & 0 \end{bmatrix} \begin{bmatrix} v_{oabc} \\ z_{v} \\ w_{v} \end{bmatrix} + \begin{bmatrix} \frac{k_{pv}}{C_{f}} \\ k_{iv} \\ 0 \end{bmatrix} v_{oabc}^{*}$$

$$0 \qquad (4.19)$$

$$V_{dc} + C_{dc} + C_{dc} + C_{dc} + C_{dc} + C_{f} + C_{f$$



Figure 0.2: Schematic diagram of the voltage control loop using a PR controller.

# B. Analysis of ac dynamic equations using proportional-integral (PI) as a voltage regulator

In the synchronous reference frame, the three-phase ac quantities are transformed into dc components to ensure any steady-state error is eliminated. Based on Figure 0.1, the inverter ac side dynamics in island mode can be expressed in the d-q axis reference frame as follows:

$$\frac{di_{cd}}{dt} = \frac{1}{L_f} \left[ v_{id} + \omega L_f i_{oq} - v_{od} \right]$$
(4.20)

$$\frac{di_{cq}}{dt} = \frac{1}{L_f} \left[ v_{iq} + \omega L_f i_{od} - v_{oq} \right]$$
(4.21)

$$\frac{dv_{od}}{dt} = \frac{1}{C_f} \left[ i_{id} - \frac{v_{od}}{R_o} + \omega C_f v_{oq} \right]$$
(4.22)

$$\frac{dv_{oq}}{dt} = \frac{1}{C_f} \left[ i_{iq} - \frac{v_{oq}}{R_o} - \omega C_f v_{od} \right]$$
(4.23)

Let  $U_d = i_{id} + \omega C_f v_{oq}$  and  $U_q = i_{iq} - \omega C_f v_{od}$ , therefore (4.22) and (4.23) can be written as:

$$\frac{dv_{od}}{dt} = -\frac{v_{od}}{R_o C_f} + \frac{U_d}{C_f}$$
(4.24)

$$\frac{dv_{oq}}{dt} = -\frac{v_{oq}}{R_o C_f} + \frac{U_q}{C_f}$$
(4.25)

The variables  $U_d$  and  $U_q$  can be obtained from the proportional-integral (PI) controller as follows:

$$U_{d} = k_{p} \left( v_{od}^{*} - v_{od} \right) + k_{i} \int \left( v_{od}^{*} - v_{od} \right) dt$$
(4.26)

$$U_{q} = k_{p} \left( v_{oq}^{*} - v_{oq} \right) + k_{i} \int \left( v_{oq}^{*} - v_{oq} \right) dt$$
(4.27)

Replacing the integral parts of (4.26)and (4.27), with  $Z_d$  and  $Z_q$ , the following sets of equations are obtained:

$$U_{d} = k_{p} \left( v_{od}^{*} - v_{od} \right) + Z_{d}$$
(4.28)

$$U_{q} = k_{p} \left( v_{oq}^{*} - v_{oq} \right) + Z_{q}$$
(4.29)

$$\frac{dv_{od}}{dt} = -\frac{\left(1 + R_o k_p\right)}{R_o C_f} v_{od} + \frac{k_p v_{od}^*}{C_f} + \frac{Z_d}{C_f}$$
(4.30)

$$\frac{dv_{oq}}{dt} = -\frac{\left(1 + R_o k_p\right)}{R_o C_f} v_{oq} + \frac{k_p v_{oq}^*}{C_f} + \frac{Z_q}{C_f}$$
(4.31)

$$\frac{dZ_d}{dt} = k_i \left( v_{od}^* - v_{od} \right) \tag{4.32}$$

$$\frac{dZ_q}{dt} = k_i \left( v_{oq}^* - v_{oq} \right) \tag{4.33}$$

Differential equations (4.30) to (4.33) can be written in state space form as:

$$\frac{d}{dt} \begin{bmatrix} v_{od} \\ v_{oq} \\ Z_{d} \\ Z_{q} \end{bmatrix} = \begin{bmatrix} -\frac{(1+R_{o}k_{p})}{R_{o}C_{f}} & 0 & \frac{1}{C_{f}} & 0 \\ 0 & -\frac{(1+R_{o}k_{p})}{R_{o}C_{f}} & 0 & \frac{1}{C_{f}} \\ -k_{i} & 0 & 0 & 0 \\ 0 & -k_{i} & 0 & 0 \end{bmatrix} \begin{bmatrix} v_{od} \\ v_{oq} \\ Z_{d} \\ Z_{q} \end{bmatrix} + \begin{bmatrix} \frac{k_{p}}{C_{f}} & 0 \\ 0 & \frac{k_{p}}{C_{f}} \\ k_{i} & 0 \\ 0 & k_{i} \end{bmatrix} \begin{bmatrix} v_{od} \\ v_{oq} \\ z_{d} \\ z_{q} \end{bmatrix}$$

(4.34)

$$i_{id} = U_d - \omega C_f v_{oq} = k_p \left( v_{od}^* - v_{od} \right) + k_i \int \left( v_{od}^* - v_{od} \right) dt - \omega C_f v_{oq}$$
(4.35)

$$i_{iq} = U_q + \omega C_f v_{od} = k_p \left( v_{oq}^* - v_{oq} \right) + k_i \int \left( v_{oq}^* - v_{oq} \right) dt + \omega C_f v_{od}$$
(4.36)

where the frequency  $\omega$  is from the PLL and is treated as an input.



Figure 0.3: Control system of a VSI using PI control in the dq-frame.

## 4.3.1. Simulation results of a three-phase VSI using PR control and PI control for island mode

The performance of a three-phase VSI using PR control and PI control for an island mode are investigated using Matlab/Simulink simulations. Both controllers are selected due to their ability to achieve zero steady-state error. The simulation results are based on the parameters in Table 0.1.

Parameters		V	alues
dc-link voltage	V <sub>dc</sub>	200	V
Nominal frequency	f	50	Hz
Switching frequency	$f_{sw}$	2000	Hz
Filter inductance per phase	$L_f$	5	mH
Filter capacitance per phase	$C_{f}$	30	μF
PI controller proportional gain	<i>k</i> <sub>p</sub>	0.5	
PI controller integral gain	<i>k</i> <sub>i</sub>	300	
Resonant controller proportional gain	k <sub>pv</sub>	10	
Resonant controller integral gain	k <sub>iv</sub>	1200	

Table 0.1: System parameters.

The VSI performances in island mode are investigated using a voltage-controlled PR controller and SPWM as the modulator, and are presented in Figure 0.4(a) to (d). The parameters used are listed in Table 0.1.



Figure 0.4: Simulation waveforms illustrating three-phase VSI controlled by PR controller in island mode.

The three-phase VSI output voltage is controlled using PI control in the dq-frame, with the VSI supplying active and reactive power for a balanced load. The controller performance is assessed using a 1kW step balanced resistive load change at 0.2s. Figure 0.5(a) presents the ability of the measured output voltage,  $V_d$ , to track the reference while  $V_q$  is maintained at 0V. The output active and reactive powers are shown in Figure 0.5(b). Figure 0.5(c) to (e) show the phase voltage, line-to-line voltage, and sinusoidal output voltage waveforms in a steady-state condition. A disturbance exists during the output voltage transient, see Figure 0.5(e).A balanced, sinusoidal output current for the resistive load following a step change from no load is shown in Figure 0.5(f), while a modulation index of nearly 0.9 is achieved in Figure 0.5(g).All three phases are controlled independently, unlike with synchronous dq-frame control, which is expected to give more robust performance for both balanced and unbalanced loads. From the simulation results, both controllers achieve zero steady-state error and obtain balanced, sinusoidal output voltages and currents for a resistive load.



Figure 0.5: Simulation waveforms when three-phase VSI is controlled by synchronous frame PI controller and SPWM as the modulator is in island mode.

### 4.3.2. Experimental verification of three-phase VSI using PR control and PI control for island mode

The corresponding experimental results using PR control in the *abc* and synchronous dq-frame PI control systems are shown in Figure 0.6 and 4.7 respectively. The reference output voltages,  $V_o$  are set to 90V and Figure 0.6(a) and 4.7(a) show that the measured voltages successfully track their respective references without any oscillatory and delay effects. In Figure 0.6(b) and 4.7(b), both controllers produce similar active and reactive power totals. The load currents and modulating signals for the VSIs operating in an island mode, using PR control, are shown in Figure 0.6(c) and (d). Figure 0.7(c) to (e) present the experimental output voltages, load currents and modulation index results when the VSI is controlled using PI control in the dq-reference frame.



Figure 0.6:dSPACE 1006 control desk experimental results of the three-phase VSI scaled-prototype connected to balanced loads when controlled using a PR controller in an *abc* frame.



Figure 0.7: dSPACE 1006 control desk experimental results of a three-phase VSI scaled-prototype connected to balanced loads when it is controlled using a synchronous *dq*-frame PI controller.

4.4. Analysis of ac-side dynamic equations of grid connected VSI using proportional-resonance control in *abc* coordinates



Figure 0.8: Schematic diagram of a grid-connected three-phase VSI.

Based on Figure 0.8, the differential equation describing the inverter ac-side dynamics, when considering only fundamental ac currents and voltages, is:

$$\frac{dI_{fabc}}{dt} = -\frac{R_f}{L_f} I_{fabc} + \frac{V_{con\_abc} - V_{fabc}}{L_f}$$
(4.37)

By setting  $Y = V_{abc\_con} - V_{fabc}$ , equation (4.37) can be written as:

$$\frac{dI_{fabc}}{dt} = -\frac{R_f}{L_f} I_{fabc} + \frac{Y}{L_f}$$
(4.38)

From (4.37)and (4.38), by feeding the grid current error to the PR controller, the value of Y is obtained from (4.12)

$$Y = K_{pi} \left( I_{fabc}^* - I_{fabc} \right) + \int \left( K_{ii} \left( I_{fabc}^* - I_{fabc} \right) - \int \omega_o^2 Z_i dt \right) dt$$
(4.39)

Substituting (4.39) and (4.37)

$$\frac{dI_f}{dt} = \frac{K_{pi}I_f^*}{L_f} - \frac{K_{pi}I_f}{L_f} - \frac{R_fI_f}{L_f} + \frac{Z_i}{L_f}$$
(4.40)

where

$$Z_i = \int \left( K_{ii} \left( I_f^* - I_f \right) - W_i \right) dt$$
(4.41)

and

$$W_i = \int \omega_o^2 Z_i dt \tag{4.42}$$

where  $K_{pi}$  is the current controller proportional gain and  $K_{ii}$  is the current controller integral gain.

By taking the derivative of (4.41) and (4.42):

$$\frac{dZ_i}{dt} = K_{ii} \left( I_f^* - I_f \right) - W_i \tag{4.43}$$

$$\frac{dW_i}{dt} = \omega_o^2 Z_i \tag{4.44}$$

From (4.40), (4.43) and (4.44), the state-space model of the current loop controller is:

$$\frac{d}{dt}\begin{bmatrix} I_{f} \\ Z_{i} \\ W_{i} \end{bmatrix} = \begin{bmatrix} -\frac{\left(R_{f} + K_{pi}\right)}{L_{f}} & \frac{1}{L_{f}} & 0 \\ -K_{ii} & 0 & -1 \\ 0 & \omega_{o}^{2} & 0 \end{bmatrix} \begin{bmatrix} I_{f} \\ Z_{i} \\ W_{i} \end{bmatrix} + \begin{bmatrix} \frac{K_{pi}}{L_{f}} \\ K_{ii} \\ 0 \end{bmatrix} I_{f}^{*}$$
(4.45)

From (4.45), the proportional-resonance control structure in *abc* coordinates used to regulate the output current and deliver the active and reactive powers into the ac grid, is illustrated in Figure 0.9.



Figure 0.9: Schematic diagram consists of PR control system in *abc* coordinates that is used to control the output current of grid-connected VSI.

Figure 0.10 shows the experimental results from the dSPACE 1006 DSP platform for a grid-connected VSI using the PR control system in Figure 0.9. Figure 0.10(a) to (c) show the ability of the output currents to track the 5A references during transient and steady-state conditions. It is confirmed that the output currents successfully track the references while delivering active power of 600W at unity power factor, as demonstrated in Figure 0.10(d). The ac grid voltage is shown in Figure 0.10(e).



(a) Output phase current with their respective references



(d) Active and reactive power exchange (e) Grid voltage with grid voltage

Figure 0.10: Experimental results showing the performance of a VSI injecting current into the grid.

### 4.5. Analysis of AC side dynamic equation of a grid-connected VSI using proportional-integral control in the *dq*-frame

This section discusses the control systems of the grid connected inverter, and discusses the control variables that could be manipulated from a power electronic systems standpoint, including most of the possible physical limitations imposed by the ac side passive elements which are not well documented. A generic phasor diagram and P-Q envelope are presented that illustrate the operation of grid connected converters.

The active and reactive powers are regulated using direct power control, where the filter bus voltage is assumed constant. Inverter power injection into the grid is regulated in the synchronous reference frame, where the voltage magnitude at the filter bus is aligned with the *d*-axis (the reference for the whole system, in other words all measurements in the *d*-*q* axes will be relative to the filter bus voltage). Based on this principle and Figure 0.8, the inverter ac side dynamics can be expressed in the *d*-*q* axis reference frame as follows:

$$\frac{dI_{fd}}{dt} = -\frac{R_f}{L_f}I_{fd} + \frac{\left\lfloor V_{cd} - V_{fd} + \omega L_f I_{fq} \right\rfloor}{L_f}$$

$$\frac{dI_{fq}}{dt} = -\frac{R_f}{L_f}I_{fq} + \frac{\left\lfloor V_{cq} - V_{fq} - \omega L_f I_{fd} \right\rfloor}{L_f}$$

$$(4.46)$$

Let  $Q_d = V_{cd} - V_{fd} + \omega L_f I_{fq}$  and  $Q_q = V_{cq} - V_{fq} - \omega L_f I_{fd}$ , therefore (4.46) and (4.47) can be written as:

$$\frac{dI_{fd}}{dt} = -\frac{R_f}{L_f}I_{fd} + \frac{Q_d}{L_f}$$
(4.48)

$$\frac{dI_{fq}}{dt} = -\frac{R_f}{L_f} I_{fq} + \frac{Q_q}{L_f}$$
(4.49)

Variables  $Q_d$  and  $Q_q$  can be obtained from the proportional-integral (PI) controller as follows:

$$Q_{d} = K_{P}(I_{fd}^{*} - I_{fd}) + K_{I} \int (I_{fd}^{*} - I_{fd}) dt$$
(4.50)

$$Q_{q} = K_{P}(I_{fq}^{*} - I_{fq}) + K_{I} \int (I_{fq}^{*} - I_{fq}) dt$$
(4.51)

Replacing the integral parts of (4.50) and (4.51) with  $W_1$  and  $W_2$ , the following sets of equations are obtained:

$$Q_d = K_P (I_{fd}^* - I_{fd}) + W_1$$
(4.52)

$$Q_{q} = K_{P}(I_{fq}^{*} - I_{fq}) + W_{2}$$
(4.53)

$$\frac{dI_{fd}}{dt} = -\frac{\left(R_f + K_P\right)}{L_f}I_{fd} + \frac{W_1}{L_f} + \frac{K_P}{L}I_{fd}^*$$
(4.54)

$$\frac{dI_{fq}}{dt} = -\frac{\left(R_f + K_p\right)}{L_f}I_{fq} + \frac{W_2}{L_f} + \frac{K_p}{L}I_{fq}^*$$
(4.55)

$$\frac{dW_1}{dt} = K_I (I_{fd}^* - I_{fd})$$
(4.56)

$$\frac{dW_2}{dt} = K_I (I_{fq}^* - I_{fq})$$
(4.57)

Differential equations (4.54) to (4.57) can be written in state space form as:

$$\begin{bmatrix} \frac{dI_{fd}}{dt} \\ \frac{dI_{fq}}{dt} \\ \frac{dW_1}{dt} \\ \frac{dW_2}{dt} \\ \frac{dW_2}{dt} \end{bmatrix} = \begin{bmatrix} -\frac{\left(R_f + K_p\right)}{L_f} & 0 & \frac{1}{L_f} & 0 \\ 0 & -\frac{\left(R_f + K_p\right)}{L_f} & 0 & \frac{1}{L_f} \\ -K_I & 0 & 0 & 0 \\ 0 & -K_I & 0 & 0 \end{bmatrix} \begin{bmatrix} I_{fd} \\ W_1 \\ W_2 \end{bmatrix} + \begin{bmatrix} \frac{K_p}{L_f} & 0 \\ 0 & \frac{K_p}{L_f} \\ K_I & 0 \\ 0 & K_I \end{bmatrix} \begin{bmatrix} I_{fd}^* \\ I_{fq}^* \end{bmatrix}$$
(4.58)

The Eigen-values of grid current controller is computed in (4.58). These Eigen-values represent the poles of the entire system with all controllers incorporated. It means that this approach take into account all the potential interaction between the cascaded loops. These Eigen-values hold important information such as damping factor, settling time, natural frequency and oscillation frequency. On this basis, all controller theoretical gains can be selected and adjusted accordingly to ensure all Eigen-values (poles) have negative parts, which an indication of stable system. The imaginary part of these Eigen-values represent the oscillation frequencies.

The *d-q* voltage components at the converter, which represent the voltage references to the modulator, can be estimated from the expressions for  $Q_d$  and  $Q_q$ , taking into account feed-forward terms as follows [4.35, 36]:

$$V_{cd} = Q_d + V_{fd} - \omega L_f I_{fq}$$
(4.59)

$$V_{cq} = Q_q + V_{fq} + \omega L_f I_{fd}$$

$$\tag{4.60}$$

The current references are calculated directly from the desired active and reactive power set points as follows:

$$\begin{bmatrix} I_{fd}^{*} \\ I_{fq}^{*} \end{bmatrix} = \frac{2}{3} \frac{1}{V_{fd}^{2} + V_{fq}^{2}} \begin{bmatrix} V_{fd} & V_{fq} \\ V_{fq} & -V_{fd} \end{bmatrix} \begin{bmatrix} P^{*} \\ Q^{*} \end{bmatrix}$$
(4.61)

In PV applications,  $P^*$  can be obtained from the maximum power tracking error. A dc voltage controller that adjusts the PV cells operating point in order to maximize power extraction is used.

Figure 0.11 summarizes the grid connected inverter control systems used in this research, including the SHE implementation and THI pulse width modulation. Figure 0.12(a) shows the generic phasor diagram for the voltage source grid connected inverter. This phasor diagram illustrates the control principle of the VSI, including operating regions and boundaries in steady-state. When the voltage vector at the converter terminal  $\vec{V}_c$  is in advance of the filter bus  $\vec{V}_f$ , the power flow is from the dc side to the ac network, and vice versa. When the voltage magnitude at converter terminal  $V_c$  is larger than at the filter bus  $V_f$ , where power injection into the grid is controlled, reactive power flows from the converter to the ac network (provision of leading VAr). Therefore converter active and reactive powers can be manipulated using the load angle  $\delta$  and voltage magnitude at the converter terminal,  $V_c$ .

In VSIs, the phase interfacing inductor  $L_f$  is not only required for filtering and fault purposes, and is also necessary to provide phase shift  $\delta$  to enable active power control. Also, it has to be sized appropriately not to limit the converter's ability to generate reactive power for a given dc link voltage. In this chapter the interfacing inductor  $L_f$  is sized so as to ensure the converter is able to provide its maximum design value of reactive power. The selection criterion for  $L_f$  is described as follows: as  $R_f$  is normally small compared to the reactance of the phase interfacing reactor-resistance can be neglected. Therefore, the steady-state  $I_{fd}$  and  $I_{fq}$  can be obtained from equations (4.46) and (4.47) by setting the derivative terms to zero, and  $V_{fq}=0$  since the voltage vector is aligned with the *d*-axis:

$$I_{fd} = \frac{V_{cq}}{\omega L_f} \tag{4.62}$$

$$I_{fq} = -\frac{(V_{cd} - V_{fd})}{\omega L_f}$$
(4.63)

Active and reactive powers P and Q at the filter bus can be expressed as:

$$P = \frac{3}{2} V_{fd} I_{fd} = \frac{3}{2} \frac{V_{cq} V_{fd}}{\omega L_f}$$
(4.64)

$$Q = -\frac{3}{2} V_{fd} I_{fq} = \frac{3}{2} \frac{V_{cd} V_{fd}}{\omega L_f} - \frac{3}{2} \frac{V_{fd}^2}{\omega L_f}$$
(4.65)

Equation (4.64) can be rewritten as shown in (4.66):

$$P + \frac{3}{2} \frac{V_{cd}^2}{\omega L_f} = \frac{3}{2} \frac{V_{cd} V_{fd}}{\omega L_f}$$
(4.66)

The converter active and reactive power capability curve (*P-Q* envelope) can be obtained by squaring, then adding, (4.66) and(4.65). The voltage at converter terminal is  $V_c^2 = V_{cd}^2 + V_{cq}^2$ :

$$P^{2} + (Q + \frac{3}{2} \frac{V_{a}^{2}}{\omega L_{f}})^{2} = (\frac{3}{2} \frac{V_{c}V_{fd}}{\omega L_{f}})^{2}$$
(4.67)

Figure 0.12(b) depicts the converter's active and reactive power capability curve. The upper and lower of active-power axis (*P*) represent the over-excitation and underexcitation operating regions, where the converter releases and absorbs reactive energy to and from the ac grid. For medium-voltage where the PV inverter has to provide reactive power support to the ac network,  $L_f$  must be sized based on (4.67), assuming maximum active and reactive power capabilities. Therefore equation (4.67) can rewritten as:

$$X^{2}P_{\max}^{2} + (XQ_{\max} + \frac{3}{2}V_{fd}^{2})^{2} - (\frac{3}{2}V_{c}V_{fd})^{2} = 0$$
(4.68)
where  $X = \omega L$ .

For an inverter with  $V_{dc}$ =300V,  $P_{max}$ =1000W and  $Q_{max}$ =±663VAr, connected to a 110V ac network, the maximum interfacing inductance obtained after solving (4.68) for X is  $L_f \leq 36$ mH, assuming  $V_c = \frac{1}{2}m_a V_{dc} = 0.575 V_{dc}$ . For reactive power applications, such as a STATCOM [4.30], the phase interfacing inductor  $L_f$  can be selected based on maximum converter reactive power capability  $Q_{max}$  when P=0. Therefore, equation (4.67)reduces to:

$$Q_{\max} = \frac{3}{2} \frac{V_{fd}}{\omega L_f} (V_c - V_{fd})$$
(4.69)

Assuming the same inverter parameters:

$$L_{f} \leq \frac{3}{2} \frac{V_{fd}(V_{c} - V_{fd})}{\omega Q_{\max}} \leq \frac{3}{2} \times \frac{155 \times (173 - 155)}{2 \times \pi \times 50 \times 663} \leq 21 \text{mH}.$$
(4.70)

This discussion is intended to show that sizing of the phase interfacing reactors for grid connected inverters is not merely based on filter design as found in many papers in the literature. Sizing has to take into consideration the converter capability curve as demonstrated in this chapter. The capacitor  $C_f$  is sized here to place the low-pass filter cut-off frequency between 150Hz and 200Hz. Despite this approach being commonly used for low power applications, it is not appreciated in high-power inverters that are connected to medium and high voltages. Instead, tuned filters target the significant switching frequency harmonic components and their sidebands that are not cancelled in the line-to-line voltage.



Figure 0.11: Schematic diagram of a grid connected PV inverter with SHE and THI-PWM using a dSPACE1006 hardware implementation.



Figure 0.12: Voltage source converter (a) generic phasor diagram and (b) theoretical capability curve.

#### 4.6. Simulation results for a VSI in a medium voltage photovoltaic application

This section presents simulation results for a medium-voltage PV-inverter when controlled with SHE and sinusoidal PWM with triplen harmonic injection. These results aim to show that SHE can be used instead of carrier based PWM strategies to control grid connected PV inverters, whilst maintaining all of the control features achieved with carrier based strategies. Table 0.2 shows the simulation parameters used in this chapter. Figure 0.13 to Figure 0.17 show current, voltage and power waveforms obtained with SHE and THIPWM when exchanging active and reactive power. The results presented in Figure 0.13 and 4.8 are obtained when the active power current component  $I_d$  is varied from 0 to 1.0 pu (25MVA and 11kV based). SHE produces a similar performance to that of THIPWM at approximately half the switching frequency, including systems dynamics, and ac voltage and current waveform quality. Similar SHE performances are seen in Figure 0.15 and for THIPWM in Figure 0.16 when the inverter operates at full active power, 0.8 power factor lagging. VSI operation at zero power factor is presented in Figure 0.17. For further illustration of the potential reductions achievable in switching losses with SHE, Table 0.3 and 4.4 are presented. Table 0.3 presents the IGBT parameters used in the switching loss calculations, based on equations (4.71) and (4.72)[4.37, 38]:

$$P_{sw_{IGBT}} = \frac{f_{sw}}{4\pi} \int_{\varphi}^{\pi+\varphi} \left( \left( k_{on1} + k_{on2} \right) \cdot I_m \sin(\omega t - \varphi) + k_{off1} + k_{off2} \right) d\omega t$$
  
$$= \frac{f_{sw} \left( k_{on1} + k_{on2} \right) \cdot I_m}{2\pi} + \frac{f_{sw}}{4} \left( k_{off1} + k_{off2} \right)$$
(4.71)

$$P_{sw\_diode} = \frac{f_{sw}}{4\pi} \int_{\varphi+\pi}^{2\pi+\varphi} \left( k_{D1} \cdot I_m \sin(\omega t - \varphi) + k_{D2} \right) d\omega t$$
  
$$= \frac{f_{sw} \cdot k_{D1} \cdot I_m}{2\pi} + \frac{f_{sw}}{4} k_{D2}$$
(4.72)

Table 0.4 summarises the inverter switching losses for three operating conditions. Based on these results it can be concluded that SHE produces lower losses than THIPWM, for similar operating conditions.

Parameters	Values					
Converter power rating (MVA)	Converter power rating (MVA)					
DC link voltage (kV)		20				
AC voltage (line-to-line in kV)		11				
Active power capability (MW)		20				
Reactive power capability (MVA	15					
Grid frequency (Hz)	50					
Switching frequency (kHz)	THIPWM	2				
Switching frequency (KHZ)	SHE	0.95				
Current control proportional	THIPWM	70				
gain, $K_p$	SHE	220				
Current control integral gain,	THIPWM	3000				
K <sub>i</sub>	SHE	5000				

Table 0.2: Simulation parameters of a multi-megawatt PV inverter.

Table 0.3: Converter switching device data used for switching loss estimation, including turn-on, turn-off, and diode recovery losses, based on IGBT MG1200FXF1US53.

R <sub>on</sub> (IGBT)	$1.75 \mathrm{m}\Omega$	Konl	0.0016J/A
R <sub>on</sub> (diode)	1.64mΩ	K <sub>on2</sub>	0.029J
$V_0$ (IGBT) or $V_T$	1.2V	K <sub>off1</sub>	0.0017J/A
V <sub>0</sub> (diode) or V <sub>F</sub>	1.05V	K <sub>off2</sub>	0.35J
IGBT rating	3.3kV	K <sub>D1</sub>	0.0006J/A
Number of series devices per arm	14	K <sub>D2</sub>	0.398J

Table 0.4: Summary of switching losses with THIPWM and SHE modulation strategies for a 25MVA PV-inverter.

Operating condition	P=20MW at 0.8 power factor lagging	Q=15MVAr at zero power factor	20MW at unity power factor
SHE	190.6 kW	190.2kW	88.7kW
THIPWM	401.2kW	400.4kW	401.2kW





(e) Modulation index with variation of active power

Figure 0.13: Simulation waveforms demonstrating operation of a high-power medium-voltage grid connected PV-inverter when controlled using SHE-PWM, with a 950Hz switching frequency, 20kV dc link voltage, and connected to a 11kV ac grid (step change in active power).



Figure 0.14: Simulation waveforms demonstrating operation of a high-power medium-voltage grid connected PV-inverter when controlled using THI-PWM, with a 2kHz switching frequency, 20kV dc link voltage, and connected to a 11kV ac grid (step change in active power).

#### B. 20MW at 0.8 power factor lagging







Figure 0.16: Simulation waveforms demonstrating operation of grid connected PV-inverter when controlled using THIPWM during rated power of 20MW, 0.8 power factor lagging.

#### C. 15MVAr at zero power factor



Figure 0.17: Simulation waveforms demonstrating operation of grid connected PV-inverter when controlled using THIPWM during rated power of 15MVAr, zero power factor lagging.

#### 4.7. Experimental results of a low-loss grid-connected VSI

The experimental scaled-down test-rig results in this section aim to demonstrate the superiority of low switching frequency SHE over carrier based or space vector PWM strategies for grid connected two-level three phase VSIs. Third harmonic injection PWM is used as representative of high frequency carrier based and space vector PWM strategies. All the experimental results presented are based on parameters summarised in Table 4.5 and taken from the dSPACE1006 control desk.

Figure 4.18 and 4.19 summarises the practical implementation of the grid connected PV inverter using SHE and THIPWM. Modulation and control algorithms are implemented in dSPACE1006, which uses Matlab/Simulink and Real-Time Workshop as a front-end programming platform. The experimental results demonstrate the potential benefits of adopting SHE for PV inverters instead of commonly used carrier based PWM strategies in various grid mode conditions.



Figure 0.18: Hardware diagram of the test-rig



Figure 0.19: per-unit scaled-down test rig of a multi-megawatt PV inverter prototype.

Parameter	THIPWM	SHE-PWM
Two-level converter rating (VA)	1200	1200
DC link voltage (V)	300	300
AC grid voltage per phase (V)	110	110
Active power capability (W)	1000	1000
Reactive power capability (VAr)	663	663
Interfacing inductor per phase (mH)	10	10
Filter capacitor (µF)	40	40
Grid frequency (Hz)	50	50
Switching frequency (kHz)	2	0.95
Proportional gain, $K_P$	4	120
Integral gain, $K_I$	1200	5000
Natural frequency, $\omega_n$ (rad/s)	346	707.1
Transformer turns ratio	1:1	1:1

Table 0.5: Parameters of per-unit scaled-down test rig for prototype of megawatt PV inverter.

#### A. Step change in active power

Figure 0.20 illustrates the results when  $I_d$  is varied from 2A to 6A while  $I_q$  is zero in order to achieve unity power factor using sinusoidal PWM with triplen harmonic injection as the modulator. Figure 0.20(a) shows that the measured currents,  $I_d$  and  $I_q$  are able to track their respective reference currents with a short transient response time. In Figure 0.20(b), the active power demand exchanges from 300W to near rated power of 960W, with respect to the ac grid voltage. The quality of injected current into the grid in Figure 0.20(c) can be further improved by increasing the order of the ac power filter and increasing the witching frequency. Figure 0.20(d) presents an ac grid voltage where the measurement is taken at the filter bus (output) to the interfacing transformer. The in-

phase voltage and current waveforms in Figure 0.20(e) demonstrate unity power factor. In this application, the modulation index is varied from 0.85 to 0.95 following the variation of active and reactive currents when the PV inverter operates at full load.



Figure 0.20: Waveforms demonstrating the operation of a PV inverter using sinusoidal PWM with triplen harmonic injection, when  $I_d$  is varied from 2A to 6A.

Figure 0.21 shows the experimental results when the grid connected inverter is controlled using SHE, with a reactive current component  $I_{qref}$ =4A, and a step change from 2A to 6A is applied to the active power current component  $I_{dref}$ . Figure 0.21(a) shows  $I_d$  and  $I_q$  are able to track their references when the inverter operates close to rated power, when SHE is used. Figure 0.21(b) shows active and reactive powers the inverter exchanges with the grid. Figure 0.21(c) and (d) show the good quality current

and voltage waveforms the inverter presents to the grid using SHE with a switching frequency of less than 1kHz. Figure 0.21(e) shows modulation index variation as  $I_{dref}$  is varied from 2A to 6A.



Figure 0.21: Experimental waveforms demonstrating operation of the grid connected inverter using SHE-PWM (step change in active power).

#### B. Step change in reactive power

Similar results illustrating inverter operation when  $I_{qref}$  is varied from 0A to 4A and  $I_{dref}$  remains at 6A using THIPWM, are presented in Figure 0.22. The inverter is able to operate satisfactorily, with proper tracking of the set points as demonstrated in Figure 0.22(a). The active and reactive power exchanges with the ac grid voltage and variation of modulation index with respect to changes in reactive power are presented in Figure 0.22(b) and (c) respectively. Figure 0.23 shows the inverter operating results using SHE during variation of the reactive power.



Figure 0.22: Experimental waveforms demonstrating PV inverter using sinusoidal PWM with triplen harmonic injection, when  $I_q$  is varied from 0 to 4A while  $I_d=2A$ .



Figure 0.23: Experimental waveforms demonstrating operation of a grid connected inverter using SHE-PWM (step change in reactive power).

Figure 0.20 and 4.22 are for the case when the inverter is controlled using THI-PWM, with a 2kHz switching frequency, where  $I_{dref}$  is varied from 2A to 6A, and  $I_{qref}$  is varied from 0A to 4A. These results show the inverter under consideration stably exchanges active and reactive powers with the grid and presents high quality ac current and voltage waveforms at the point of common coupling (PCC). The output waveform quality in this case is similar to that obtained with SHE as demonstrated in Figure 0.21 and 4.23, despite THIPWM using a switching frequency approximately twice that used with SHE. Figure 0.20(f), 4.21(e), 4.22(c), and 4.23(c) confirm the variation of modulation index given in the generic phasor diagram for the voltage source converter in Figure 0.12(b), regarding increased modulation index margin for reactive power support.

Table 0.6 and 4.7 show the individual current and voltage harmonics for SHE and THI-PWM. Both modulation approaches are able to meet IEEE Standard 519-1922 and IEC 61000-3-6 for medium voltages [4.39, 40], and IEEE 1574 and IEC61727 for low voltage applications in terms of individual current and voltage harmonic limits. These results show SHE is able to meet strict harmonic requirements at the PCC, at a reduced switching frequency, such as 950Hz. Table 0.8 and 4.9 present the switching device parameters used for the 1200VA pu scaled PV-inverter prototype and the estimated switching losses obtained for the two operating conditions. The switching losses are halved with SHE, compared to those with THIPWM. Based on these results it can be concluded that SHE can be considered for low loss large-scale PV inverters required for a sustainable low carbon economy as it is able to provide the flexibility of carrier based and space vector PWM strategies, including fast dynamic response at reduced losses. The SHE implementation for grid connected inverter is simple and exploits modern DSP floating point capabilities.

Table 0.6: Harmonic content in the experimental current for SHE-PWM and THI-PWM inverters.

		Harmonics (% of fundamental)								
Order of t	the harmonics	1 <sup>st</sup>	5 <sup>th</sup>	7 <sup>th</sup>	11 <sup>th</sup>	13 <sup>th</sup>	17 <sup>th</sup>	19 <sup>th</sup>	23 <sup>rd</sup>	25 <sup>th</sup>
Current	SHE-PWM	100.00	0.3	0.3	0.2	0.15	0.01	0.1	0.03	0.01
Current	THI-PWM	100.00	0.35	0.38	0.16	0.19	0.02	0.09	0.06	0.02

Table 0.7: Harmonic content in the experimental voltage for SHE-PWM and THI-PWM inverters.

		Harmonics (% of fundamental)								
Order of t	the harmonics	$1^{st}$	5 <sup>th</sup>	7 <sup>th</sup>	11 <sup>th</sup>	13 <sup>th</sup>	17 <sup>th</sup>	19 <sup>th</sup>	23 <sup>rd</sup>	25 <sup>th</sup>
Voltago	SHE-PWM	100.00	2.1	1.54	1.1	0.9	0.5	0.05	0.03	0.02
Voltage	THI-PWM	100.00	2	1.5	0.96	0.8	0.46	0.03	0.01	0.01

Table 0.8: Converter switching device data used for switching loss estimation, including turn-on, turn-off, and diode recovery losses, based on the IGBTIRG4PH50KDbF.

IGBT rating	1.2kV
K <sub>1</sub>	0.44mJ/A
K <sub>2</sub>	0

Table 0.9: Summary of switching losses with THIPWM and SHE modulation strategies for a 1200VA PV-inverter.

Operating condition	P=878.6W and Q at 0.83 power factor lagging	P=913W at unity power factor
SHE	18.1 W	15.04W
THI-PWM	38.1W	31.7W

#### 4.8. Typical application of VSI for grid connection of PV arrays

This section presents a case study illustrating a typical VSI application for grid integration of PV arrays, when a maximum power tracking system is incorporated. In this mode, the inverter dc link voltage is adjusted as the weather changes in an attempt to ensure maximum power extraction. This has been simulated with the aid of a lookup table which contains a pre-defined PV *I-V* characteristic. Weather condition variation is simulated by variable current from the PV arrays, which is the assumed input signal to the lookup table. In this case study, the VSI dc voltage regulator is used to track the maximum power point (MPP), as previously described. For converter protection against over-current and to ensure sinusoidal output current, an inner current controller based on a synchronous reference frame PI controller is incorporated. Under this control mode, all the power generated by the PV will be injected into the ac grid, without creating active power imbalance between the inverter ac and dc sides [4.41-44]. Figure 0.24 summarises the typical control system of the PV inverter adopted in this chapter.



Figure 0.24: Schematic diagram representing the control structure which consists of a dc-link voltage regulator and an output current regulator.

For performance demonstration of the grid-connected VSI under varying weather conditions, the test system shown in Figure 0.24 is simulated with the following parameters: the PV array is modelled as current source that varies its current injection  $(I_{pv})$  with weather, and in this case  $I_{pv}$  is varied from 50A to 60A; dc-link capacitor=8mF; converter apparent rated power=100kVA; rated ac grid line voltage=380V; switching frequency=4kHz; and an LC-filter with 5mH and 40µF inductance and capacitance respectively.

Simulation results are displayed in Figure 0.25. The maximum power point tracker varies the inverter dc-link voltage from 1179V to 1188V as the PV array's output current  $I_{pv}$  varies with weather as shown in Figure 0.25 (a) and (b). A unity power factor is maintained by the VSI at the point of common coupling, as shown in Figure 0.25(d), independent of the PV operating point. The active and reactive powers the VSI exchanges with the grid are presented in Figure 0.25(e).

To further illustrate the possibility of the continual support to ac grid with low or no radiation from the sun, the PV array output power is reduced to near zero at t=1s while the VSI reactive power exchange with the ac grid is increased at t=2s. Results obtained are displayed in Figure 0.26. These results demonstrated that the PV inverter can inject active power into grid during the day and can operate as a typical static synchronous compensator during the night, providing reactive power support to the grid.



Figure 0.25: Simulation results of PV system behaviour in response to a step-change in PV arrays output current using three-phase voltage source inverter.



Figure 0.26: Simulation results showing the performance of the grid-connected photovoltaic VSI when there is low or no radiation from sun.

#### 4.9. Summary

This chapter presents the analysis of the control systems used in a three-phase two-level VSI connected for island and grid mode operation. The discussed control systems are proportional-resonance (PR) and proportional-integral (PI) controllers. Proportional-integral (PI) control in the dq-frame was selected for further analysis, and itssuitability investigated for high-power medium-voltage grid connected inverters used as interfacing units for large-scale integration of renewable energy sources, when selective harmonic elimination is used. It has been demonstrated that the use of SHE does not compromise any grid connected inverter functionalites that are normally provided when controlled using carrier based PWM strategies. Additionally, this chapter established that contrary to traditional SHE implementation, adjustment of the 3<sup>rd</sup> harmonic magnitude leads to a universal solution for harmonic elimination equations that evolves switching angles that are better spread over 90°. This feature can be used to simplify pratical realization of SHE at high modulation indices when a large number of harmonics are eliminated. The proposed system was verified in unity and lagging power

factor applications by simulation and experimentation. For completeness, an additional case study was presented that illustrates typical implementation of a PV inverter with maximum power point tracking.

#### References

- [4.1] S. Alepuz, S. Busquets-Monge, J. Bordonau, J. Gago, D. Gonzalez, and J. Balcells, "Interfacing Renewable Energy Sources to the Utility Grid Using a Three-Level Inverter," *Industrial Electronics, IEEE Transactions on*, vol. 53, pp. 1504-1511, 2006.
- [4.2] S. Daher, J. Schmid, and F. L. M. Antunes, "Multilevel Inverter Topologies for Stand-Alone PV Systems," *Industrial Electronics, IEEE Transactions on*, vol. 55, pp. 2703-2712, 2008.
- [4.3] L. Asiminoaei, R. Teodorescu, F. Blaabjerg, and U. Borup, "A digital controlled PV-inverter with grid impedance estimation for ENS detection," *Power Electronics, IEEE Transactions on*, vol. 20, pp. 1480-1490, 2005.
- [4.4] E. Babaei and J. Ebrahimi, "A New Topology of Cascaded Multilevel Converters with Reduced Number of Components for High-Voltage Applications," *Power Electronics, IEEE Transactions on*, vol. PP, pp. 1-1, 2011.
- [4.5] X. Chavarria, D. Biel, F. Guinjoan, C. Meza, and J. Negroni, "Energy-Balance Control of PV Cascaded Multilevel Grid-Connected inverters for Phase-Shifted and Level-Shifted Pulse-Width Modulations," *Industrial Electronics, IEEE Transactions on*, vol. PP, pp. 1-1, 2012.
- [4.6] W. Fei, B. Wu, and Y. Huang, "Half-wave symmetry selective harmonic elimination method for multilevel voltage source inverters," *Power Electronics, IET,* vol. 4, pp. 342-351, 2011.
- [4.7] D. Floricau and F. Richardeau, "New Multilevel Converters Based on Stacked Commutation Cells With Shared Power Devices," *Industrial Electronics, IEEE Transactions on*, vol. 58, pp. 4675-4682, 2011.
- [4.8] J. Chiasson, B. Ozpineci, D. Zhong, and L. M. Tolbert, "Conditions for Capacitor Voltage Regulation in a Five-Level Cascade Multilevel Inverter: Application to Voltage-Boost in a PM Drive," in *Electric Machines & Drives Conference, 2007. IEMDC '07. IEEE International*, 2007, pp. 731-735.
- [4.9] A. Kavousi, B. Vahidi, R. Salehi, M. Bakhshizadeh, N. Farokhnia, and S. S. Fathi, "Application of the Bee Algorithm for Selective Harmonic Elimination Strategy in Multilevel Inverters," *Power Electronics, IEEE Transactions on*, vol. 27, pp. 1689-1696, 2012.
- [4.10] J. Shuai, C. Dong, L. Yuan, and P. Fang Zheng, "Grid-Connected Boost-Half-Bridge Photovoltaic Microinverter System Using Repetitive Current Control and Maximum Power Point Tracking," *Power Electronics, IEEE Transactions on*, vol. 27, pp. 4711-4722, 2012.
- [4.11] N. A. Rahim, K. Chaniago, and J. Selvaraj, "Single-Phase Seven-Level Grid-Connected Inverter for Photovoltaic System," *Industrial Electronics, IEEE Transactions on*, vol. 58, pp. 2435-2443, 2011.

- [4.12] N. A. Rahim and J. Selvaraj, "Multistring Five-Level Inverter With Novel PWM Control Scheme for PV Application," *Industrial Electronics, IEEE Transactions* on, vol. 57, pp. 2111-2123, 2010.
- [4.13] S. B. Kjaer, J. K. Pedersen, and F. Blaabjerg, "A review of single-phase gridconnected inverters for photovoltaic modules," *Industry Applications, IEEE Transactions on*, vol. 41, pp. 1292-1306, 2005.
- [4.14] E. Villanueva, P. Correa, J. Rodriguez, and M. Pacas, "Control of a Single-Phase Cascaded H-Bridge Multilevel Inverter for Grid-Connected Photovoltaic Systems," *Industrial Electronics, IEEE Transactions on*, vol. 56, pp. 4399-4406, 2009.
- [4.15] K. Feel-Soon, P. Sung-Jun, C. Su Eog, U. K. Cheul, and T. Ise, "Multilevel PWM inverters suitable for the use of stand-alone photovoltaic power systems," *Energy Conversion, IEEE Transactions on*, vol. 20, pp. 906-915, 2005.
- [4.16] Y. Bo, L. Wuhua, Z. Yi, and H. Xiangning, "Design and Analysis of a Grid-Connected Photovoltaic Power System," *Power Electronics, IEEE Transactions* on, vol. 25, pp. 992-1000, 2010.
- [4.17] R. Gonzalez, J. Lopez, P. Sanchis, and L. Marroyo, "Transformerless Inverter for Single-Phase Photovoltaic Systems," *Power Electronics, IEEE Transactions on*, vol. 22, pp. 693-697, 2007.
- [4.18] J. M. Chang, W. N. Chang, and S. J. Chung, "Single-phase grid-connected PV system using three-arm rectifier-inverter," *Aerospace and Electronic Systems, IEEE Transactions on*, vol. 42, pp. 211-219, 2006.
- [4.19] P. P. Dash and M. Kazerani, "Dynamic Modeling and Performance Analysis of a Grid-Connected Current-Source Inverter-Based Photovoltaic System," *Sustainable Energy, IEEE Transactions on*, vol. 2, pp. 443-450, 2011.
- [4.20] B. Sahan, A. N. Vergara, N. Henze, A. Engler, and P. Zacharias, "A Single-Stage PV Module Integrated Converter Based on a Low-Power Current-Source Inverter," *Industrial Electronics, IEEE Transactions on*, vol. 55, pp. 2602-2609, 2008.
- [4.21] S. Jain and V. Agarwal, "A Single-Stage Grid Connected Inverter Topology for Solar PV Systems With Maximum Power Point Tracking," *Power Electronics, IEEE Transactions on*, vol. 22, pp. 1928-1940, 2007.
- [4.22] H. Patel and V. Agarwal, "A Single-Stage Single-Phase Transformer-Less Doubly Grounded Grid-Connected PV Interface," *Energy Conversion, IEEE Transactions on*, vol. 24, pp. 93-101, 2009.
- [4.23] C. Yang and K. Smedley, "Three-Phase Boost-Type Grid-Connected Inverter," *Power Electronics, IEEE Transactions on*, vol. 23, pp. 2301-2309, 2008.
- [4.24] C. Yang and K. M. Smedley, "A cost-effective single-stage inverter with maximum power point tracking," *Power Electronics, IEEE Transactions on*, vol. 19, pp. 1289-1294, 2004.
- [4.25] M. Mohr and F. W. Fuchs, "Comparison of three phase current source inverters and voltage source inverters linked with DC to DC boost converters for fuel cell generation systems," in *Power Electronics and Applications, 2005 European Conference on,* 2005, pp. 10 pp.-P.10.
- [4.26] H. Hu, S. Harb, D. Zhang, X. Fang, Q. Zhang, J. Shen, and I. Batarseh, "A Three-port Flyback for PV Micro-Inverter Applications with Power Pulsation Decoupling Capability," *Power Electronics, IEEE Transactions on*, vol. PP, pp. 1-1, 2012.
- [4.27] J. Huusari and T. Suntio, "Dynamic Properties of Current-Fed Quadratic Full-Bridge Buck Converter for Distributed Photovoltaic MPP-Tracking Systems," *Power Electronics, IEEE Transactions on*, vol. 27, pp. 4681-4689, 2012.
- [4.28] D. Ahmadi, Z. Ke, L. Cong, H. Yi, and W. Jin, "A Universal Selective Harmonic Elimination Method for High-Power Inverters," *Power Electronics*, *IEEE Transactions on*, vol. 26, pp. 2743-2752, 2011.
- [4.29] V. G. Agelidis, A. I. Balouktsis, and M. S. A. Dahidah, "A Five-Level Symmetrically Defined Selective Harmonic Elimination PWM Strategy: Analysis and Experimental Validation," *Power Electronics, IEEE Transactions* on, vol. 23, pp. 19-26, 2008.
- [4.30] A. Cetin and M. Ermis, "VSC-Based D-STATCOM With Selective Harmonic Elimination," *Industry Applications, IEEE Transactions on*, vol. 45, pp. 1000-1015, 2009.
- [4.31] Z. Fanghua and Y. Yangguang, "Selective Harmonic Elimination PWM Control Scheme on a Three-Phase Four-Leg Voltage Source Inverter," *Power Electronics, IEEE Transactions on*, vol. 24, pp. 1682-1689, 2009.
- [4.32] M. S. A. Dahidah and V. G. Agelidis, "Selective Harmonic Elimination PWM Control for Cascaded Multilevel Voltage Source Converters: A Generalized Formula," *Power Electronics, IEEE Transactions on*, vol. 23, pp. 1620-1630, 2008.
- [4.33] P. N. Enjeti, P. D. Ziogas, and J. F. Lindsay, "Programmed PWM techniques to eliminate harmonics: a critical evaluation," *Industry Applications, IEEE Transactions on*, vol. 26, pp. 302-316, 1990.
- [4.34] Z. Keliang and W. Danwei, "Relationship between space-vector modulation and three-phase carrier-based PWM: a comprehensive analysis [three-phase inverters]," *Industrial Electronics, IEEE Transactions on,* vol. 49, pp. 186-196, 2002.
- [4.35] I. J. Gabe, V. F. Montagner, and H. Pinheiro, "Design and Implementation of a Robust Current Controller for VSI Connected to the Grid Through an LCL Filter," *Power Electronics, IEEE Transactions on*, vol. 24, pp. 1444-1452, 2009.
- [4.36] T. Yi, L. Poh Chiang, W. Peng, C. Fook Hoong, and G. Feng, "Exploring Inherent Damping Characteristic of LCL-Filters for Three-Phase Grid-Connected Voltage Source Inverters," *Power Electronics, IEEE Transactions* on, vol. 27, pp. 1433-1443.
- [4.37] Y. Zhang, G. P. Adam, T. C. Lim, S. J. Finney, and B. W. Williams, "Voltage source converter in high voltage applications: Multilevel versus two-level converters," in AC and DC Power Transmission, 2010. ACDC. 9th IET International Conference on, 2010, pp. 1-5.
- [4.38] W. Tsai-Fu, C. Chih-Hao, L. Li-Chiun, and K. Chia-Ling, "Power Loss Comparison of Single- and Two-Stage Grid-Connected Photovoltaic Systems," *Energy Conversion, IEEE Transactions on*, vol. 26, pp. 707-715, 2011.
- [4.39] S. M. Halpin, "Comparison of IEEE and IEC harmonic standards," in *Power* Engineering Society General Meeting, 2005. IEEE, 2005, pp. 2214-2216 Vol. 3.
- [4.40] X. Wilsun, "Comparisons and comments on harmonic standards IEC 1000-3-6 and IEEE Std. 519," in *Harmonics and Quality of Power, 2000. Proceedings. Ninth International Conference on*, 2000, pp. 260-263 vol.1.

- [4.41] J. Selvaraj and N. A. Rahim, "Multilevel Inverter For Grid-Connected PV System Employing Digital PI Controller," *Industrial Electronics, IEEE Transactions on*, vol. 56, pp. 149-158, 2009.
- [4.42] R. Kadri, J. P. Gaubert, and G. Champenois, "An Improved Maximum Power Point Tracking for Photovoltaic Grid-Connected Inverter Based on Voltage-Oriented Control," *Industrial Electronics, IEEE Transactions on*, vol. 58, pp. 66-75, 2011.
- [4.43] H. Zhang, Z. Hongwei, R. Jing, L. Weizeng, R. Shaohua, and G. Yongjun, "Three-phase grid-connected photovoltaic system with SVPWM current controller," in *Power Electronics and Motion Control Conference, 2009. IPEMC* '09. IEEE 6th International, pp. 2161-2164, 2009.
- [4.44] D. Zhou, Z. Zhao, M. Eltawil, and L. Yuan, "Design and control of a three-phase grid-connected photovoltaic system with developed maximum power point tracking," in *Applied Power Electronics Conference and Exposition*, 2008. *APEC 2008. Twenty-Third Annual IEEE*, pp. 973-979, 2008.

# **Chapter Five**

# **Current Source Inverter for Photovoltaic Application**

## 5.0. Introduction

This chapter discusses possible current source inverter (CSI) topologies worthy of attention for grid interfacing of photovoltaic systems. The suitable topologies are singlephase H-bridge, three-phase, and multilevel current source inverters are briefly discussed, taking into consideration the features suiting photovoltaic system requirements. A three-phase CSI is selected for grid integration in this thesis and further investigation in respect of the modulation strategies and control system are presented in this and subsequent chapters. Several modulation strategies, namely carrier-based sinusoidal PWM, selective harmonic elimination (SHE), space vector modulation, and twelve-step direct PWM, are discussed and compared to a new modulation technique termed direct regular-sampled pulse width modulation (DRSPWM). The theoretical basis of the proposed modulation technique is described in detail and includes dwell time calculations and switching sequence selection. The proposed DRSPWM technique is simple to implement digitally, and offers a degree of freedom and flexibility in switching sequence and pulse placement. The validity and suitability of the presented direct regular-sampled pulse width modulation technique is substantiated using simulations and experimentation.

### 5.1. Background

Though the voltage source inverter is the dominant topology in distributed generation (DG) applications, it has drawbacks such as short lifetime of the dc-link capacitor and inherent buck characteristics. Therefore, researchers and manufacturers continue their investigation into development of another efficient topology. The topology which has attracted considerable attention is the current source inverter due to its inherent boost

characteristics, longer lifetime of the energy storage unit, inherent fault protection capability, and direct control of the output current.

In photovoltaic applications, pulse width modulated CSIs are used for grid-interfacing between the photovoltaic system and ac main grid [5.1]. The current source option for PV grid interfacing has been advocated as a means of improving the effectiveness of maximum power point tracking (MPPT), as the number of series connected PV cells in each string is reduced [5.2, 3]. This reduces the impact of partial shading, and increases system efficiency by using one conversion stage instead of two stages, thereby benefiting from the voltage boost capability of the CSI [5.4-9]. Furthermore, PV system reliability is increased by replacing the shunt input electrolytic capacitor in the VSI with a series input inductor in the CSI.

# 5.2. Current source inverter topologies

This section presents current source inverter topologies being used in photovoltaic applications. The CSI has not been extensively investigated for grid-connected renewable energy systems. The CSI appears to be a feasible alternative to the VSI for grid connection of PV distributed generation, for the following reasons:

- The dc input current is continuous which is important for PV application.
- System reliability is increased by replacing the shunt input electrolytic capacitor with a series input inductor.
- The inherent CSI voltage boosting capability allows a low-voltage PV array to be grid interfaced without the need of a transformer or an additional boost stage.

# 5.2.1 Single-phase H-bridge current source inverter

A single-phase H-bridge current source inverter has four IGBT switches and four diodes, assigned as  $S_1$  to  $S_4$ . Each diode is connected in series with an IGBT switch for reverse voltage blocking capability. A diagram of the single-phase H-bridge CSI topology is shown in Figure 0.1. A single-phase system has even harmonics on the dc side, which affect MPPT, reduce PV cell lifetime, and are associated with odd order

harmonics on the grid side [5.6, 10]. Therefore, eliminating the even harmonics on the dc side is essential when a single-phase H-bridge CSI is used.



Figure 0.1: Topology of a single-phase H-bridge CSI.

Power fluctuation, caused by even harmonics on the dc-side in the single-phase singlestage PV application, affects the PV MPPT system, reduces PV lifetime, and reflects grid side 3<sup>rd</sup> harmonics[5.6, 10, 11].

#### 5.2.2 Three-phase current source inverter

The three-phase CSI has six self-commutated unidirectional switches. The switches experience bipolar voltage stresses, with a maximum peak voltage equal to that of the ac output voltage. Similar to the single-phase H-bridge CSI, due to the low reverse voltage blocking capability of the IGBT, diodes have to be connected in-series with the switches. To eliminate the need for series diodes, symmetrical IGCTs may be an attractive alternative. The energy storage component at the dc-side is an inductor. The filter capacitor at the inverter output is used to attenuate high frequency harmonics associated with the modulation switching frequency. The CSI in Figure 0.2 generates three-level current output levels  $\pm I_{dc}$  and 0. In each instant, one switch from the upper group (S<sub>1</sub>, S<sub>3</sub> and S<sub>5</sub>) and one from the lower group (S<sub>4</sub>, S<sub>6</sub> and S<sub>2</sub>) must be on. A zero current level at the output is achieved by turning on both switching devices in the phase leg. Unlike the single-phase system, the three-phase CSI does not contain even harmonics at the dc side or 3<sup>rd</sup> harmonics at the ac side of the system.



Figure 0.2: Topology of the two-level three-phase CSI.

## 5.2.3. Multilevel current source inverter

Multilevel current source inverters have gained attention in recent years and have been studied in high-voltage and high-power applications. By increasing the number of levels, the output currents in the case of a CSI are staircase current waveforms. With a large number of steps, the output current waveforms tend to sinusoidal waveforms, with a reduced THD and di/dt, and these can be achieved with a reduced switching frequency. Moreover, operation of a multilevel inverter at a low switching frequency results in lower switching loss and higher power transfer capability, compared to a standard three-phase inverter. For their advantageous features, multilevel CSIs have been adopted in distributed generation (DG) applications, particularly photovoltaic system.

Most of the multilevel CSIs reported in the literature are based on a single-phase cell [5.12-15], and are basically variations of the generalized multi-cell structure proposed in [5.16]. The generalized structured proposed in [5.17] has been extended to a three-phase topology in [5.13]. The topology proposed in [5.18] is a three-phase seven-level CSI that uses a phase-shifted trapezoidal PWM scheme for better harmonic performance with a low switching frequency. A modified three-phase multi-cell CSI topology proposed in [5.19] is different from the earlier proposed topologies in terms of the number of current levels. In [5.20], a three-phase multilevel CSI with 2(n+1) output current levels is introduced, with an associated closed-loop control strategy. Though a multilevel topology is able to overcome certain disadvantages of having single-stage CSIs, the advantages are achieved at the expense of a more complex PV system.

Furthermore, a bulky transformer and an electrolytic capacitor are required. Thus, the three-phase CSI is chosen for detailed analysis in a grid interfacing PV system while the H-bridge single phase and multilevel CSI are not discussed further in this chapter.

## 5.3. Modulation strategies for current source inverter

A number of pulse width modulation (PWM) strategies can be used to generate gating signals for three phase current source inverters, including those currently operational in grid mode. Examples of PWM strategies used to control a three-phase CSI are selective harmonic elimination (SHE) [5.21-29], carrier-based sinusoidal pulse width modulation (CSPWM) [5.15, 30], and space vector modulation (SVM) [5.31-44]. Two switching constraints must always be met to properly gate the power switches of the CSI. These are:

- The ac-side is mainly capacitive therefore it must not be short-circuited. This implies that at most one upper switch (S<sub>1</sub>, S<sub>3</sub> and S<sub>5</sub>) or one lower switch (S<sub>4</sub>, S<sub>6</sub> and S<sub>2</sub>) should be closed at any instant.
- The dc bus is a current-source type, therefore cannot become open-circuit and at least one top switch (S<sub>1</sub>, S<sub>3</sub> and S<sub>5</sub>) and one lower switch (S<sub>4</sub>, S<sub>6</sub> and S<sub>2</sub>) must be closed at every instant.

# 5.3.1. Carrier-based sinusoidal pulse width modulation

Carrier-based SPWM is simple to implement in a VSI and can be realized with a relatively low switching frequency. However, implementation of CSI carrier-based SPWM is not as straightforward as for the VSI case. The common approach used for the implementation of CSI carrier-based SPWM is to modify the gating signals of the VSI PWM using logic mapping circuitry as shown in Figure 0.3(a) [5.15, 30, 45]. Referring to Figure 0.3(b), the reference modulating signals ( $v_{ma}$ ,  $v_{mb}$  and  $v_{mc}$ ) are compared to high frequency carrier signals ( $v_{cr}$ ), and with logic mapping circuitry, generate the CSI gating signals (S<sub>1</sub> to S<sub>6</sub>) as illustrated in Figure 0.4(a). A sample of pre-filter inverter current ( $i_{ai}$ ), and three-phase load current and voltage are shown in Figure 0.4(b) to (d) respectively.

The fundamental inverter current magnitude,  $I_m$  in relation to the amplitude modulation index,  $m_a$ , for carrier-based SPWM, can be expressed as:

$$m_a = \frac{I_m}{I_{dc}} \tag{5.1}$$

Since the maximum achievable fundamental current magnitude with carrier-based SPWM is only  $I_{m,\max} = m_a \frac{\sqrt{3}}{2} I_{dc} = 0.866 I_{dc}$  for  $m_a=1$ , carrier-based SPWM suffers from poor dc link current utilization. Furthermore, the use of logic mapping circuitry increases modulator complexity, and leads to suboptimal switching sequences from the switching loss viewpoint.



(b) Three phase reference and high-frequency carrier signals of VSI carrier-based SPWM before passing through logic mapping circuitry to generate CSI gating signals

Figure 0.3: Modulation process of the CSI using carrier-based SPWM



(d) Three phase load voltage ( $v_{ao}$ ,  $v_{bo}$  and  $v_{co}$ ) Figure 0.4: Sample CSI open-loop simulation waveforms using SPWM

# 5.3.2 Selective harmonic elimination

Generally, selective harmonic elimination is favoured over space vector and carrier based pulse width modulation in high-power medium-voltage CSI applications [5.28, 29]. This is due to its ability to synthesize high quality output current and voltage by

eliminating a number of pre-defined low-order harmonics, without a significant increase in switching frequency. There are two possible SHE implementations that can be used to generate gating signals for a CSI, as shown in Figure 0.5(a). One possible way is by direct calculation of the switching angles based on a pre-determined inverter phase current pattern to eliminate specific harmonics. An example of an inverter current waveform,  $i_i$  using SHE with two switching angles or notches over a  $\frac{1}{2}\pi$ period to eliminate the 5<sup>th</sup> and 7<sup>th</sup> harmonics, is illustrated in Figure 0.5(b). The Fourier series expansion of the symmetric PWM waveform can be expressed as:

$$I_i(\omega t) = \sum_{n=1,3,5,\dots}^{\infty} \frac{4I_{dc}}{n\pi} (\cos(n\alpha_1) - \cos(n\alpha_2) \dots + \cos(n\alpha_n)) \sin(n\omega t)$$
(5.2)

where *n* is the order of the harmonic and  $\alpha_n$  is the *n*<sup>th</sup> switching angle.

Based on(5.2), the following group of polynomial equations can be utilised to calculate n switching angles and realize the selective harmonic elimination of up to the  $m^{th}$  order:

$$\frac{4I_{dc}}{\pi} \left( \cos(\alpha_1) - \cos(\alpha_2) \dots + \cos(\alpha_n) \right) = I_i$$
  

$$\cos(5\alpha_1) - \cos(5\alpha_2) \dots + \cos(5\alpha_n) = 0$$
  

$$\cos(7\alpha_1) - \cos(7\alpha_2) \dots + \cos(7\alpha_n) = 0$$
  

$$\dots$$
  

$$\cos(m\alpha_1) - \cos(m\alpha_2) \dots + \cos(m\alpha_n) = 0$$
  
(5.3)

In this equation group, the first equation is used to establish the magnitude of the fundamental component  $(I_i)$  and the other equations are utilised to ensure the elimination of the selected harmonics. Thus, by calculating the *n* switching angles, *n*-1 harmonics can be eliminated [5.24, 46, 47].

Alternatively, SHE can be realised by adapting the approaches presented in [5.21] and [5.25] that modify the chopping angles of a VSI using logic mapping circuitry to generate the gating signals for the CSI. The Fourier series expansion and related nonlinear equations for a VSI were discussed in chapter three.

The main disadvantage of SHE is that the output current waveform quality deteriorates as the modulation index decreases. This problem can be avoided at the expense of increased switching loss by increasing the number of notches per quarter fundamental cycle as the modulation decreases [5.22-24]. Another limitation of SHE is it does not ensure complete elimination of the selected harmonics over the full modulation index

linear range as the windows between adjacent switching angles tend to be small at low and high modulation indices, which is difficult to realise practically.



Figure 0.5: (a) Diagram of possible PWM gating signal generation using selective harmonic elimination and (b) typical inverter current waveform  $(i_i)$  relative to dc-link current  $(\pm I_{dc})$ .

## 5.4. Direct CSI space vector modulation

A version of space vector modulation that is specific to the CSI is shown in Figure 0.6, and is derived from transformation of the inverter phase currents into the  $\alpha$ - $\beta$  reference frame using Clark Transform, based on similar assumptions to those used for the VSI [5.35, 36]. This approach offers a higher degree of freedom and flexibility in terms of pulse placement and switching sequence selection.

To satisfy CSI switching constraints, there are nine switching states: three zero states and six active states, as listed in Table 0.1. One zero state [14] represents  $S_1$  and  $S_4$  in phase leg 'a' being simultaneously on while the other four switches are off. The dc-link current bypasses the outputs, specifically  $i_{ai}=i_{bi}=i_{ci}=0$ . The active state [61] represents  $S_6$ in phase leg 'b' and  $S_1$  in phase leg 'a', being on. The dc-link current flow through  $S_6$ , the load,  $S_1$ , and then back to dc source resulting in  $i_{ai}=I_{dc}$  and  $i_{bi}=-I_{dc}$ .

			Inverter current		rrent	
State type	Switch state	On-switch state	i <sub>ai</sub>	i <sub>bi</sub>	i <sub>ci</sub>	Space vector
	[14]	$S_1, S_4$				
Zero states	[36]	S <sub>3</sub> , S <sub>6</sub>	0	0	0	$\vec{I}_0$
	[52]	$S_5, S_2$				
Active states	[61]	S <sub>6</sub> , S <sub>1</sub>	$I_{dc}$	$-I_{dc}$	0	$\vec{I}_1$
	[12]	S <sub>1</sub> , S <sub>2</sub>	$I_{dc}$	0	$-I_{dc}$	$\vec{I}_2$
	[23]	S <sub>2</sub> , S <sub>3</sub>	0	$I_{dc}$	$-I_{dc}$	$\vec{I}_3$
	[34]	S <sub>3</sub> , S <sub>4</sub>	$-I_{dc}$	$I_{dc}$	0	$\vec{I}_4$
	[45]	S <sub>4</sub> , S <sub>5</sub>	$-I_{dc}$	0	$I_{dc}$	$\vec{I}_5$
	[56]	S <sub>5</sub> , S <sub>6</sub>	0	$-I_{dc}$	$I_{dc}$	$\vec{I}_6$

Table 0.1: Switch states and space vector.

The space vector diagram for direct-CSI SVM is shown in Figure 0.6 where  $\vec{I}_1$  to  $\vec{I}_6$  are the active vectors and  $\vec{I}_0$  is a zero vector located at the centre of the hexagon. As mentioned, using the Clark transformation, the inverter phase current is transformed to  $\alpha$ - $\beta$  reference:

$$\begin{bmatrix} i_{\alpha}(t) \\ i_{\beta}(t) \end{bmatrix} = \frac{2}{3} \begin{bmatrix} 1 & -\frac{1}{2} & -\frac{1}{2} \\ 0 & \sqrt{3} & -\frac{\sqrt{3}}{2} \end{bmatrix} \begin{bmatrix} i_{ai}(t) \\ i_{bi}(t) \\ i_{ci}(t) \end{bmatrix}$$
(5.4)



Figure 0.6: Diagram of PWM gating signal generation using direct SVM.

During one switch period  $T_s$ , PWM modulation can be summarized as:

$$\vec{I}_{ref}T_s = \vec{I}_i T_1 + \vec{I}_{i+1}T_2 + \vec{I}_0 T_0$$
  
$$T_s = T_1 + T_2 + T_0$$
(5.5)

where  $T_{I}$ ,  $T_{2}$  and  $T_{0}$  are the dwell times of the chosen switches and i=0,1,2,...,6. The *i* value depends on the sector that  $\vec{I}_{ref}$  lies in. An example of  $\vec{I}_{ref}$  falling in sector I and synthesized by  $\vec{I}_{1}$ ,  $\vec{I}_{2}$  and  $\vec{I}_{0}$  is illustrated in Figure 0.7.



Figure 0.7: Synthesis of  $\vec{I}_{ref}$  by  $\vec{I}_1$ ,  $\vec{I}_2$  and  $\vec{I}_0$ .

Equation (5.5) becomes

$$\vec{I}_{ref}T_s = \vec{I}_1T_1 + \vec{I}_2T_2 + \vec{I}_0T_0$$
  
$$T_s = T_1 + T_2 + T_0$$
(5.6)

$$\vec{I}_{ref} = I_{ref} e^{j\theta}, \ \vec{I}_1 = \frac{2}{\sqrt{3}} I_{dc} e^{-j\frac{\pi}{6}}, \ \vec{I}_2 = \frac{2}{\sqrt{3}} I_{dc} e^{j\frac{\pi}{6}} \ and \ \vec{I}_0 = 0$$
 (5.7)

Substituting (5.7) into (5.6) yields dwell times ( $T_1$ ,  $T_2$  and  $T_0$ ) for sector I as:

$$T_1 = m_a \sin\left(\frac{\pi}{6} - \theta\right) T_s$$
  

$$T_2 = m_a \sin\left(\frac{\pi}{6} + \theta\right) T_s$$
  

$$T_0 = T_s - T_1 - T_2$$
  
(5.8)

where  $m_a$  is the modulation index given by  $m_a = \frac{I_{ref}}{I_{dc}} = \frac{I_i}{I_{dc}}$  and  $I_i$  is the peak of fundamental component in  $i_i$ .

The selection of a zero vector sequence must ensure minimum switch transitions while maintaining switch utilization and losses. Table 0.2 lists zero vector sequences to satisfy the mentioned criterion.

Sector	Initial state	Final state	Zero state
Sector	$I_i$	$I_{i+1}$	$I_z$
Ι	$\vec{I}_1 = S_1, S_2$	$\vec{I}_2 = S_2, S_3$	$\vec{I}_0 = S_5, S_2$
II	$\vec{I}_2 = S_2, S_3$	$\vec{I}_3 = S_3, S_4$	$\vec{I}_0 = S_3, S_6$
III	$\vec{I}_3 = S_3, S_4$	$\vec{I}_4 = S_4, S_5$	$\vec{I}_0 = S_1, S_4$
IV	$\vec{I}_4 = S_4, S_5$	$\vec{I}_5 = S_5, S_6$	$\vec{I}_0 = S_5, S_2$
V	$\vec{I}_5 = S_5, S_6$	$\vec{I}_6 = S_6, S_1$	$\vec{I}_0 = S_3, S_6$
VI	$\vec{I}_6 = S_6, S_1$	$\vec{I}_1 = \mathbf{S}_1,  \mathbf{S}_2$	$\vec{I}_0 = S_1, S_4$

Table 0.2: Selection of zero vector sequence in sector I to VI.

The simulation results of gating signal generation (S<sub>1</sub> to S<sub>6</sub>) with the associated sectors and phase '*a*' inverter current (*i<sub>ai</sub>*) relative to  $I_{dc}$  for a three-phase CSI are presented in Figure 0.8. Direct-SVM is used in accordance with the switching sequence in Table 0.2, with modulation index of 0.8,  $I_{dc}$ =8A and switching frequency,  $f_{sw}$ =2.1kHz.



Figure 0.8: SVM switching sequence of a CSI over a fundamental cycle.

## 5.4.1. VSI-space vector modulation with logic mapping circuitry

The alternative approach, shown in Figure 0.9, is to use VSI space vector modulation to generate the gating signals which are then modified using logic mapping circuitry to suit the CSI [5.31, 32, 34]. The PWM modulation of VSI-SVM for a CSI is discussed in chapter three. The angles between the active space vectors for both the VSI and CSI are 60°, however the CSI active vectors lead the VSI case by 30°. This is one of disadvantages to be taken into account: a 30° phase shift in output current compared to that of the VSI. The approach suffers from the same disadvantages as the CSPWM method when using logic mapping circuitry.



Figure 0.9: Diagram of VSI-SVM with logic mapping circuit used to generate CSI gating signals.

Both CSI SVM implementations are suitable for applications that use a relatively high switching frequency. In addition to the SVM discussion, there are several new SVM implementations aimed at achieving different objectives [5.33, 37, 38]. SVM is suited for digital implementation and real-time generation of the gating signals with different patterns, and offers the advantages of faster dynamics, controllability, and flexibility. The use of low switching frequency SVM may however, lead to the appearance low-order harmonics in the baseband [5.43]. In particular, the low-order harmonics, such as the 5<sup>th</sup> and 7<sup>th</sup>, may be close to the resonant frequency of the CSI output LC filter, which may excite LC resonance. Some attempts to exploit different SVM switching sequences and manipulate dwell times to minimize the magnitudes of the CSI 5<sup>th</sup> and 7<sup>th</sup> harmonics, have been achieved with relatively low switching frequency SVM [5.39-42, 44].

### 5.5. Twelve-Step Direct Pulse Width Modulation (TS-PWM)

In [5.48], a CSI PWM scheme is proposed that offers natural soft commutation, and a 50% reduced CSPWM switching frequency. The PWM scheme exploits the natural commutation in the Graetz converter to achieve natural commutation and low switching losses by using the grid voltages as modulating signals. However, the use of these reference compromises the basic features that make PWM CSIs competitive with respect to the VSI, such as black-start capability, and operation independent of the network short circuit ratio. In other words, such a scheme in its present form cannot be used to power a dead ac network that has no live ac voltage.

#### 5.6. Proposed direct-regular sampled pulse width modulation

Because of the discussion on CSI modulation techniques, this thesis proposes a direct regular-sampled pulse width modulation (DRSPWM) technique that aims to simplify the CSI modulation process. The modulation strategy is suited for digital implementation and offers the flexibility of SVM in terms of pulse placement and switch sequence selection. The proposed modulation strategy exploits a combination of forced and natural commutation imposed by the co-existence of an insulated gate bipolar transistor (IGBT) in series with a diode used in the CSI in Figure 0.2, to determine device dwell times and switching sequences.

### 5.6.1. Theoretical basis of the proposed DRSPWM

Figure 0.10 depicts a fundamental cycle of the target modulating signals divided into six sectors, designated I to VI. During the first sector, switches  $S_1$  and  $S_5$  operate alternatively with  $S_6$  in order to synthesize non-zero output currents in the inverter phases. A zero current in the inverter output is achieved by selecting one of the following combinations:  $S_1$  and  $S_4$  (1001000),  $S_3$  and  $S_6$  (001001), or  $S_5$  and  $S_2$  (010010). The CSI switch states in Figure 0.10 are expressed by  $\prod_{j=1}^{6} S_j$ , where  $S_j=1$  and 0 represents the 'ON' and 'OFF' states respectively of switch  $S_j$ . The dwell time calculations for the respective sectors are listed in Table 0.3.



Figure 0.10: Reference modulating signal.

Table 0.3: Dwell times calculations for all sequences and sections identification.

Sector	Dwell times	Sector identification	Sector	Dwell times	Sector identification
I	$t_{000011} = m_a T_s \sin \omega t$ $t_{100001} = m_a T_s \sin(\omega t + \frac{2}{3}\pi)$ $t_{001001} = T_s - t_{000011} - t_{100001}$	$I_a > 0$ and $I_c > 0$	IV	$t_{011000} = -m_a T_s \sin(\omega t + \frac{2}{3}\pi)$ $t_{001100} = -m_a T_s \sin \omega t$ $t_{001001} = T_s - t_{011000} - t_{001100}$	$I_c < 0$ and $I_a < 0$
II	$t_{100001} = -m_a T_s \sin(\omega t + \frac{4}{3}\pi)$ $t_{110000} = -m_a T_s \sin(\omega t + \frac{2}{3}\pi)$ $t_{100100} = T_s - t_{100001} - t_{110000}$	$I_b < 0$ and $I_c < 0$	V	$t_{001100} = m_a T_s \sin(\omega t + \frac{4}{3}\pi)$ $t_{000110} = m_a T_s \sin(\omega t + \frac{2}{3}\pi)$ $t_{100100} = T_s - t_{001100} - t_{000110}$	$I_c > 0$ and $I_b > 0$
III	$t_{110000} = m_a T_s \sin \omega t$ $t_{011000} = m_a T_s \sin(\omega t + \frac{2}{3}\pi)$ $t_{010010} = T_s - t_{110000} - t_{011000}$	$I_b > 0$ and $I_a > 0$	VI	$t_{000110} = -m_a T_s \sin \omega t$ $t_{000011} = -m_a T_s \sin(\omega t + \frac{4}{3}\pi)$ $t_{010010} = T_s - t_{000110} - t_{000011}$	$I_b < 0$ and $I_a < 0$

Given  $i_a+i_c=-i_b$  in a three-phase system, and assuming that within each switching cycle the dwell time associated with each active switch state is proportional to the phase current magnitude, then the dwell times associated with active states I-(ii) and I-(i) in first sector are:

$$I_{dc} \times t_{100001} = T_s \times i_a$$
  

$$I_{dc} \times t_{000011} = T_s \times i_c$$
(5.9)

If the phase currents are expressed by:  $i_a = I_m \sin(\omega t + \delta)$ ,  $i_b = I_m \sin(\omega t + \delta + \frac{4}{3}\pi)$  and  $i_c = I_m \sin(\omega t + \delta + \frac{2}{3}\pi)$ , then the dwell times  $t_{100001}$  and  $t_{000011}$  can be expressed by[5.49]:

$$t_{100001} = m_a T_s \sin(\omega t + \delta) t_{000011} = m_a T_s \sin(\omega t + \delta + \frac{2}{3}\pi)$$
(5.10)

where  $m_a$  is modulation index,

 $I_m$  is peak phase current, A;  $T_s$  is switching period, s;  $\omega$  is supply frequency, rad/s; and  $\delta$  is the load angle, rad.

The amplitude modulation index,  $m_a$  is defined as:

$$m_a = \frac{I_m}{I_{dc}} \tag{5.11}$$

where  $0 \le m_a \le 1$ , and  $m_f = \frac{f_{sw}}{f_o}$  is the ratio of switching frequency  $f_{sw}$  to fundamental frequency  $f_o$ .

The dwell time associated with a zero state is given by:

$$t_0 = T_s - t_{100001} - t_{000011} \tag{5.12}$$

There are several switching sequences that can be realized in each sector with the aim to minimize the number of switching transitions, including soft transitions between successive sectors. Table 5.4 lists all the possible switching sequences realizable over the full fundamental cycle. Figure 0.11 (a) to (c) show an example of the pulse placements for the switch sequences I and III in the first sector. Based on Figure 0.11 and Table 5.4, the switch sequences I and II operate the CSI as with typical VSI deadband modulation [5.50] that suspends each inverter phase for 120°, and hence low switching loss is achieved. For example, for  $30^{\circ} \le \omega t \le 150^{\circ}$  phase 'a' is clamped to the positive rail by maintaining switch S<sub>1</sub> in the on-state, while switches S<sub>2</sub> and S<sub>6</sub> are switched on and off to synthesize the currents in phases b and c. During the periods when S<sub>2</sub> and S<sub>6</sub> alternatively operate, S<sub>1</sub> contributes to generate non-zero current in all the phases. With sequences I and II the switch states used to produce zero current in each phase change every  $60^{\circ}$ : for example, S<sub>1</sub> and S<sub>4</sub> correspond to 000101 in sector II for  $60^{\circ} \le \omega t \le 120^{\circ}$ .

For sequence III switching, the CSI phases over the entire fundamental period are observed in Figure 0.11(c). In this sequence, the switching frequency per device and switching loss are increased. In sequence III, the switch states that produce zero output phase currents are selected to avoid any one switch operating for the full sampling period, as shown in Figure 0.11(c). Period  $t_0$ , calculated using (5.12), is assigned to the dwell times of the zero switch states, for example,  $t_{001001}=t_0$  in the first sector as shown in Table 5.4. The selection and placement of the zero switch state directly affects inverter switching loss and switching frequency per device. Sequences I and III can be realized by placing the switch states that produce zero current at the right or left, aligned within each switching cycle. Sequence III does not result in gating signals with 100 percent duty cycles, therefore it is suitable for low cost digital signal processor (DSPs) or microcontroller implementation. On this basis, the dwell times and switching sequences for the remaining sectors are determined.



Figure 0.11: Examples of the possible switch sequences in sector I.

#### Table 0.4: Summary of the realizable switch sequences for the CSI.

(The symbol \$ links reversal switch states that could be re-ordered such that the second state could precede the first without affecting the modulation process; and symbols ♂ and ♂ represent re-ordering process that places the last switch state of the sequence at beginning, and first state to the end of the sequence, without affecting the modulation process)

Sector	Active states	Zero states	Possible switching sequences			
			I	П	III	
Ι	(i) 000011 (ii) 100001	<ul> <li>(a) 100100</li> <li>(b) 001001</li> <li>(c) 010010</li> </ul>	000011≒100001 → <b>001001 ౮</b>	<b>2</b> 000011→ <b>001001</b> →100001 <b>5</b>	000011→100001 → <b>100100</b>	
II	(i) 100001 (ii) 110000	(a) 100100 (b) 001001 (c) 010010	100001≒110000 → <b>100100 ౮</b>	<b>U</b> 100001→ <b>100100</b> →110000 <b>U</b>	100001→110000 → <b>010010</b>	
III	(i) 110000 (ii) 011000	(a) 100100 (b) 001001 (c) 010010	110000 <b></b> ⇔011000 → <b>010010 ౮</b>	<b>0</b> 110000→ <b>010010</b> →011000 <b>0</b>	110000→011000 → <b>001001</b>	
IV	(i) 011000 (ii) 001100	<ul> <li>(a) 100100</li> <li>(b) 001001</li> <li>(c) 010010</li> </ul>	011000 <b></b> ⇔001100 → <b>001001 ౮</b>	<b>℃</b> 011000→ <b>001001</b> →001100 <b>౮</b>	011000→001100 → <b>100100</b>	
v	(i) 001100 (ii) 000110	(a) 100100 (b) 001001 (c) 010010	001100 <b></b> ⇔000110 → <b>100100 ౮</b>	<b>2</b> 001100→ <b>100100</b> →000110 <b>5</b>	001100→000110 → <b>010010</b>	
VI	(i) 000110 (ii) 000011	<ul> <li>(a) 100100</li> <li>(b) 001001</li> <li>(c) 010010</li> </ul>	000110 <b>⇔</b> 000011 → <b>010010౮</b>	<b>2</b> 000110→ <b>010010</b> →000011 <b></b>	000110→000011 → <i>001001</i>	

After re-ordering the states of sequences I, II, and III, it is established that switching sequence III is unique and non-reversible (any re-ordering to its active or zero states produces switching sequences that cannot be realized practically). Switch sequence II however, permits re-ordering of the active states, but at the expense of fewer optimized switching sequences as the transitions between successive sectors necessitate switching of more than one inverter leg at the same time. The active and zero vectors of switch sequence I can be reordered as shown in Figure 0.12(a) to (c) and Table 5.4, without affecting the modulation process. Reordering of the active states in sequences I and II results in new realizable switching sequences belonging to the original groups (sequences I and II respectively). Also, positioning of the zero-current vector at the beginning or at the end of the switching period in sequence I produces new valid sequences belonging to the same group. Figure 0.12(a) to (c) show re-ordering of active and zero states of the sequence A (example of switch combinations belonging to

sequences I) that produces a new switching sequence belonging to the same group, with the turn-on order of the switching devices changed.



Figure 0.12: Pictorial representation of active states interchanged for sequence I.

## 5.6.2 Open-loop validation of the proposed DRSPM

This section presents simulation and experimental results to validate the performance of DRSPWM when used to control a CSI. To further highlight the features of the proposed DRSPWM, CSPWM and SVM are compared in terms of switching frequency per device, harmonic distribution, fundamental inverter current, and total harmonic distortion (THD). The simulation and experimental results presented are obtained when the CSI feeds a resistive load of 22 $\Omega$  through a CL filter, with capacitance  $C_F$ =40 $\mu$ F and inductance  $L_F$ =3mH, at a 2.1kHz switching frequency and a dc link inductance of 10mH. This test is necessary to isolate issues associated with the proposed modulation strategy from those of the control system. The dc link current is 5A and the modulation index is 0.8.

The proposed DRSPWM is implemented as an embedded function in Simulink, with all real-time restrictions implemented so that it can be used for simulation and experimentation using dSPACE1006, thereby benefiting from code generation facilities of the real-time workshop embedded coder. Block diagrams that summarise

dSPACE1006 and generic digital implementations of the proposed DRSPWM are shown in Figure 0.13.



(a) dSPACE implementation of the proposed DRSPWM



(b) Generic implementation on any general purpose DSP (digital signal processing) or microcontroller ( $d_{s1}$  to  $d_{s6}$  represent duty cycles of the switches  $S_1$  to  $S_6$ )

Figure 0.13: dSPACE and generic block diagrams that summarise digital implementation of the proposed DRSPWM.



Figure 0.14: Illustration of arbitrary pulse generation.

Switches		Switches	
$S_I$	$t_1 = t_{000011}$ and $t_2 = t_{100001}$	$S_4$	$t_1=0$ and $t_2=0$
$S_2$	$t_1=0 \text{ and } t_2=0$	$S_5$	$t_1 = 0$ and $t_2 = t_{000011}$
$S_3$	$t_1 = t_{000011} + t_{100001}$ and $t_2 = t_0$	$S_6$	$t_1=0$ and $t_2=T_s$

Table 0.5: Example calculations of the gating pulses delay-time and on-time in the first sector.

Detailed implementation is summarized as follows:

The custom block that permits arbitrary generation of pulse patterns in dSPACE1006 accepts two variables  $t_1$  and  $t_2$ , in addition to the switching period,  $T_s$ , where  $t_1$  represents pulse delay time, and  $t_2$  is pulse on time (see Figure 0.14). A combination of assembly and low-level C code, and C-sfunction that communicates with the hardware in the dSPACE1006 its the custom block, are written and then integrated into Simulink to permit arbitrary signal generation. The inputs to the C-sfunction are switching period  $T_s$ , and  $t_1$  and  $t_2$  for each switch. The DRSPWM scheme is coded as an embedded function in the Simulink environment. The dwell times in each sector are calculated according to Table 0.3, and the delay-time and on-time for each switch in the CSI are set according to the example mapping in Table 5.5.

In this manner, all the proposed switching sequences I, II and III highlighted in Table 5.4 are realized. This explanation shows the suitability of the proposed DRSPWM for digital implementation, having the same flexibility as SVM in controlling switching sequence selection and pulse placement. These features are not available in CSPWM which uses hardware mapping or digital based mapping circuits, or in TS-PWM[5.48]. In this manner the gating pulses can be placed anywhere within the switching period.

#### 5.6.3. Simulation results

The open loop simulation results in Figure 0.15 illustrate the performance of the proposed DRSPWM at  $I_{dc}$ =5A,  $m_a$ =0.8 and a 2.1kHz switching frequency. The generated gating signal S<sub>1</sub> to S<sub>6</sub> using the proposed sequences (sequence I,II and III) are shown in Figure 0.15(a) and (b). The figure indicates that the number of switching transitions per fundamental cycle using sequences I and II is less than sequence III. The

current and voltage waveforms displayed in Figure 0.15(c) to (f) confirm the effectiveness of the proposed DRSPWM in producing a good quality output without having a complex mapping circuit for PWM generation.





Figure 0.15: Open loop simulation waveforms illustrating the performance of the proposed DRSPWM at  $I_{dc}$ =5A,  $m_a$ =0.8 and a2.1kHz switching frequency.

#### 5.6.4. Experimental verification of the proposed DRSPWM

To illustrate the potential deterioration of output current waveform quality as the modulation index varies over the full linear range (0~1) for a fixed input filter ( $C_F$ =40µF and  $L_F$ =3mH), experimental waveforms for the modulating signals, and pre-filter and three-phase load currents obtained with 0.3, 0.5, 0.8 and 0.95 modulation indices, are presented in Figure 0.16 and 5.17. The load current quality remains virtually the same, except where the modulation index approaches unity. In this region, the pulse widths approach 100%, and dSPACE is unable to generate such pulses with sufficient accuracy. Pulse dropping introduces limited output current distortion. Therefore, the distortion shown at 0.95 modulation index is due to its implementation and this can be avoided using the Tri-Core Microcontroller TC1796 from Infineon.

Figure 0.18 present experimental results from a three-phase CSI having the same parameters and operating conditions as those in the simulations presented in Figure 0.15, with switching sequences I, II and III. The gating signal generation using the possible switching sequences are presented in Figure 0.18(a) and (b) while Figure 0.18(c) and (d) display the load voltage and current. The spectra of the pre-filter load current obtained with switching sequences I, II and III are displayed in Figure 0.19(a) to(c), with THDs of 59.13%, 58.65% and 59.27% respectively. Sequence II shows the lowest THD, with the magnitudes of low-order harmonics and lower sidebands being

larger than those of sequences I and III, which is agreement with the simulation results. The low-order harmonics appearing in the baseband of sequences I and III are due to the implementation of the proposed DRSPWM in dSPACE1006 (inability to generate pulses with 0 and 100% duty cycle, even with custom coding). The viability of the proposed CSI DRSPWM in a grid mode is investigated in chapter seven.



Figure 0.16: Experimental waveforms illustrating reference modulating signal, pre-filter inverter current and load current at  $m_a=0.3$  and  $m_a=0.5$  using the proposed DRSPWM.



Figure 0.17: Experimental waveforms illustrating reference modulating signal, pre-filter inverter current and load current at m<sub>a</sub>=0.8 and m<sub>a</sub>=0.95 using the proposed DRSPWM.



Figure 0.18: Experimental waveforms illustrating the viability of the proposed RSPWM at  $m_a$ = 0.8, 2.1kHz switching frequency, and 70µF filter capacitor.



(c) Sequence III Figure 0.19: Experimental harmonic spectrum of inverter current based on the proposed sequences I, II and III.

## 5.7. Assessment of PWM modulation techniques

A comparison between the proposed DRSPWM, the SVM and CSPWM strategies is presented. SVM and CSPWM are implemented using logic mapping circuits, all with a 2.1kHz switching frequency. Figure 0.20(a) to (d) display current waveforms in the switch  $S_I$  when the CSI is controlled using the proposed DRSPWM with switching sequences I, II and III, CSPWM and SVM respectively. The proposed DRSPWM produces the lowest number of switching transitions per fundamental period. In DRSPWM with switching sequences I and II, each CSI switching device operates at half the assigned switching frequency, which is 21 pulses compared to 40 and 27 pulses for CSPWM, and SVM and DRSPWM with sequence III. This implies that the proposed technique has lower switching losses, making it promising for mediumvoltage high-power applications. CSPWM has the highest switching rate.

Figure 0.21(a) to (d) present experimental gating signals for switch  $S_1$ ,for DRSPWM, CSPWM and SVM, to substantiate the simulation waveforms presented in Figure 0.20. The results in Figure 0.21(a) and (b) show that with the proposed DRSPWM sequences I and II, switch  $S_1$  operates only at half switching frequency. Additionally, the experimental results in Figure 0.21(c) and (d) confirm that switch  $S_1$  operates at 64%

and nearly 100% of the switching frequency when SVM and CSPWM are used respectively. This unequivocally shows DRSPWM with sequences I and III have lower switching losses, compared to SVM and CSPWM.

Simulation spectra in Figure 0.22(a) to (c) show that sequences I and III have no loworder harmonics in the baseband, with the sidebands located at lower frequencies than the switching frequency components, and with magnitudes which are much smaller than for CSPWM and SVM. Sequence II produces a significant number of low-order harmonics but has the lowest THD (72.37%) of all the discussed modulation techniques. Despite the difference in pulse placement within each switching cycle of sequences I and III, they have a similar harmonic distribution, including sideband harmonic magnitudes and similar THD (76.73% and 76.65% respectively). As the switching frequency increases, the magnitudes of the low-order harmonics produced by sequence II decrease rapidly. As a results, sequence II can be used for applications that use a high switching frequency. On the other hand, Figure 0.22(d) shows that CSPWM has loworder harmonics and its sideband placements are identical to that of a VSI, which are located at  $mf_s \pm nf$ , where n is even for odd-carrier frequencies (m=2k+1, k=0,1,2...), and odd for even carrier frequencies (m=2k, k=1,2,3...). The only exception is the absence of the carrier frequency components, which is expected in the CSI. Furthermore, CSPWM has the highest THD (91.12%). The SVM spectrum in Figure 0.22(e) shows that the sidebands around the first switching frequency component are widely spread, and that those located below the switching frequency component may behave as loworder harmonics. These may affect the ac filter design in a similar manner as low-order harmonics. Also, no switching frequency components are presence in the spectrum. The calculated THD for SVM is 76.55%.

Figure 0.23(a) and (b) present the relationships between modulation and attainable fundamental current magnitude (pre-filter), and THD for DRSPWM, CSPWM and SVM. The proposed DRSPWM matches SVM in the entire modulation index range, with both achieving a maximum fundamental current magnitude,  $I_{m,max}$  equal to  $I_{dc}$  where  $m_{a,max}=1$ . The maximum achievable fundamental current magnitude with CSPWM is only  $I_{m,max} = m_a \frac{\sqrt{3}}{2} I_{dc} = 0.866 I_{dc}$  (see Figure 0.23(a)). The plot of THD

versus modulation index in Figure 0.23(b) shows that the proposed DRSPWM and SVM have lower THD than CSPWM.



(d) Current waveform for switch  $S_l$  using SVM (27 pulses per cycle)

. Figure 0.20: Current waveforms for switch S<sub>1</sub> for the proposed DRSPWM technique, CSPWM and SVM, at  $I_{dc}$ =5A,  $m_a$ =0.8 and a 2.1kHz switching frequency.







Figure 0.22: Theoretical spectrum of inverter current before ac filters.



Figure 0.23: Comparison of PWM strategies with varying modulation index for (a) per-unit fundamental pre-filter inverter current and (b) total harmonic distortion (%THD), all at 2.1kHz switching frequency.

# 5.8. Summary

A new direct regular sampled pulse width modulation (DRSPWM) strategy for CSIs, including its theoretical basis, has been presented. The proposed DRSPWM is suitable for DSP and microcontroller digital implementation, and is simpler to implement than the other CSI modulation strategies presented in this chapter. It has been shown that some of the switch sequences of the proposed DRSPWM optimally operate the CSI from a switching loss prospective, in the same manner that dead-band modulation suspends operation of each VSI arm for 120°. The viability of the presented DRSPWM strategy has been confirmed using simulations and experimentation.

It is concluded that the distinct features of the proposed DRSPWM are:

- Reduced complexity of CSI gate signal generation through elimination of logic mapping circuitry, hence implicit pulse pattern generation of the carrier based modulation is avoided.
- The same degree of freedom, flexibility and dc link utilization as space vector modulation, while expecting to maintain the robustness of the carrier based SPWM during unbalanced operation in a grid connected environment.
- Some of its switching sequences, such as I and II, halve the switching frequency per device, compared to carrier based SPWM, thus lower switching losses are anticipated.

#### References

- [5.1] F. Blaabjerg, C. Zhe, and S. B. Kjaer, "Power electronics as efficient interface in dispersed power generation systems," *Power Electronics, IEEE Transactions on*, vol. 19, pp. 1184-1194, 2004.
- [5.2] R. Gonzalez, J. Lopez, P. Sanchis, and L. Marroyo, "Transformerless Inverter for Single-Phase Photovoltaic Systems," *Power Electronics, IEEE Transactions on*, vol. 22, pp. 693-697, 2007.
- [5.3] J. M. Chang, W. N. Chang, and S. J. Chung, "Single-phase grid-connected PV system using three-arm rectifier-inverter," *Aerospace and Electronic Systems, IEEE Transactions on*, vol. 42, pp. 211-219, 2006.
- [5.4] P. P. Dash and M. Kazerani, "Dynamic Modeling and Performance Analysis of a Grid-Connected Current-Source Inverter-Based Photovoltaic System," *Sustainable Energy, IEEE Transactions on*, vol. 2, pp. 443-450, 2011.
- [5.5] B. Sahan, A. N. Vergara, N. Henze, A. Engler, and P. Zacharias, "A Single-Stage PV Module Integrated Converter Based on a Low-Power Current-Source Inverter," *Industrial Electronics, IEEE Transactions on*, vol. 55, pp. 2602-2609, 2008.
- [5.6] S. Jain and V. Agarwal, "A Single-Stage Grid Connected Inverter Topology for Solar PV Systems With Maximum Power Point Tracking," *Power Electronics, IEEE Transactions on*, vol. 22, pp. 1928-1940, 2007.
- [5.7] H. Patel and V. Agarwal, "A Single-Stage Single-Phase Transformer-Less Doubly Grounded Grid-Connected PV Interface," *Energy Conversion, IEEE Transactions on,* vol. 24, pp. 93-101, 2009.
- [5.8] C. Yang and K. Smedley, "Three-Phase Boost-Type Grid-Connected Inverter," *Power Electronics, IEEE Transactions on*, vol. 23, pp. 2301-2309, 2008.
- [5.9] C. Yang and K. M. Smedley, "A cost-effective single-stage inverter with maximum power point tracking," *Power Electronics, IEEE Transactions on*, vol. 19, pp. 1289-1294, 2004.
- [5.10] S. B. Kjaer, J. K. Pedersen, and F. Blaabjerg, "A review of single-phase gridconnected inverters for photovoltaic modules," *Industry Applications, IEEE Transactions on*, vol. 41, pp. 1292-1306, 2005.
- [5.11] B. N. Alajmi, K. H. Ahmed, G. P. Adam, and B. W. Williams, "Single-Phase Single-Stage Transformer less Grid-Connected PV System," *Power Electronics, IEEE Transactions on*, vol. 28, pp. 2664-2676, 2013.
- [5.12] Z. Du, L. M. Tolbert, and J. N. Chiasson, "Active harmonic elimination for multilevel converters," *Power Electronics, IEEE Transactions on*, vol. 21, pp. 459-469, 2006.
- [5.13] M. S. A. Dahidah and V. G. Agelidis, "Selective harmonic elimination multilevel converter control with variant DC sources," in *Industrial Electronics* and Applications, 2009. ICIEA 2009. 4th IEEE Conference on, pp. 3351-3356, 2009.
- [5.14] M. Liserre, V. Monopoli, A. Dell'Aquila, A. Pigazo, and V. Moreno, "Multilevel Phase-Shifting Carrier PWM Technique in case of Non-Equal dc-Link Voltages," in *IEEE Industrial Electronics, IECON 2006-32nd Annual Conference on*, pp. 1639-1642, 2006.

- [5.15] S. Suroso and T. Noguchi, "Multilevel Current Waveform Generation Using Inductor Cells and H-Bridge Current-Source Inverter," *Power Electronics, IEEE Transactions on*, vol. 27, pp. 1090-1098, 2012.
- [5.16] P. E. Melin, J. R. Espinoza, N. R. Zargari, L. A. Moran, and J. I. Guzman, "A novel multi-level converter based on current source power cell," in *Power Electronics Specialists Conference, 2008. PESC 2008. IEEE*, pp. 2084-2089, 2008.
- [5.17] M. C. Chandorkar, D. M. Divan, and R. H. Lasseter, "Control techniques for multiple current source GTO converters," *Industry Applications, IEEE Transactions on*, vol. 31, pp. 134-140, 1995.
- [5.18] D. Xu and B. Wu, "Multilevel Current Source Inverters with Phase Shifted Trapezoidal PWM," in *Power Electronics Specialists Conference*, 2005. PESC '05. IEEE 36th, pp. 2540-2546, 2005.
- [5.19] S. H. Hosseini, M. F. Kangarlu, and A. K. Sadigh, "A new topology for multilevel current source inverter with reduced number of switches," in *Electrical and Electronics Engineering*, 2009. ELECO 2009. International Conference on, pp. I-273-I-277, 2009.
- [5.20] P. P. Dash and M. Kazerani, "A multilevel current-source inverter based gridconnected photovoltaic system," in *North American Power Symposium (NAPS)*, 2011, pp. 1-6, 2011.
- [5.21] J. R. Espinoza, G. Joos, J. I. Guzman, L. A. Moran, and R. P. Burgos, "Selective harmonic elimination and current/voltage control in current/voltage-source topologies: a unified approach," *Industrial Electronics, IEEE Transactions on*, vol. 48, pp. 71-81, 2001.
- [5.22] F. Wanmin, D. Xiaoli, and W. Bin, "A Generalized Half-Wave Symmetry SHE-PWM Formulation for Multilevel Voltage Inverters," *Industrial Electronics, IEEE Transactions on*, vol. 57, pp. 3030-3038, 2010.
- [5.23] V. G. Agelidis, A. I. Balouktsis, and C. Cossar, "On Attaining the Multiple Solutions of Selective Harmonic Elimination PWM Three-Level Waveforms Through Function Minimization," *Industrial Electronics, IEEE Transactions on*, vol. 55, pp. 996-1004, 2008.
- [5.24] D. Ahmadi, Z. Ke, L. Cong, H. Yi, and W. Jin, "A Universal Selective Harmonic Elimination Method for High-Power Inverters," *Power Electronics, IEEE Transactions on*, vol. 26, pp. 2743-2752, 2011.
- [5.25] J. I. Guzman, J. R. Espinoza, L. A. Moran, and G. Joos, "Selective Harmonic Elimination in Multimodule Three-Phase Current-Source Converters," *Power Electronics, IEEE Transactions on*, vol. 25, pp. 44-53, 2010.
- [5.26] M. S. A. Dahidah, V. G. Agelidis, and M. V. Rao, "On Abolishing Symmetry Requirements in the Formulation of a Five-Level Selective Harmonic Elimination Pulse-Width Modulation Technique," *Power Electronics, IEEE Transactions on*, vol. 21, pp. 1833-1837, 2006.
- [5.27] F. Wanmin, R. Xinbo, and W. Bin, "A Generalized Formulation of Quarter-Wave Symmetry SHE-PWM Problems for Multilevel Inverters," *Power Electronics, IEEE Transactions on*, vol. 24, pp. 1758-1766, 2009.
- [5.28] H. F. Bilgin and M. Ermis, "Design and Implementation of a Current-Source Converter for Use in Industry Applications of D-STATCOM," *Power Electronics, IEEE Transactions on*, vol. 25, pp. 1943-1957, 2010.

- [5.29] L. Yun Wei, M. Pande, N. R. Zargari, and W. Bin, "An Input Power Factor Control Strategy for High-Power Current-Source Induction Motor Drive With Active Front-End," *Power Electronics, IEEE Transactions on*, vol. 25, pp. 352-359, 2010.
- [5.30] J. R. Espinoza and G. Joos, "Current-source converter on-line pattern generator switching frequency minimization," *Industrial Electronics, IEEE Transactions* on, vol. 44, pp. 198-206, 1997.
- [5.31] D. N. Zmood and D. G. Holmes, "A generalised approach to the modulation of current source inverters," in *Power Electronics Specialists Conference*, 1998. *PESC 98 Record. 29th Annual IEEE*, pp. 739-745 vol.1, 1998.
- [5.32] D. N. Zmood and D. G. Holmes, "Improved voltage regulation for currentsource inverters," *Industry Applications, IEEE Transactions on*, vol. 37, pp. 1028-1036, 2001.
- [5.33] L. Yun Wei, W. Bin, D. Xu, and N. R. Zargari, "Space Vector Sequence Investigation and Synchronization Methods for Active Front-End Rectifiers in High-Power Current-Source Drives," *Industrial Electronics, IEEE Transactions* on, vol. 55, pp. 1022-1034, 2008.
- [5.34] Y. L. Familiant, D. G. Holmes, T. A. Lipo, and B. P. McGrath, "A General Modulation Strategy for a Five-Level Three-Phase Current Source Inverter with Regulated Intermediate DC Link Currents," in *Industry Applications Conference, 2007. 42nd IAS Annual Meeting. Conference Record of the 2007 IEEE*, pp. 581-588, 2007.
- [5.35] L. Lopes and M. F. Naguib, "Space Vector Modulation for Low Switching Frequency Current Source Converters With Reduced Low-Order Noncharacteristic Harmonics," *Power Electronics, IEEE Transactions on*, vol. 24, pp. 903-910, 2009.
- [5.36] B. Mirafzal, M. Saghaleini, and A. K. Kaviani, "An SVPWM-Based Switching Pattern for Stand-Alone and Grid-Connected Three-Phase Single-Stage Boost Inverters," *Power Electronics, IEEE Transactions on*, vol. 26, pp. 1102-1111, 2011.
- [5.37] B. S. Dupczak, A. J. Perin, and M. L. Heldwein, "Space Vector Modulation Strategy Applied to Interphase Transformers-Based Five-Level Current Source Inverters," *Power Electronics, IEEE Transactions on*, vol. 27, pp. 2740-2751, 2012.
- [5.38] D. Jingya, L. Yongqiang, W. Bin, X. Dewei, and N. R. Zargari, "A Multisampling SVM Scheme for Current Source Converters With Superior Harmonic Performance," *Power Electronics, IEEE Transactions on*, vol. 24, pp. 2436-2445, 2009.
- [5.39] Z. Ning, D. Xu, W. Bin, N. R. Zargari, M. Kazerani, and L. Fangrui, "Common-Mode Voltage Reduction Methods for Current-Source Converters in Medium-Voltage Drives," *Power Electronics, IEEE Transactions on*, vol. 28, pp. 995-1006, 2013.
- [5.40] J. Minsoo, M. Ciobotaru, and V. G. Agelidis, "A Single-Phase Grid-Connected Fuel Cell System Based on a Boost-Inverter," *Power Electronics, IEEE Transactions on*, vol. 28, pp. 279-288, 2013.
- [5.41] W. Zheng, W. Bin, X. Dewei, and N. Zargari, "Hybrid PWM for High-Power Current-Source-Inverter-Fed Drives With Low Switching Frequency," *Power Electronics, IEEE Transactions on*, vol. 26, pp. 1754-1764, 2011.
- [5.42] R. T. H. Li, H. S. H. Chung, L. Wing-Hong, and B. Zhou, "Use of Hybrid PWM and Passive Resonant Snubber for a Grid-Connected CSI," *Power Electronics, IEEE Transactions on*, vol. 25, pp. 298-309, 2010.
- [5.43] M. F. Naguib and L. Lopes, "Minimize Low-Order Harmonics in Low-Switching-Frequency Space-Vector-Modulated Current Source Converters With Minimum Harmonic Tracking Technique," *Power Electronics, IEEE Transactions on*, vol. 24, pp. 881-893, 2009.
- [5.44] M. F. Naguib and L. Lopes, "Harmonics Reduction in Current Source Converters Using Fuzzy Logic," *Power Electronics, IEEE Transactions on*, vol. 25, pp. 158-167, 2010.
- [5.45] Suroso and T. Noguchi, "Common-emitter topology of multilevel current-source pulse width modulation inverter with chopper-based dc current sources," *Power Electronics, IET,* vol. 4, pp. 759-766, 2011.
- [5.46] B. Wu. (2006). High Power Converter and Ac Drives.
- [5.47] H. R. Karshenas, H. A. Kojori, and S. B. Dewan, "Generalized techniques of selective harmonic elimination and current control in current source inverters/converters," *Power Electronics, IEEE Transactions on*, vol. 10, pp. 566-573, 1995.
- [5.48] B. Zhihong, Z. Zhongchao, and R. Xinbo, "A Natural Soft-Commutation PWM Scheme for Current Source Converter and Its Logic Implementation," *Industrial Electronics, IEEE Transactions on*, vol. 58, pp. 2772-2779, 2011.
- [5.49] S. R. Bowes and D. Holliday, "Optimal Regular-Sampled PWM Inverter Control Techniques," *Industrial Electronics, IEEE Transactions on*, vol. 54, pp. 1547-1559, 2007.
- [5.50] V. G. Agelidis, P. D. Ziogas, and G. Joos, ""Dead-band" PWM switching patterns," *Power Electronics, IEEE Transactions on*, vol. 11, pp. 522-531, 1996.

# **Chapter Six**

# DC Current Offset Compensation Technique for Grid Connected Inverters

#### 7.0. Introduction

This chapter discusses de current offset injection for integration of grid connected inverters which are applicable to medium-scale solar farms or any group of grid connected inverters confined within a relatively small area. DC current injection severely affects distribution system components such as isolation transformers, measurement units and protective relays. Several dc current offset compensation techniques claim to be appropriate to compensate dc current injected by grid connected inverters. This chapter presents a dc current offset compensation technique that exploits a simple resonance controller in combination with a proportional-integral controller to reduce capital cost of medium-scale solar farms. The proposed technique permits transformerless inverter operation within the solar farm, and uses only one large isolation transformer at the point of common coupling. Simulation and experimentation are used to demonstrate the practicability of the proposed system.

#### 7.1. Background

Power quality is an important issue for the integration of grid connected inverters. The majority of research in the power quality area is commonly focused on active harmonics filtering techniques using active power filters and voltage sag compensation using series compensators, such as dynamic voltage restorers. Issues related to dc current injection in inverter based distributed generation have been generally overlooked, despite growing concerns regarding its effect on distribution system components. One of the most severe problems caused by dc currents is saturation of the distribution transformers, resulting in waveform distortion, excessive losses, overheating and

reduced lifespan. In addition, it also may cause inaccuracy in some of the measurement and protective relays systems [6.1-4].

A possible source of dc current injection into the grid arises from the semiconductor circuits in inverter systems. The appearance of unwanted dc current components in the output currents of such systems can be attributed to several factors including minor asymmetry in the pulse width modulation signals that may be caused by numerical truncated errors in the calculation of the switching instances, nonlinearity of the switching devices, and small errors and offset drift in the voltage and current measurement sensors used to provide feedback signals for the control systems [6.5, 6].

A number of guidelines and standards imposed are to minimize the impact of dc current injection into the ac grid [6.7]. Among these standards and recommendations are guidelines to limit the allowable amount of dc current injection into the distribution network. These are listed in Table 0.1, and vary from country to country. For example, British Standard G83/1 limits dc injection to 20mA for distributed generation with phase currents below  $16A_{rms}$ [6.8]. In IEEE standard 'IEEE 929-2000' the dc limit is 0.5% of the inverter rated current [6.9]. The dc current injection in the Australian standard AS4777.2 should not exceed 0.5% of the inverter rated output current or 5mA, whichever is the greater [6.10]. These standards, however, do not guarantee that the cumulative effect of numerous power converters connected to the grid remains within the permissible limit, without harmful effects on electric system components sensitive to dc current injection by grid-connected inverters in photovoltaic systems. These are discussed in detail in Section 6.2.

Table 0.1: Collection of general guidelines that impose limits on the allowable level of dc current				
injection into the grid by photovoltaic inverter systems.				

Country	Standard	Maximum dc current permitted with transformer	Maximum dc current permitted without transformer
United Kingdom	ER G83/1	-	5mA
United States	IEEE 929-2000	<sup>1</sup> / <sub>2</sub> % of rated power inverter	<sup>1</sup> / <sub>2</sub> % of rated power inverter
Australia	AS 4777.2	<sup>1</sup> / <sub>2</sub> % of rated power inverter or 5mA (whichever greater)	<sup>1</sup> / <sub>2</sub> % of rated power inverter or 5mA (whichever greater)
Germany	DIN VDE 126	-	1000mA
Japan	Technical Guideline for the Grid Interconnection	1% of rated power inverter	1% of rated power inverter
Spain	RD 1663/2000	-	-

# 7.2. Review of dc current compensation techniques

This section provides insight into the existing methods that have been proposed in the literature to minimize dc current injection into the distribution network. These methods can be categorized as passive and active.

# 6.2.1. Passive methods

These methods use passive components, such as a coupling or isolation transformers and capacitors, to prevent dc current injection from the grid connected inverter.

# A. Isolation transformer

A line frequency isolation transformer is frequently employed at the inverter output to prevent penetration of unwanted dc currents into the grid. Also, it provides galvanic isolation between the photovoltaic system and the grid. However, a 50/60Hz transformer is large, heavy and is a substantial cost in grid connected inverter systems. Additionally, it contributes to system losses and footprint. Thus, transformerless systems have become attractive in the recent years. It has been claimed that adoption of a transformerless PV inverter system reduces the overall system costs by 25%, compared to the cost of a system that includes a transformer [6.11], and improves system efficiency by 1.5 to 2% [6.12, 13].

#### B. Blocking capacitors

Some researchers use dc and ac side capacitors to block dc current injection into the grid. For example, [6.14]present a transformerless single-phase half-bridge inverter with the neutral connected to the supply mid-point (dc node formed by connection of two dc side capacitors in series as shown in Figure 0.1(a). However, this approach is not readily extendable to three-phase inverter systems.

An alternative method that prevents dc current injection into the distribution network is to place an ac capacitor in series with the grid, with a diode network connected across the capacitor for over-voltage protection, as shown in Figure 0.1(b) [6.15]. However, this approach is risky as it may excite resonance with the ac grid and can potentially create harmonic current magnification, and hence worsen the power quality and degrade system stability.



(a) Single-phase half-bridge inverter with dc blocking capacitors



(b) Single-phase H-bridge inverter with ac blocking capacitor Figure 0.1: Blocking capacitor circuits for single-phase grid connected inverters.

#### 6.2.2. Active methods

Existing active methods for preventing dc current injection into the grid mainly rely on high precision current sensors in combination with some form of measurement or voltage transformer (VT) to control dc injection. However, the ability of a current sensor to detect a small dc component in a large ac current is expensive and challenging. These approaches are expensive to implement in medium to large scale solar farms as each inverter unit requires complex transformers and current sensors.

#### A. Current sensor and controller

A method of detecting and compensating dc offset injection in an H-bridge gridconnected inverter is proposed in [6.16]. The technique uses a voltage transformer (VT) with a 1:1 turns ratio, and an RC circuit. The primary side of the VT is connected across the grid while the secondary is connected in series with the RC circuit. With an assumption of a 1:1 turns ratio, the dc offset appears across the RC circuit capacitor and is fed back to a PI controller which in turn adjusts the inverter reference current as shown in Figure 0.2. The dc offset is eliminated, but there is no experimentation presented to validate the proposed technique. References [6.1, 17, 18] studied and verified the method introduced in [6.16], and a mathematical model-based controller with experimental validation is presented. However, the method adds cost and power loss, and increases the system footprint.



Figure 0.2: DC offset current control loop with a voltage transformer (VT).

In [6.6, 19], a method for dc current compensation in a transformerless full-bridge inverter is introduced. The method compensates the dc voltage in the inverter output, which is assumed to be proportional to the dc current offset that appears in the inverter output currents. The dc offset sensor is connected across the inverter terminals and consists of a magnetic circuit which is implemented using a low power toroidal transformer, as shown in Figure 0.3. The toroidal magnetic core is small, with winding resistance ranging from  $30\Omega$  to  $80\Omega$ . Operation depends on saturation of the reactor. When a dc voltage appears at the inverter output, in a closed-loop condition, distortion results in the reactor current. The experimental results show that this method is able to minimize the dc current component to less than 5mA for an injected grid current of  $8A_{rms}$ , which is within the IEEE standard limit.



Figure 0.3: DC compensation technique using low power toroidal transformer as a dc offset sensor.

#### B. Auto-calibrating technique of dc link sensor and controller

Figure 0.4 shows dc compensation based on an auto-calibrating technique, where a current transducer is placed between the dc link capacitor and inverter switches to measure the dc current at the output, as proposed in [6.5]. This calibration technique is based on the operational state of the inverter. A connecting state is when output current flows to the load via the dc link sensor, and the freewheeling state where there no current flows in the sensor. During the freewheeling states, the current sensor output can be sampled and used to remove any offset current present in the inverter output current. The authors presented experimental results showing this auto-calibration technique is able to limit the dc current component to less than 8.63mA for a 10A<sub>rms</sub>fundamental current component in the inverter output. It is claimed that this method offers the possibility of operation in a transformerless mode of grid connected inverter systems.

In [6.20], two dc link current sensors are used to compensate the dc offset current in the three-level half-bridge inverter shown in Figure 0.5. The upper sensor is connected between the upper terminal of dc capacitor  $C_1$  and the upper switch  $S_1$  to measure the positive dc link current during the positive half of the mains cycle. The dc offset is calibrated during the negative half cycle, when the positive dc link current drops to zero. The lower sensor is coupled between the bottom terminal of dc capacitor  $C_2$  and bottom switch  $S_4$  to measure the negative dc link current during the negative half cycle. In this case, the dc offset is removed during the positive half cycle when the negative dc link current drops to zero during the positive half mains cycle. The proposed method is able to limit the dc component to 7.08mA in a  $5A_{rms}$  inverter output current[6.14]. However, the claimed success of these methods is limited because the current sensors are placed where they measure discontinuous currents, which is impractical.



Figure 0.4: H-bridge inverter using dc-link current sensor to compensate dc current injection into the ac grid.



Figure 0.5: Three-level half bridge inverter using positive and negative dc current sensors to compensate dc current injection to the load.

# 7.3. Proposed dc current offset compensation technique for grid-connected inverters

This thesis proposes a viable technique to suppress dc current injection into the ac grid, which is applicable to medium-scale solar farms or any group of grid connected inverters confined within a relatively small area, as illustrated in Figure 0.6. The significance of this technique is that it does not rely on high-precision current measurement devices and does not use any form of complex transformer. Additionally, it allows transformerless operation of all inverters within the solar farm, and uses one power transformer rated at the full power of the solar farm. At least one inverter must be designated to perform dc offset compensation besides its basic function of active power interfacing into the ac grid. This inverter is required to compensate for all dc offsets that may exist in the total solar farm current before reaching the main interfacing transformer, and hence the risk of transformer saturation is avoided.

Referring to Figure 0.6, dc-link voltage of this inverter may come from PV panels with similar ratings as the other grid connected inverters if it is designed to perform dc offset compensation and active power interfacing into ac grid. However, a small rating of the inverter may require if it is designed to compensate dc offset only.



Figure 0.6: Generic diagram of grid connected inverters with dc offset compensation inverter in a solar farm.

The approach estimates the dc offset in the total current (excluding the current of the inverter designated for dc offset compensation) by computing the mean current for each phase independently, and forces the dc offset in the total current to zero using combinations of proportional-integral (PI) and proportional resonance (PR) controllers using phase variables (*abc*). The proportional-integral (PI) or proportional (P) controller estimates the dc offsets in the total current. The resonance controller is used to control the fundamental currents needed for exchanging the active and reactive power with the ac grid. Figure 0.7(a) to (d) summarise the possibilities of using voltage and current source inverters for dc compensation.



Figure 0.7: Possible VSI and CSI configurations for the proposed dc offset compensation method.

The polluted phase current with dc offset for phase 'a' can, for example, be expressed as:

$$i_a = I_{dc} + I_m \sin(\omega t + \varphi) \tag{6.13}$$

where  $I_{dc}$  represents the dc component of the phase current,  $I_m$  is the peak phase current,  $\varphi$  is the phase angle in radians, and  $\omega$  is the system frequency in rad/s.

The mean or average current for phase 'a' is:

$$\tilde{i}_{a} = \frac{1}{T} \sum_{k=0}^{N} i_{a}(k\omega T_{s}) = \frac{1}{T} \int_{0}^{T} i_{a}(\omega t) dt = \frac{1}{2\pi} \int_{0}^{2\pi} \left[ I_{dc} + I_{m} \sin(\omega t + \varphi) \right] d\omega t = I_{dc}$$
(6.14)

In this manner, the dc offsets for the remaining phases are calculated, where  $T_s$  is the sampling period and N is the number of samples per fundamental period T.

The dc voltage component to be injected into the modulating signal of each phase of the designated inverter, in order to generate the dc current that will counter the existing dc component in the polluted phase currents, can be obtained from a proportional or proportional-integral controller as:

$$u_{dc\_abc} = k_{pdc} \left( I_{dc}^* - I_{dc\_abc} \right) + k_{idc} \int \left( I_{dc}^* - I_{dc\_abc} \right) dt$$
(6.15)

where  $I_{dc\_abc}$  represents the vector of the dc offsets for the three phases.

The fundamental currents in response to the main active and reactive power exchange of the inverter additionally designated for dc current compensation, can be controlled using a PR controller tuned at fundamental frequency  $\omega_0$ . The PR controller is favoured over a conventional PI controller because of its effectiveness in eliminating steady-state error in the *abc* instantaneous frame. The traditional PR controller transfer function used in this chapter is expressed as:

$$y = k_p e + k_i \frac{es}{s^2 + \omega_o^2}$$
(6.16)

where  $k_p$  is the proportional gain,  $k_i$  is the integral gain,  $e = \tilde{i}_{abc}^* - \tilde{i}_{abc}$  is the error signals between the fundamental ac currents and their references, and  $\omega_o$  is the fundamental angular frequency.

The PR controller transfer function in (6.16) is discretized using a forward Euler DSP implementation, with details of this discretization provided in Chapter 4 (equations 4.21)

to 4.27). However, for general controller design, and with the aid of Figure 0.8, equation (6.16) has been transformed from the *s*-domain to the time-domain, to be incorporated into the inverter state space equations:

$$\tilde{y}(s) = \tilde{y}_1(s) + \tilde{y}_2(s)$$
 (6.17)

$$\tilde{y}_{1}(s) = k_{p}e(s) = k_{p}\left(\tilde{i}_{abc}^{*}(s) - \tilde{i}_{abc}(s)\right)$$
(6.18)

$$\tilde{y}_{2}(s) = k_{i} \frac{s}{s^{2} + \omega_{0}^{2}} e(s) = k_{i} \frac{1}{s + \frac{\omega_{0}^{2}}{s}} \left( \tilde{i}_{abc}^{*}(s) - \tilde{i}_{abc}(s) \right)$$
(6.19)

Equation (6.19) can be rewritten as:

$$s\tilde{y}_{2}(s) + \frac{1}{s}\omega_{0}^{2}\tilde{y}_{2}(s) = k_{i}\left(\tilde{i}_{abc}^{*}(s) - \tilde{i}_{abc}(s)\right)$$
(6.20)

With a change of variable,  $\tilde{y}_3(s) = \frac{1}{s} \tilde{y}_2(s)$ , the following sets of first-order differential equations describe the resonance controller suitable for incorporation into to the inverter state space model:

$$\tilde{y}_1(t) = k_p \left( \tilde{i}_{abc}^*(t) - \tilde{i}_{abc}(t) \right)$$
(6.21)

$$\frac{d\tilde{y}_2(t)}{dt} + \tilde{y}_3(t) = k_i \left( \tilde{i}^*_{abc}(t) - \tilde{i}_{abc}(t) \right)$$
(6.22)

$$\frac{d\tilde{y}_{3}(t)}{dt} = \omega_{0}^{2}\tilde{y}_{2}(t)$$
(6.23)

$$\tilde{y}(t) = \tilde{y}_1(t) + \tilde{y}_2(t)$$
 (6.24)



Figure 0.8: Block diagram that represents the proportional resonance controller.

The differential equation that describes inverter ac-side dynamics, when considering only the fundamental ac currents and voltages, is:

$$\frac{d\tilde{i}_{abc}}{dt} = -\frac{R_f}{L_f}\tilde{i}_{abc} + \frac{\tilde{v}_{abc\_con} - \tilde{v}_{abcg}}{L_f}$$
(6.25)

By setting  $\tilde{y}_{abc} = \tilde{v}_{abc\ con} - \tilde{v}_{abcg}$ , equation (6.25) can be written as:

$$\frac{d\tilde{i}_{abc}}{dt} = -\frac{R_f}{L_f}\tilde{i}_{abc} + \frac{\tilde{y}_{abc}}{L_f}$$
(6.26)

Variable  $\tilde{y}_{abc}$  is the PR controller output in Figure 0.8. Therefore the normalized reference voltages provided to the modulator, taking into account the grid voltage as feed-forward are:

$$\tilde{v}_{abc}^* = \frac{2}{V_{dc}} \left( \tilde{y}_{abc} + \tilde{v}_{abcg} \right)$$
(6.27)

Substitute  $\tilde{y}_{abc} = k_p \left( \tilde{i}_{abc}^*(t) - \tilde{i}_{abc}(t) \right) + \tilde{y}_2$  into (6.26), results in the following equation:

$$\frac{d\tilde{i}_{abc}}{dt} = -\frac{\left(R_f + k_p\right)}{L_f}\tilde{i}_{abc} + \frac{1}{L_f}\tilde{y}_2 + \frac{k_p}{L_f}\tilde{i}_{abc}^*$$
(6.28)

Differential equations (6.22), (6.23) and (6.28) can be merged into a single state space equation that can be used for selection of the PR controller gains, where the inverter internal dynamics are set by its passive components:

$$\frac{d}{dt}\begin{bmatrix} \tilde{i}_{abc} \\ \tilde{y}_{2} \\ \tilde{y}_{3} \end{bmatrix} = \begin{bmatrix} -\frac{(R_{f} + k_{p})}{L_{f}} & \frac{1}{L_{f}} & 0\\ -k_{i} & 0 & -1\\ 0 & \omega_{0}^{2} & 0 \end{bmatrix} \begin{bmatrix} \tilde{i}_{abc} \\ \tilde{y}_{2} \\ \tilde{y}_{3} \end{bmatrix} + \begin{bmatrix} \frac{k_{p}}{L_{f}} \\ k_{i} \\ 0 \end{bmatrix} \begin{bmatrix} \tilde{i}_{abc} \\ \tilde{i}_{abc} \end{bmatrix}$$
(6.29)

The current controller designs for the other inverters that are responsible only for active power injection into grid are similar.

Figure 0.9 summarises the control system for the inverter designated for dc current offset compensation. The control of fundamental current and dc offset can be performed with independent controllers as described, or both components can be controlled using a single PR controller, provided the proportional part of the PR controller is sufficient for suppressing the dc offsets to zero.



Figure 0.9: Proposed dc current offset compensation control system.

# 7.4. Simulation result

To demonstrate the effectiveness of the proposed dc current offset compensation technique presented in this thesis, the test system comprises of two grid-connected inverters (label'1' and '2') connected to the same bus as shown in Figure 0.9. Both inverters 1 and 2 are operated under current control mode, where inverter 1 mimics output currents of the solar farm that contains dc current component in either one or all of the phases. Injection of the dc offsets with unknown magnitudes into the output currents of inverter 1 is achieved by adding dc offsets to the modulating signals produced by the current controller. The output currents of inverter 1 are used to extract the dc components in each phase that are used as feedback signals to the dc offset controller of inverter 2. The aim of the dc offset controller in inverter 2 is to suppress dc offset to zero in the output currents before entering the main interfacing transformer that

connects both inverters to the grid. The parameters used in the simulations are listed in Table 0.2. Two configurations have been simulated, while only one configuration has been used for experimental validation. The first configuration uses two voltage source inverters: one creates a dc offset and the second performs dc offset compensation. The second configuration uses the current source inverter for offset compensation. However, experimental validation is carried out for the first configuration only.

		Configuration I		Configuration II	
Parameters		Values			
DC-link voltage, V <sub>dc</sub>	V	300	300	350	-
DC-link current, I <sub>dc</sub>	Α	-	-	-	10
Power capability, S	VA	1000	1000	1000	1000
Switching frequency, f <sub>sw</sub>	Hz	2100	2100	2100	2100
Grid voltage per phase, V <sub>g</sub>	V	110	110	110	110
Filter inductor of inverter 1, $L_{fl}$	mH	20	-	20	-
Filter capacitor of inverter 1, C <sub>fl</sub>	μF	-	-	-	-
Filter inductor of inverter 2, L <sub>f2</sub>	mH	-	10	-	3
Filter capacitor of inverter 2, C <sub>f2</sub>	μF	-	70	-	150
Proportional gain of PR control, K <sub>p</sub>	Ω	4	8	8	0.5
Integral gain of PR control, K <sub>i</sub>	$\Omega s^{-1}$	900	1200	650	50

Table 0.2: Parameters used in the simulations.

#### 6.4.1. Configuration I: Combination of VSIs

Figure 0.10 shows simulation results of inverter 1 when acting as an inverter that contains some dc current offset. The reference current,  $i_{abc1}^*$  is set at 2A with dc current offsets injected into each phase. The dc current offsets are 5%, 10% and -5% of the fundamental current, and are injected into phases *a*, *b* and *c* respectively. Figure 0.10(a) illustrates the phase current with injected dc current offsets. Figure 0.10(b) is the fundamental phase current and dc offset components are shown in Figure 0.10(c). The modulating signals of inverter 1 are shown in Figure 0.10(d).

The currents and voltages of inverter 2 are shown in Figure 0.11 (a) to (g). Using the proposed dc compensation strategy in Figure 0.9, the reference current in inverter 2,  $i_{abc2}^*$  is generated from  $i_{abc1}^*$  of inverter 1 adding the extracted dc offset from the output

current of inverter 1,  $i_{abc1}$ . The output currents of inverter 2,  $i_{abc2}$  with their respective references,  $i_{abc2}^*$  are shown in Figure 0.11(a). Figure 0.11(b) to (d) compare the phase current of inverter 2 ( $i_{abc2}$ ) with reference of  $i_{abc1} \cdot i_{abc2}$  is the inverse of  $i_{abc1}$ , with the same amount of dc offset. Adding both  $i_{abc1}$  and  $i_{abc2}$  eliminates the dc offset, resulting in sinusoidal output current,  $i_{abco}$ . The modulating signals and the resultant output currents are shown in Figure 0.11(e) and (f) respectively while the grid voltage is illustrated in Figure 0.11(g).





(c) DC offset components of the three phases



(b) Fundamental output current of inverter 1(when dc component is filtered)



(d) Modulating signals of inverter 1 when dc offsets of 5%, 10% and -5% are injected

Figure 0.10: Simulation results of configuration I showing inverter 1 is mimicked an inverter that contains dc current offset components in its output current.





Figure 0.11: Simulation results of configuration I showing a VSI as a dc compensation inverter.

# 6.4.2. Configuration II: VSI and CSI combination



Figure 0.12(a) to (f) show simulation results for configuration II where the VSI mimics an inverter with a dc current offset component while the CSI is the dc current compensation inverter. The results shown in Figure 0.11 confirm that configuration I with a VSI as a dc current compensation inverter is superior, being able to compensate a higher level of dc offset while retaining balanced ac output currents. When a CSI is used as the dc compensation inverter, similar levels of dc offset as in Figure 0.10(c) are not achievable since instability and very poor dynamic performance results.



Figure 0.12: Simulation results for configuration II showing CSI as a dc compensation inverter.

#### 7.5. Experimental verification

The parameters used in the experimental validation are listed in Table 0.3, and Figure 6.13 and 6.14 illustrate the practical implementation of the proposed dc compensation technique using configuration I.

		Inverter 1	Inverter 2	
Parameters		Values		
DC-link voltage, V <sub>dc</sub>	V	200	200	
Power capability, S	VA	1000	1000	

Table 0.3: Experimental Parameters.

Switching frequency, f <sub>sw</sub>	Hz	4200	4200
Line-to-line grid voltage, V <sub>g</sub>	V	86.6	86.6
Filter inductor of inverter 1, $L_{fl}$	mH	20	-
Filter capacitor of inverter 1, $C_{fl}$	μF	-	-
Filter inductor of inverter 2, $L_{f2}$	mH	-	10
Filter capacitor of inverter 2, $C_{f2}$	μF	-	70
Proportional gain of PR control, K <sub>p</sub>	Ω	1.2	0.8
Integral gain of PR control, K <sub>i</sub>	$\Omega s^{-1}$	180	144



Figure 0.13: Hardware diagram of test-rig



Figure 0.14: Experimental test rig setup of grid-connected inverters in parallel arrangement.

# 6.5.1. Experimental results: inverters 1 and 2 without dc offset injection

Figure 0.15(a) to (h) show the experimental results when inverters 1 and 2 operate in a parallel arrangement, with both injecting power into the grid, using one interfacing transformer. These results show that both inverters are able to operate satisfactorily and produce clean sinusoidal output currents. Figure 0.15(a) and (b) display inverters 1 and 2 output currents superimposed over their corresponding references currents, with currents of 3A per phase. The modulating signals for both inverters are stable as shown in Figure 0.15(c) and (d). Figure 0.15(e) and (f) show the three-phase output currents from inverters 1 and 2. The resultant three-phase output currents of the two inverters entering the interfacing transformer, and grid voltage are displayed in Figure 0.15(g) and (h) respectively.



Figure 0.15: Experimental results illustrating the current and voltage waveforms when two gridconnected VSIs exchange active and reactive power with the grid voltage without the introduction of dc current offset.

# 6.5.2. Experimental results: the proposed dc current compensation technique with different levels of dc current injected into the output of inverter 1

Figure 0.16 to 6.20 show the experimental results used to validate the proposed dc current compensation technique, with different current injection levels in the output of inverter 1. Inverter 2 performs the proposed dc compensation.

Figure 0.16 presents the experimental results when the dc components added into the modulating signals of the inverter 1 are adjusted randomly to give dc offsets of 19.6%, - 2.67% and -26.67% in phases a, b and c (these percentage ratios are expressed with respect to a peak current of 3A). The results in Figure 0.16 (a) to (d) show the three-phase output currents of inverters 1 and 2, including the amount of dc offsets in the output currents of inverter 1 that need to be compensated, inverter 1 three-phase output currents, inverter 2 three-phase output currents superimposed on their reference currents, and the resultant output currents after compensation and seen by the coupling transformer. Figure 0.16(e) shows the grid voltage at the point of common coupling. These results show that the dc offsets in the output currents with near zero dc offset. These results are achieved despite a significant level of dc offset being injected into the output of inverter 1.



Figure 0.16: Experimental results demonstrating the practicability of the proposed dc offset compensation for grid connected inverters (Component of dc offsets injected relative to output current in each phase: a=0.59A, b=-0.08A and c=-0.8A).



(g) Compensated three-phase output currents entering interfacing transformer(2A/div, 5ms) Figure 0.17 shows the experimental results when dc offsets of -4.7%, 2.3% and 13.3% relative to a peak phase current of 3A are injected into the output currents of inverter 1. These results are in agreement with the previous results presented in Figure 0.16 as inverter 2 is able to compensate the injected dc offsets.



(n) Compensated three-phase output currents entering interfacing transformer(2A/div, 5ms) Figure 0.17 (d) shows that the modulating signals of inverter 1 contain the necessary dc component, while



(u) Compensated three-phase output currents entering interfacing transformer(2A/div, 5ms) Figure 0.17(e) and (f) show the output current of inverter 2 used to counter the dc offset introduced by inverter 1 output. The resultant output phase currents entering the interfacing transformer are sinusoidal, meaning there is no risk of transformer dc saturation.



Figure 0.17: Experimental results demonstrating the practicability of the proposed dc offset compensation for grid connected inverters (Component of dc offsets injected relative to the output current in each phase: a=-0.14A, b=0.07A and c=0.4A).



Figure 0.18 and 6.19 shows the results when relatively large dc current offsets are injected into the output phase currents of inverter 1 (-18.6%, 25% and -33.3% measured relative to the peak phase current of 3A). These results show that inverter 2 is able to compensate the high level of dc current injected into the output phase currents of inverter 1, as the resultant output phase currents entering the interfacing transformer become sinusoidal.





Figure 0.18: Experimental results demonstrating the practicability of the proposed dc offset compensation for grid connected inverters (Component of dc offsets injected relative to the output current in each phase: a=-0.56A, b=0.75A and c=-1A).



Figure 0.19: Experimental results demonstrating the practicability of the proposed dc offset compensation for grid connected inverters (Component of dc offsets injected relative to the output current in each phase: a=-0.56A, b=0.75A and c=-1A).

Experimental results with increasingly higher levels of dc injection (16.6%, -46.3% and 28.3% measured relative to the peak phase current of 3A) are shown in



(g) Compensated three-phase output currents entering interfacing transformer(2A/div, 5ms)

Figure 0.20. These results show that the proposed technique is able to compensate such high levels of dc current in the output phases of inverter 1. The results in Figure 6.19 to

6.20 demonstrate the viability of the proposed technique, even in applications with relatively large dc components in the phase currents, as expected in a medium-scale solar farm with transformerless grid connected inverters.



their corresponding references



(n) Compensated three-phase output currents entering interfacing transformer(2A/div, 5ms)

Figure 0.20: Experimental results demonstrating the practicability of the proposed dc offset compensation for grid connected inverters (Component of dc offsets injected relative to the output current in each phase: a=0.5A, b=-1.2A and c=0.85A).

### 7.6. Summary

This chapter discussed passive and active dc offset compensation techniques that claim to be applicable to minimize dc current injected into grid connected inverters. A new compensation technique is proposed which is applicable to medium-scale solar farms or any group of grid connected inverters confined within a relatively small area. This dc offset compensation technique exploits a simple proportional-resonance controller that aims to suppress dc offsets to zero in the output currents of grid-connected inverters before entering the main 50/60Hz interfacing transformer. At least one inverter is designated to perform dc offset compensation as well as its basic function of active power interfacing to the ac grid. The significance of this technique is it reduces the capital cost of a medium-scale solar farm since it enables transformerless operation of all inverters within the solar farm, and uses only one power transformer at the point of common coupling. Additionally, this technique does not rely on high-precision current measurement devices and does not use any form of complex transformer. To substantiate the proposed dc current compensation technique through simulation and experimentation, one inverter mimicked the output currents that contain dc current components in either one or all of the phases, while the second inverter performed the proposed dc compensation.

#### References

[6.1] T. L. Ahfock and L. Bowtell, "DC offset elimination in a single-phase gridconnected photovoltaic system," presented at the AUPEC 2006: 16th Australasian Universities Power Engineering Conference, Melbourne, Australia, 2006.

- [6.2] University of Strathclyde, A technical report on "An Investigation on DC Injection into Low-voltage AC System," Project Steering Group Work Stream 4 Project WS4-P11, Contract number DG/CG/00002/00/00, DTI/OFGEM Distributed Generation Co-ordinating Group, June 2005.
- [6.3] S.-B. Lee and T. G. Habetler, "An on-line stator winding resistance estimation technique for temperature monitoring of line-connected induction machines," in *Industry Applications Conference, 2001. Thirty-Sixth IAS Annual Meeting. Conference Record of the 2001 IEEE*, 2001, pp. 1564-1571.
- [6.4] P. Aylott, "Stray current is for life-not just for Christmas. Stray current corrosion management strategies for DC traction systems," in DC Traction Stray Current Control-Offer a Stray a Good Ohm? (Ref. No. 1999/212), IEE Seminar on, 1999, pp. 7/1-7/6.
- [6.5] M. Armstrong, D. J. Atkinson, C. M. Johnson, and T. D. Abeyasekera, "Auto-Calibrating DC Link Current Sensing Technique for Transformerless, Grid Connected, H-Bridge Inverter Systems," *Power Electronics, IEEE Transactions on*, vol. 21, pp. 1385-1393, 2006.
- [6.6] G. Buticchi, E. Lorenzani, and G. Franceschini, "A DC Offset Current Compensation Strategy in Transformerless Grid-Connected Power Converters," *Power Delivery, IEEE Transactions on*, vol. 26, pp. 2743-2751, 2011.
- [6.7] S. B. Kjaer, J. K. Pedersen, and F. Blaabjerg, "A review of single-phase gridconnected inverters for photovoltaic modules," *Industry Applications, IEEE Transactions on*, vol. 41, pp. 1292-1306, 2005.
- [6.8] E. N. Association, "Recommendations for the connection of small-scale embedded generators (up to 16 A per phase) in parallel with public low-voltage distribution networks, Engineering Recommendation G83/1, UK," ed, 2003.
- [6.9] "IEEE Recommended Practice for Utility Interface of Photovoltaic (PV) Systems," *IEEE Std 929-2000*, p. i, 2000.
- [6.10] T. A. Standard, "AS4777.2-2005 Grid Connection of Energy Systems via Inverters- Inverter Requirements," ed, 2005.
- [6.11] M. Calais, V. G. Agelidis, and M. S. Dymond, "A cascaded inverter for transformerless single-phase grid-connected photovoltaic systems," *Renewable Energy*, vol. 22, pp. 255-262, 2001.
- [6.12] H. Haeberlin, "Evolution of Inverters for Grid connected PV-Systems from 1989 to 2000," *measurement*, vol. 2, p. 1, 2001.
- [6.13] T. Kerekes, R. Teodorescu, P. Rodriguez, G. Vazquez, and E. Aldabas, "A New High-Efficiency Single-Phase Transformerless PV Inverter Topology," *Industrial Electronics, IEEE Transactions on*, vol. 58, pp. 184-191, 2011.
- [6.14] F. Berba, D. Atkinson, and M. Armstrong, "A review of minimisation of output DC current component methods in single-phase grid-connected inverters PV applications," in *Environment Friendly Energies and Applications (EFEA), 2012* 2nd International Symposium on, 2012, pp. 296-301.
- [6.15] W. M. Blewitt, D. J. Atkinson, J. Kelly, and R. A. Lakin, "Approach to low-cost prevention of DC injection in transformerless grid connected inverters," *Power Electronics, IET*, vol. 3, pp. 111-119, 2010.

- [6.16] R. Sharma, "Removal of DC offset current from transformerless PV inverters connected to utility," presented at the 40th International Universities Power Engineering Conference (UPEC 2005), Cork, Ireland, 2005.
- [6.17] L. Bowtell and A. Ahfock, "Direct current offset controller for transformerless single-phase photovoltaic grid-connected inverters," *Renewable Power Generation, IET,* vol. 4, pp. 428-437, 2010.
- [6.18] L. A. Bowtell and A. L. Ahfock, "Dynamic analysis of a DC offset controller for grid-connected inverters," in Universities Power Engineering Conference (AUPEC), 2010 20th Australasian, 2010, pp. 1-5.
- [6.19] G. Buticchi, G. Franceschini, E. Lorenzani, C. Tassoni, and A. Bellini, "A novel current sensing DC offset compensation strategy in transformerless grid connected power converters," in *Energy Conversion Congress and Exposition*, 2009. ECCE 2009. IEEE, 2009, pp. 3889-3894.
- [6.20] F. Berba, D. Atkinson, and M. Armstrong, "Minimisation of DC current component in transformerless Grid-connected PV inverter application," in *Environment and Electrical Engineering (EEEIC), 2011 10th International Conference on*, 2011, pp. 1-4.
# **Chapter Seven**

# Grid Integration of a Current Source Inverter

#### 7.0. Introduction

The relevant attributes of the current source inverter for grid integration of photovoltaic systems are highlighted in chapter five, including discussions of their modulation techniques. This chapter presents the control structures needed to enable safe operation of current source inverters in grid mode, meeting all the relevant standards and grid code requirements. Additionally, the feasibility of the CSI to operate with a passive network with no generation, where the inverter needs to set the voltage (as in the case with the VSI and synchronous machines) during black-start is demonstrated. The discussion presented in this chapter shows that when a current source inverter is correctly controlled, it can provide all the flexibility needed for grid operation, including provision of leading and lagging reactive power needed for voltage support and survival during low-voltage ride-through. However, the issues related to low-voltage fault ride-through are not discussed in this chapter. This chapter also presents simulations and experimentation that demonstrates the suitability of the direct regular sample PWM as proposed in chapter five, for grid application.

#### 7.1. Background

Distributed power generation based on renewable sources is experiencing fast growth and development worldwide, especially in countries such as China, United States, Germany and Japan [7.1]. Grid connected inverters play an important role in the integration of distribution generation onto ac grids. Such integration is governed by strict grid codes, which include the control of active and reactive power, permissible levels of harmonics, and other power quality aspects. Among two basic inverter technologies available, the current source inverter (CSI) is perceived to be suitable for grid integration of multi-megawatt wind turbines and photovoltaic systems [7.2, 4-7]. For example, the CSI provides a simple topological solution and satisfactory performance in grid mode, giving sinusoidal current, fully controlled power factor, and natural protection against dc side short-circuit faults.

Although, as yet, the CSI does not play a major role as a grid interfacing unit in distributed generation systems, it is a proven approach in high power, medium voltage ac drives systems with ac voltages of 2.3kV to 13.8kV [7.8, 9]. It is estimated that a minimum of 700 units of large CSI-fed drives are annually produced worldwide [7.10]. The features of a simple converter structure, low switch count, low dv/dt and reliable over-current/short circuit protection make the CSI competitive compared to the voltage source inverter (VSI) and the multilevel inverter.

In a photovoltaic system, the CSI has been advocated as a means of improving the effectiveness of maximum power-point tracking (MPPT), as the number of series connected PV in each string is reduced [7.11]. This reduces the impact of partial shading, and increases system efficiency by using one conversion stage instead of two, therein benefiting from the voltage boost capability of the CSI[7.12-18].

The single-phase CSI for photovoltaic grid integration is proposed in [7.11-14, 17, 18] which introduce various transformerless topologies to minimize the common-mode current that arises through the stray capacitance between the PV array and ground. Also, the proposed single-phase topologies need to cater for even harmonics on the dc side [7.17, 18]. These need to be eliminated since they affect MPPT, reduce PV inverter lifetime, and are associated with odd harmonics on the grid side.

It is advantageous to use a three-phase system instead of single-phase one because there are no even harmonics or  $3^{rd}$  harmonic on the grid side. Several applications of three-phase CSI in photovoltaic systems are proposed in [7.15, 16, 19, 20]. Analysis and modelling of a grid connected CSI, using PI control in the *dq*-frame and carrier-based SPWM, are presented in [7.15]. However, the controller only consists of a current control loop, which affects system reliability.

In [7.16], the proposed system consists of only one control loop for maximum power point tracking, and is claimed to produce an ac output current with a total harmonic distortion (THD) of 4.5%. However, with varying weather conditions, an output current control loop is important to limit the current and quickly recover the grid current variation. In [7.19], a modified carrier-based SPWM technique is proposed which adds a short-circuit pulse to the conventional SPWM to prevent an open-circuit CSI at any instant, and which is useful in adjustment of the PV-array voltage. Although a simple MPPT control structure is introduced, limited experimental results and information on the control system are presented [7.19].

In this chapter, a single-stage grid connected PV system based on a three-phase CSI is proposed. The control system uses the proposed direct regular-sampled pulse width modulation (DRSPWM) technique to regulate active and reactive power exchanges with the ac grid under various operating conditions, giving an ac output current with acceptable THD according to the IEEE 929-2000 standard for utility interfacing of photovoltaic systems [7.21].

# 7.2. Control system in island and grid modes

In principle, a CSI injects its output active and reactive power into the grid, assuming the grid is sufficiently large and able to absorb the injected powers without affecting its power balance [7.1, 22, 23]. In this mode, the grid dictates the ac voltage and frequency, while the CSI is seen as a slave that uses a phase locked loop to synchronize with the grid and manipulate its output currents relative to the grid voltage to achieve the desired active and reactive power exchange[7.24]. Whilst in an islanding mode, the CSI operates as a static synchronous generator that needs to set the ac frequency and gradually build-up its output voltage from 0 to rated voltage. It then establishes a stiff ac bus, where the loads are connected. During black start and operation in islanding mode, the CSI is controlled in a voltage mode where it sets the ac voltage.

#### 7.2.1. Voltage control in island operation

The inverter voltage controller in island mode controls both the frequency and the quality of the output voltage supplied to the load. Two voltage control methods are considered for the three-phase CSI. These are proportional-resonant control in the *abc* frame, and proportional integral control in the synchronous reference d-q frame.



Figure 0.1: Circuit diagram of CSI in island mode.

# A. Proportional-integral control in the abc frame

Figure 0.1illustrates a power circuit of a three-phase CSI connected to a passive load. From Figure 0.1, the dynamic equations that describe the CSI in the *abc* frame for island mode of operation are:

$$\frac{dv_{abco}}{dt} = \frac{i_{abci} - i_{abco}}{C_c}$$
(7.30)

$$v_{abco} = Ri_{abco} \tag{7.31}$$

Let

$$u = i_{abci} - i_{abco} \tag{7.32}$$

where  $v_{abco}$ ,  $i_{abci}$  and  $i_{abco}$  represent the output voltages, inverter currents and load currents respectively, while  $C_f$  and R are the filter capacitance and the load resistance. Figure 0.2 illustrates a proportional-resonant controller in the *abc* frame that is used to control the output voltage. The controlled variables are independent between phases, hence this controller can be applied to single and three phase systems. The PR controller transfer function in (7.33) is discretized using a forward Euler method for DSP implementation, where discretization details are presented in chapter four (equations 4.21 to 4.27).

$$y = k_p e + \int \left(k_i e - \int \omega_o^2 z dt\right) dt$$
(7.33)

From (7.30) and (7.32), whilst including the output voltage error into the PR controller, u is obtained from (7.33):

$$u = k_{pv} \left( v_{abco}^* - v_{abco} \right) + \int \left( k_{iv} \left( v_{abco}^* - v_{abco} \right) - \int \omega_o^2 z_v dt \right) dt$$
(7.34)

Substituting (7.34) into (7.30)

$$\frac{dv_{abco}}{dt} = \frac{k_{pv}v_{abco}^*}{C_f} - \frac{k_{pv}v_{abco}}{C_f} + \frac{z_v}{C_f}$$
(7.35)

where

$$z_{v} = \int \left( k_{iv} \left( v_{abco}^{*} - v_{abco} \right) - w_{v} \right) dt$$
 (7.36)

and

$$w_{v} = \int \omega_{o}^{2} z_{v} dt \tag{7.37}$$

The derivatives of (7.36) and (7.37) are:

$$\frac{dz_{v}}{dt} = k_{iv} \left( v_{abco}^{*} - v_{abco} \right) - w_{v}$$
(7.38)

$$\frac{dw_v}{dt} = \omega_o^2 z_v \tag{7.39}$$

From (7.35), (7.38) and (7.39), the state space model of the voltage loop controller can be written in the form:  $\frac{dx(t)}{dt} = Ax(t) + Bu(t)$ 

Figure 0.2: Control diagram of PR voltage controller for current source inverter.

# B. Proportional-integral (PI) controller in island mode

The output voltage regulator is a PI controller in the synchronous rotating frame. The three-phase ac quantities are transformed into dc components in the dq-reference frame to eliminate steady-state error. This control system is illustrated in Figure 0.3. The stability is improved at the expense of response time and harmonic rejection. Referring to Figure 0.1, the dynamic equations in the dq-reference frame are:

$$\frac{dv_{od}}{dt} = \frac{1}{C_f} \left[ i_{id} - i_{od} - \omega C_f v_{oq} \right]$$
(7.41)

$$\frac{dv_{oq}}{dt} = \frac{1}{C_f} \left[ i_{iq} - i_{oq} + \omega C_f v_{od} \right]$$
(7.42)

To facilitate voltage regulator design, let

$$u_{d} = \frac{1}{C_{f}} \left( i_{id} - i_{od} - \omega C_{f} v_{oq} \right)$$
(7.43)

$$u_{q} = \frac{1}{C_{f}} \left( i_{iq} - i_{oq} + \omega C_{f} v_{od} \right)$$
(7.44)

The variables  $u_d$  and  $u_q$  can be obtained from the proportional-integral (PI) controller as follows:

$$u_{d} = k_{p} \left( v_{od}^{*} - v_{od} \right) + k_{i} \int \left( v_{od}^{*} - v_{od} \right) dt$$
(7.45)

$$u_{q} = k_{p} \left( v_{oq}^{*} - v_{oq} \right) + k_{i} \int \left( v_{oq}^{*} - v_{oq} \right) dt$$
(7.46)

where  $k_p$  and  $k_i$  are the proportional and integral gains of the voltage regulator in the *d* and *q* channels.

Replacing the integral parts in (7.45) and (7.46) with  $y_d$  and  $y_q$ , and substituting the resultant expressions(7.41) and (7.42), gives:

$$\frac{dv_{od}}{dt} = \frac{1}{C_f} \Big[ k_p v_{od}^* - k_p v_{od} + y_d \Big]$$
(7.47)

$$\frac{dv_{oq}}{dt} = \frac{1}{C_f} \left[ k_p v_{oq}^* - k_p v_{oq} + y_q \right]$$
(7.48)

$$\frac{dy_d}{dt} = k_i \left( v_{od}^* - v_{od} \right) \tag{7.49}$$

$$\frac{dy_q}{dt} = k_i \left( v_{oq}^* - v_{oq} \right) \tag{7.50}$$

Differential equations (7.47) to (7.50) can be re-arranged in state space form, x'(t) = Ax(t) + Bu(t) as:

$$\frac{d}{dt}\begin{bmatrix} v_{od} \\ y_d \\ v_{oq} \\ y_q \end{bmatrix} = \begin{bmatrix} -\frac{k_p}{C_f} & \frac{1}{C_f} & 0 & 0 \\ -k_i & 0 & 0 & 0 \\ 0 & 0 & -\frac{k_p}{C_f} & \frac{1}{C_f} \\ 0 & 0 & -k_i & 0 \end{bmatrix} \begin{bmatrix} v_{od} \\ y_d \\ v_{oq} \\ y_q \end{bmatrix} + \begin{bmatrix} \frac{k_p}{C_f} & 0 \\ k_i & 0 \\ 0 & \frac{k_p}{C_f} \\ 0 & k_i \end{bmatrix} \begin{bmatrix} v_{od}^* \\ v_{od}^* \\ v_{oq}^* \end{bmatrix}$$

(7.51)



Figure 0.3: CSI control system in island mode using PI control in the*dq*-frame.

# 7.2.2. Simulation and experimental operational verification of the CSI in island mode

PR control in the *abc*, and PI control in the dq-frame for operation of a CSI in island mode are considered using simulation and experimentation analysis. The parameters used for simulation and experimental results are listed in Table 0.1.

Parameters	Values		
DC-link current $(I_{dc})$	8A		
AC output voltage $(V_o)$	90V		
Nominal frequency $(f_o)$	50Hz		
Switching frequency $(f_{sw})$	2.1kHz		
Capacitor filter ( $C_f$ )	70µF		
Inductor filter $(L_f)$	3mH		
Load (R)	22Ω		

Table 0.1: Parameters used in simulation and experimental investigation

# A. Simulation results

Figure 0.4 and 7.5 illustrate the voltage, power and current waveforms when a PR controller in the *abc* frame and a PI controller in the *dq*-frame, are used to control the CSI in island mode, respectively.



Figure 0.4: Simulation results illustrating a three-phase CSI in island mode using PR voltage-control in the *abc* frame.



Figure 0.5: Simulation results illustrating the three-phase CSI in island mode using *dq*-frame PI voltage-control.

# B. Experimental results

To verify the results in Figure 0.5, a laboratory test-rig with the parameters in Table 0.1 was implemented and the corresponding results are presented in Figure 0.6 to 7.8. Figure 0.6(a) to (c) are experimental results from the dSPACE1006 DSP-platform for the output voltages, active and reactive powers delivered to the load, and the respective modulating signals using the PR voltage controller in an *abc* frame. The output voltages using PI control in the *dq*-axes are compared to their corresponding references in Figure 0.7(a). With a dc-link current of 8A, the supplied active power to the load is nearly 500W while the reactive power is maintained at 0VAr, as demonstrated in Figure 0.7(b). Figure 0.8(a) and (c) show three-phase load voltages and currents and their respective magnification during transient and steady-state conditions. The modulation

index plot is shown in Figure 0.8(c). Both controllers track their respective reference voltages and deliver the same active power to the load, with near unity power factor.



(c) Modulating signals

Figure 0.6: CSI experimental results in island mode using PR voltage control in an *abc* frame.



Figure 0.7:CSI experimental results in island mode using PI voltage control in the dq-frame (part a).



(c) Modulation index

Figure 0.8: CSI experimental results in island mode using PI voltage control in the dq-frame (part b).

# 7.3. Current control system for a grid-connected CSI

This section discusses the control systems used to assess the viability of the proposed DRSPWM technique for a grid connected CSI. Details are presented of the control variables manipulated in order to regulate active and reactive powers inverter exchanges with the ac grid.



Figure 0.9: Schematic of grid-connected three-phase CSI.

From Figure 0.9, the dynamic equations describing the CSI in the d-q axes, where phase 'a' of the grid voltage is aligned with the d-axis are:

$$\frac{di_{od}}{dt} = -\frac{R_F}{L_F}i_{od} + \frac{\left(v_{cd} - v_{od} + \omega L_F i_{oq}\right)}{L_F}$$
(7.52)

$$\frac{di_{oq}}{dt} = -\frac{R_F}{L_F}i_{oq} + \frac{\left(v_{cq} - v_{oq} - \omega L_F i_{od}\right)}{L_F}$$
(7.53)

$$\frac{dv_{cd}}{dt} = \frac{i_{id} - i_{od} + \omega C_F v_{cq}}{C_F}$$
(7.54)

$$\frac{dv_{cq}}{dt} = \frac{i_{iq} - i_{oq} - \omega C_F v_{cd}}{C_F}$$
(7.55)

Equations (7.52) and (7.53) describe the grid current dynamics in terms of the grid and filter capacitor voltages. Equations(7.54) and (7.55) express the capacitor voltage dynamics in terms of grid, filter capacitor and inverter currents where  $v_{od}$  and  $v_{oq}$ , and

 $v_{cd}$  and  $v_{cq}$  represent the *d*-*q* components of the grid and capacitor voltages;  $i_{od}$  and  $i_{oq}$ , and  $i_{id}$  and  $i_{iq}$  represent the *d*-*q* components of the grid and inverter currents; and  $R_F$ ,  $L_F$ and  $C_F$  are the resistance, inductance and capacitance of the ac series and shunt filters.

### 7.3.1. Power control loop

The average power control method provides high quality sinusoidal output current and controls the average power flow. By assuming a fixed magnitude grid voltage, inverter active and reactive power exchange with the grid can be regulated directly by controlling the *d*-*q* components of the grid current. The reference currents the CSI needs to inject into the grid at voltage  $v_o = v_{od} + jv_{oq}$  in order to exchange specific active and reactive powers can be expressed as:

$$\begin{bmatrix} \dot{i}_{od}^{*} \\ \dot{i}_{oq}^{*} \end{bmatrix} = \frac{2}{3} \frac{1}{v_{od}^{2} + v_{oq}^{2}} \begin{bmatrix} v_{od} & v_{oq} \\ v_{oq} & -v_{od} \end{bmatrix} \begin{bmatrix} P^{*} \\ Q^{*} \end{bmatrix}$$
(7.56)

#### 7.3.2. Current control loop

Figure 0.10 presents the control system for grid-connected CSI which consists of a phase locked loop, outer current regulator, inner voltage regulator, and the proposed DRSPWM. The presented control structure in Figure 0.10uses several feedback loops with PI controllers in the synchronous frame, and feed-forward control to enhance the system's ability to reject disturbances. The proposed DRSPWM is used to generate the CSI switch gating signals. With this arrangement, system stability, dynamic response and response time are improved, with output harmonic elimination. To control the power flow,  $I_o$  is the outer control variable while for the inner loop, capacitor voltage,  $V_c$  enhances controller bandwidth.



Figure 0.10: Block diagram of the control systems of a grid connected CSI.

The current regulator adjusts the magnitude and phase of the capacitor voltage relative to the grid voltage in order to fulfil the current commands in (7.56), designed based on equations (7.52) and (7.53). Thus the output current regulator autonomously provides a set-point to the inner loop that will be designed later based on (7.54) and (7.55). To facilitate current regulator design, let  $w_d = v_{cd}^* - v_d + \omega L_F i_{oq}$  and  $w_q = v_{cq}^* - v_q - \omega L_F i_{od}$ , where the new variables  $w_d$  and  $w_q$  are obtained from the proportional-integral (PI) controller as follows:

$$w_{d} = k_{P1} \left( i_{od}^{*} - i_{od} \right) + k_{I1} \int \left( i_{od}^{*} - i_{od} \right) dt$$
(7.57)

$$w_q = k_{P1} \left( i_{oq}^* - i_{oq} \right) + k_{I1} \int \left( i_{oq}^* - i_{oq} \right) dt$$
(7.58)

where  $k_{PI}$  and  $k_{II}$  are proportional and integral gains of the current regulators in the *d* and *q* channels.

Replacing the integral parts of (7.57) and (7.58) with  $z_d$  and  $z_q$ , and substituting the resultant expressions in (7.52) and (7.53), gives:

$$\frac{di_{od}}{dt} = -\frac{\left(R_F + k_{P1}\right)}{L_F}i_{od} + \frac{z_d}{L_F} + \frac{k_{P1}}{L_F}i_{od}^*$$
(7.59)

$$\frac{di_{oq}}{dt} = -\frac{\left(R_F + k_{P1}\right)}{L_F}i_{oq} + \frac{z_q}{L_F} + \frac{k_{P1}}{L_F}i_{oq}^*$$
(7.60)

$$\frac{dz_d}{dt} = k_{I1} \left( i_{od}^* - i_{od} \right)$$
(7.61)

$$\frac{dz_q}{dt} = k_{I1} \left( i_{oq}^* - i_{oq} \right)$$
(7.62)

Differential equations (7.59) to (7.62) can be re-arranged in state space form, x'(t) = Ax(t) + Bu(t) as:

$$\frac{d}{dt}\begin{bmatrix}i_{od}\\z_{d}\\i_{oq}\\z_{q}\end{bmatrix} = \begin{bmatrix}-\frac{(R_{F}+k_{P1})}{L_{F}} & \frac{1}{L_{F}} & 0 & 0\\-k_{I1} & 0 & 0 & 0\\0 & 0 & -\frac{(R_{F}+k_{P1})}{L_{F}} & \frac{1}{L_{F}}\end{bmatrix}\begin{bmatrix}i_{od}\\z_{d}\\i_{oq}\\z_{q}\end{bmatrix} + \begin{bmatrix}\frac{k_{P1}}{L_{F}} & 0\\k_{I1} & 0\\0 & \frac{k_{P1}}{L_{F}}\end{bmatrix}\begin{bmatrix}i_{od}^{*}\\i_{oq}\\i_{oq}\end{bmatrix}$$
(7.63)

This arrangement shows the states related to the d-channel can be solved or controlled independent of the q-channel, hence decoupled control of active and reactive power is achieved.

Since  $w_d$  and  $w_q$  represent the outputs of the current regulator for the *d* and *q* channels, the *d*-*q* components of the reference capacitor voltage for the inner control loops can be estimated, taking into account feed-forward terms as follows:

$$v_{cd}^* = w_d + v_{od} - \omega L_F i_{oq}$$
(7.64)

$$v_{cq}^{*} = w_{q} + v_{oq} + \omega L_{F} i_{od}$$
(7.65)

As stated, the inner current loop is designed based on equations (7.64) and (7.65). The inner voltage controller adjusts the magnitude and phase of the fundamental component of the inverter currents ( $i_{ai}$ ,  $i_{bi}$  and  $i_{ci}$ ) and forces the capacitor voltages to follow the references provided by the current regulator. Therefore, similar procedures as those illustrated previously are required to facilitate controller design.

In (7.64) and (7.65), let  $\lambda_d = i_{id} + \omega C v_{cq} - i_{od}$  and  $\lambda_q = i_{iq} - \omega C v_{cd} - i_{oq}$ , where the variables  $\lambda_d$  and  $\lambda_q$  are obtained from the proportional-integral (PI) controller as follows:

$$\lambda_{d} = k_{p2} \left( v_{cd}^{*} - v_{cd} \right) + k_{I2} \int (v_{cd}^{*} - v_{cd}) dt$$
(7.66)

$$\lambda_{q} = k_{p2} \left( v_{cq}^{*} - v_{cq} \right) + k_{I2} \int (v_{cq}^{*} - v_{cq}) dt$$
(7.67)

where  $k_{P2}$  and  $k_{I2}$  are the proportional and integral gains of the inner capacitor voltage regulator.

After replacing the integral parts of (7.66)and(7.67) with  $\psi_d$  and  $\psi_q$ , the following state space equations, of the form x'(t) = Ax(t) + Bu(t), are obtained:

$$\frac{d}{dt} \begin{bmatrix} v_{cd} \\ \psi_{d} \\ v_{cq} \\ \psi_{q} \end{bmatrix} = \begin{bmatrix} -\frac{k_{P2}}{C_{F}} & \frac{1}{C_{F}} & 0 & 0 \\ -k_{I2} & 0 & 0 & 0 \\ 0 & 0 & -\frac{k_{P2}}{C_{F}} & \frac{1}{C_{F}} \\ 0 & 0 & -k_{I2} & 0 \end{bmatrix} \begin{bmatrix} v_{cd} \\ \psi_{d} \\ v_{cq} \\ \psi_{q} \end{bmatrix} + \begin{bmatrix} \frac{k_{P2}}{C_{F}} & 0 \\ k_{I2} & 0 \\ 0 & \frac{k_{P2}}{C_{F}} \\ 0 & k_{I2} \end{bmatrix} \begin{bmatrix} v_{cd}^{*} \\ v_{cq}^{*} \end{bmatrix}$$
(7.68)

The reference inverter currents are obtained from the inner loop as:

$$i_{id}^* = \psi_d + i_{od} - \omega C_F v_{cq}$$
(7.69)

$$i_{id}^* = \psi_d + i_{od} + \omega C_F v_{cq} \tag{7.70}$$

The transfer functions that govern the grid current and capacitor voltage controllers can be obtained from (7.63) and (7.68). From these transfer functions, the theoretical gains for both controllers are obtained by assigning settling times and damping factors. The final controller gains used are obtained by fine-tuning based on Eigen-value analysis of the overall state matrix in (7.71).

The reference currents in (7.69) and (7.70) must be normalized by dc link current  $I_{dc}$  before passing them to the modulator. With the aid of (7.64) and (7.65), the state space equations in (7.63) can be combined with (7.68) to obtain the state equations for the overall system that account for possible interaction between the current and capacitor voltage loops.

$$\left[ \begin{bmatrix} i_{w_{d}} \\ z_{d} \\ v_{w_{d}} \\ w_{q} \\ z_{q} \\ w_{q} \end{bmatrix}^{z_{q}} \right] = \left[ \begin{bmatrix} -\frac{\left(R_{f} + k_{p1}\right)}{L_{f}} & \frac{1}{L_{f}} & 0 & 0 & 0 & 0 & 0 & 0 \\ -\frac{k_{p1}k_{p2}}{C_{f}} & \frac{k_{p2}}{C_{f}} & \frac{k_{p2}}{C_{f}} & \frac{1}{C_{f}} & -\frac{k_{p2}\omega L_{f}}{C_{f}} & 0 & 0 & 0 \\ -\frac{k_{p1}k_{r2}}{R_{p1}k_{r2}} & k_{r2} & -k_{r2} & 0 & -k_{r2}\omega L_{f} & 0 & 0 & 0 \\ 0 & 0 & 0 & 0 & 0 & -\frac{\left(R_{f} + k_{p1}\right)}{L_{f}} & \frac{1}{L_{f}} & 0 & 0 \\ 0 & 0 & 0 & 0 & 0 & -\frac{k_{p1}k_{p2}}{C_{f}} & \frac{k_{p2}}{C_{f}} & \frac{1}{C_{f}} & -\frac{k_{p2}w L_{f}}{C_{f}} & 0 & 0 \\ \frac{k_{p2}\omega L_{f}}{C_{f}} & 0 & 0 & 0 & -\frac{k_{p1}k_{p2}}{C_{f}} & \frac{k_{p2}}{C_{f}} & \frac{k_{p2}}{C_{f}} & \frac{1}{C_{f}} \\ k_{p2}\omega L_{f} & 0 & 0 & 0 & -k_{p1}k_{p2} & \frac{k_{p2}}{C_{f}} & \frac{1}{C_{f}} \\ \frac{k_{p1}k_{p2}}{C_{f}} & \frac{k_{p2}}{C_{f}} & \frac{k_{p2}}{C_{f}} & \frac{1}{C_{f}} \\ 0 & 0 & -k_{p1}k_{p2} & k_{p2} & -k_{p2} & 1 \\ k_{p2}w L_{f} & 0 & 0 & 0 \\ \frac{k_{p2}k_{p2}}{C_{f}} & \frac{k_{p2}}{C_{f}} & \frac{k_{p2}}{C_{f}} & \frac{1}{C_{f}} \\ 0 & 0 & \frac{k_{p1}k_{p2}}{R_{f}} & \frac{k_{p2}}{C_{f}} & \frac{k_{p2}}{C_{f}} & \frac{1}{C_{f}} \\ 0 & 0 & \frac{k_{p1}k_{p2}}{R_{f}} & k_{p2} & \frac{k_{p2}}{R_{f}} & 0 \\ 0 & 0 & \frac{k_{p1}k_{p2}}{R_{f}} & \frac{k_{p2}}{R_{f}} & 0 & 0 \\ 0 & 0 & \frac{k_{p1}k_{p2}}{R_{f}} & \frac{k_{p2}}{R_{f}} & \frac{k_{p2}}{R_{f}} \\ 0 & 0 & \frac{k_{p1}k_{p2}}{R_{f}} & \frac{k_{p2}}{R_{f}} & \frac{k_{p2}}{R_{f}} \\ 0 & 0 & \frac{k_{p1}k_{p2}}{R_{f}} & \frac{k_{p2}}{R_{f}} & \frac{k_{p2}}{R_{f}} \\ 0 & 0 & \frac{k_{p1}k_{p2}}{R_{f}} & \frac{k_{p2}}{R_{f}} & \frac{k_{p2}}{R_{f}} \\ 0 & 0 & \frac{k_{p1}k_{p2}}{R_{f}} & \frac{k_{p2}}{R_{f}} \\ 0 & 0 & \frac{k_{p1}k_{p2}}{R_{f}} & \frac{k_{p2}}{R_{f}} \\ 0 & 0 & \frac{k_{p1}k_{p2}}{R_{f}} & \frac{k_{p2}}{R_{f}} \\ \end{bmatrix} \right]$$

(7.71)

The relationship that determines the ability of a CSI to exchange active and reactive power with the grid at voltage level  $V_{oabc}$  can be expressed as:

$$\left[\frac{P}{v_{od}} - \omega C_F v_{cq}\right]^2 + \left[\frac{Q}{v_{od}} - \omega C_F v_{cd}\right]^2 = i_{id}^2 + i_{iq}^2$$
(7.72)

Given  $i_{id}^2 + i_{iq}^2 = m^2 i_{dc}^2$ , expression (7.72) can be rewritten as:

$$\left[\frac{P}{v_{od}} - \omega C_F v_{cq}\right]^2 + \left[\frac{Q}{v_{od}} - \omega C_F v_{cd}\right]^2 \le m_{\max}^2 i_{dc}^2$$
(7.73)

contribution to the fundamental current which may affect the dc current utilization. Inequality (7.73) shows that the ability of a CSI to exchange active and reactive powers with the grid is limited by the filter capacitance  $C_F$ , particularly when the inverter is sourcing lagging *VAr* (*Q* is negative). From (7.73) as  $V_{cd} >> V_{cq}$ , lagging reactive power tends to increase the modulation, and hence reduces the available modulation index margin for active power control. Operation with leading reactive power reduces the modulation index, and hence increases the margin for power control, for a fixed dc link current  $I_{dc}$ . Therefore, the relationship between *VAr* generation and modulation in a CSI is the opposite to that for a VSI. The equivalent relationship to (7.73) for a VSI is:

$$\left[\frac{\omega LP}{V_{do}}\right]^2 + \left[\frac{\omega LQ}{V_{do}} + V_{do}\right]^2 \le \frac{1}{4}m_{\max}^2 V_{dc}^2$$
(7.74)

where L is the total inductance of the phase interfacing reactors and transformer, neglecting resistance,  $V_{dc}$  is the VSI dc link voltage and P and Q are active and reactive power respectively.

This discussion has shown the CSI reactive power capabilities, and its possibilities for grid interfacing of renewable power where the inverter needs to meet strict requirements regarding sourcing of leading *VAr* and during low-voltage ride-through (LV-RT) conditions when attempting to support the grid voltage. Inequality (7.73) is useful for selecting the minimum dc link current required by the CSI in order to be able to exchange rated power at rated ac voltage for a given filter capacitance. The filter capacitor must be sized taking into account several factors: it must be sufficient for harmonic attenuation; a cut-off frequency of less than 200Hz to avoid control system conflict and LC filter resonance excitation due to the presence of low-order harmonics such as the 5<sup>th</sup> and 7<sup>th</sup>; and it must make a minimum

# 7.4. Simulation results

To further validate the proposed DRSPWM with a multi-loop control system, simulation results for the exchange of medium active and reactive powers are presented in this section. Table 0.2 lists the parameters used for the grid connected CSI.

Parameters		Values	SI units
Maximum dc link current	I <sub>dc</sub>	60	А
Switching frequency	$f_s$	2.1	kHz
Two-level converter rating	S	150	kVA
Active power capability	Р	120	kW
Reactive power capability	Q	90	kVAr
Line-to-line ac voltage	V <sub>ac</sub>	415	V
Filter capacitor	$C_F$	100	μF
Filter inductor	$L_F$	3	mH
Transformer turn ratio	N	1:1	
Current controller proportional gain	$k_{PI}$	25	Ω
Current controller integral gain	<i>k</i> <sub>II</sub>	3200	$\Omega s^{-1}$
Capacitor voltage controller proportional gain	$k_{P2}$	1	$\Omega^{-1}$
Capacitor voltage integral gain	<i>k</i> <sub><i>I</i>2</sub>	80	$\Omega^{-1}s^{-1}$

Table 0.2: Parameters used in simulation for grid connected current source inverter

# 7.4.1. Unity power factor application

Reference active current,  $I_{dref}$  is set to 50A while reactive current reference,  $I_{qref}$  is 0A to achieve unity power factor. Figure 0.11(a) shows that  $I_d$  and  $I_q$  are able to track their references when the CSI is operated at rated power. Figure 0.11(b) show the performance of capacitor voltage used in the inner control loop regulator. The inverter operates at full medium active power as in Figure 0.11(c) with the exchange at 415 $V_{ac}$  grid voltage, shown in Figure 0.11(d). Output current is shown in Figure 0.11(e) while Figure 0.11(f) shows that the normalized grid voltage is in phase with the current in this unity power factor application.



Figure 0.11: CSI simulation results operating with the proposed DRSPWM with unity power factor.

# 7.4.2. Lagging power factor

In a lagging power factor application, the active current  $I_d$  is remains at 50A while reactive current is changed to 20A as shown in Figure 0.12(a). The CSI operates at its active rated capability, 120kW, while the reactive power reaches approximately half its rating, -50kVAr, as illustrate in Figure 0.12(b). The output current and grid voltage for a lagging power factor are seen in Figure 0.12(c) and (d) respectively. Figure 0.12(e) shows the current waveform lags the voltage.



Figure 0.12: CSI simulation waveforms operating with the proposed DRSPWM with a lagging power factor.

#### 7.4.3. Leading power factor

The reference reactive current,  $I_{qref}$  is set to -20A while  $I_{dref}$  is maintained at 50A to achieve a leading power factor. Figure 0.13(a) show the output current,  $I_d$  and  $I_q$  are able to track their references, producing medium active and reactive powers of 120kW and 50kVAr respectively, see Figure 0.13(b). The output current and grid voltage are shown in Figure 0.13(c) and (d) respectively. Figure 0.13(e) shows the current leads the voltage by 90°. The modulation index for lagging power factor is shown in Figure 0.13(f).



Figure 0.13: CSI simulation results operating with the proposed DRSPWM with a leading power factor.

#### 7.4.4. Zero power factor

In a zero power factor application, the inverter supplies reactive power and the active power is set to zero. The reactive current component,  $I_q$  is 40A and  $I_d$  is zero as shown in Figure 0.14(a), to produce rated reactive power of 90kVAr as shown in Figure 0.14(b). The output current and grid voltage are shown in Figure 0.14(c) and (d) respectively. The current and voltage phase conform with the conditions for zero power factor, as shown in Figure 0.14(e). The modulation index is maintained approximately 0.87 for this application, as shown in Figure 0.14(f).



Figure 0.14:CSI simulation results operating with the proposed DRSPWM at zero power factor.

# 7.5. Experimental results of the proposed DRSPWM in a grid connected CSI

To investigate the viability of the proposed DRSPWM for a grid connected CSI, this section presents experimental results for a three-phase CSI connected to the grid, controlled using DRSPWM in conjunction with the control systems depicted in Figure 0.10. The experimental system parameters used are listed in Table 0.3. The hardware configuration is shown in Figure 0.15. The ac power filter parameters,  $C_F$  and  $L_F$  are selected so as to place the filter cut-off frequency at 200Hz, for reasons previously stated. With  $C_F << 200 \mu$ F, the filter corner frequency is close to the control system bandwidth, resulting in system instability due to conflict between the filter and the control system. Also there is potential for low-order harmonics, such asthe 5th and 7<sup>th</sup>, to excite LC filter resonance.

Parameters		Values	SI units
Maximum dc link current	I <sub>dc</sub>	8	А
DC link inductor	$L_{dc}$	10	mH
Line-to-line ac side voltage	$V_{ac(rms)}$	$\sqrt{3} \times 30$	V
Switching frequency	$f_s$	4	kHz
Filter capacitor	$C_F$	200	μF
Filter inductor	$L_F$	3	mH
Transformer turn ratio	Ν	1:1	
Current controller proportional gain	$k_{PI}$	0.065	Ω
Current controller integral gain	$k_{II}$	0.22	$\Omega s^{-1}$
Capacitor voltage controller proportional gain	$k_{P2}$	0.015	$\Omega^{-1}$
Capacitor voltage integral gain	<i>k</i> <sub>12</sub>	0.02	$\Omega^{-1}s^{-1}$

Table 0.3: Parameters used in the experiments

The results presented for the grid connected CSI are obtained when the inverter is fed from a dc source that mimics an ideal current source by varying the input dc voltage to maintain a constant dc link current of 8A. The viability of the proposed DRSPWM is investigated when the grid connected CSI is operated under various conditions.



(a)



(b)

Figure 0.15: (a) Hardware diagram and (b) test-rig photograph.

#### 7.5.1. Unity power factor application

Figure 0.16 shows the results when the CSI injects active power into the grid at near unity power factor, with  $i_{od}^* = 5A$  and  $i_{oq}^* = 0$ . Figure 0.16(a) shows the current controller successfully forces the active and reactive power current components  $i_{od}$  and  $i_{oq}$  to follow their references. Figure 0.16(b) also shows that the *d*-*q* components of the filter capacitor voltages ( $v_{cd}$  and  $v_{cq}$ ) follow their references set by the current controller, and confirms the assumption made regarding  $v_{cq} < v_{cd}$ .

Figure 0.16(c) displays the actual active and reactive powers exchanged with the grid, and shows zero reactive power as expected. Figure 0.16(d) shows CSI voltage waveforms presented to the interfacing transformer that is used to provide galvanic isolation and voltage matching. The current waveform injected into the grid, monitored at the primary side of the transformer, is shown in Figure 0.16(e). Figure 0.16(f) confirms the phase voltage and current are in phase. Figure 0.16(g) shows the modulation index when the CSI is operated under the conditions stated. There is a small modulation index margin left that could be utilized for active or reactive power generation.



Figure 0.16:Waveforms of CSI operating with the proposed DRSPWM with near unity power factor  $(I_{dc}=8A, i_{od}=5A, i_{oq}=0, V_{ac}=30V \text{ per phase}).$ 

#### 7.5.2. Lagging power factor

The experimental results in Figure 0.17 demonstrate successful CSI operation with the proposed DRSPWM when injecting active power and lagging reactive power at  $i_{od}$ =4A and  $i_{oq}$ =1A. Figure 0.17(a) shows the actual current tracks the reference current, and Figure 0.17(b) shows the active and reactive powers exchanged with the grid. Figure 0.17(c) and (d) show the voltage and current waveforms the CSI presents to the grid. Figure 0.17(e) shows phase voltage and current waveforms, again confirming that the phase current lags the voltage.



(e) Phase voltage and current waveforms

Figure 0.17: Waveforms demonstrating CSI operation with the proposed DRSPWM when delivering active power at lagging VAr ( $I_{dc}$ =8A,  $i_{od}$ =4A and  $i_{oq}$ =1A).

#### 7.5.2. Leading power factor

Figure 0.18 presents results when the CSI delivers active power with leading reactive power. Figure 0.18(a) shows the active and reactive power current components injected into the grid, confirming that the *d-q* currents follow their references, with  $i_{oq}$  negative for leading reactive power. Figure 0.18(b) shows the active and reactive powers injected into the grid at  $i_{od}=5A$  and  $i_{oq}=-1A$ . Figure 0.18(c) and (d) show current and voltage waveforms as the CSI delivers active power at leading reactive power. Figure 0.18(e) shows that the modulation index reduces when active power is delivered into grid with leading reactive power and the same magnitude of  $i_{od}=5A$ . This is shown in equation (7.25) and articulated in the previous sections. Figure 0.18(f) further confirms leading reactive power generation, as the phase current leads the voltage.



(e) Modulation index



Figure 0.18: Waveforms demonstrating CSI operation with the proposed DRSPWM when delivering active power into grid with leading reactive power.

### 7.5.4. Zero power factor

Figure 0.19 present results when the DRSPWM controlled CSI operates as a typical static synchronous compensator (STATCOM) with  $i_{od}=0$  and  $i_{oq}=2A$ . Figure 0.19(a) and (b) demonstrate CSI operation at zero power with zero active power, as seen in the *d-q* current components and power plots. Figure 0.19(c) and (d) show that the current and voltage waveforms presented to grid at the transformer input remain sinusoidal. Figure 0.19(e) shows that the phase current lags the voltage by 90°. The proposed DRSPWM CSI is able to operate successfully over the entire operating range, including zero power factor. Based on the results presented in Figures 7.16, 7.17, 7.18 and 7.19, it can be concluded that the proposed DRSPWM controlled CSI is viable for a wide range of applications that use closed-loop control.



Figure 0.19: Waveforms demonstrating CSI operation with the proposed DRSPWM and zero power factor (typical reactive power applications).

#### 7.5.5. Black-start capability with the proposed DRSPWM

The UK National Grid, Grid Code Operating Code OC5.5.4 – Black Start Testing section defines a black start as the procedure to recover from a total or partial shutdown of the transmission system which has caused extensive loss of supplies. This entails isolated power stations being started individually and gradually being reconnected in order to form an interconnected system again [7.25]. During this situation, operation of a grid connected CSI is similar to an islanding mode where the CSI is voltage mode controlled. The inverter must be able to set the ac voltage for the grid and local load, with excellent harmonic rejection, as well as promoting a low-loss system.

In [7.26], twelve-step PWM is introduced where the switching devices of the CSI operates at half the carrier frequency. This is an indication of low switching losses and offers the same feature as the new direct regular sampled PWM technique. However, the grid voltages are used as reference modulating signals that are compared with two saw-tooth unipolar carriers to generate the gating signals for the CSI switches. This compromises the basic features that make the PWM CSI competitive with respect to the VSI, such as black-start capability, and operation independent of the network short-circuit ratio. In other words, such a scheme cannot be used to power a dead ac network that has no live ac voltage.

Taking this aspect into consideration, the proposed new direct regular sampled PWM modulation technique does not use any carrier signals or reference voltages. Instead it divides the fundamental period into six sectors (60°), and uses reference currents to determine the dwell times of the converter switches, assuming that within each switching cycle ( $T_s$ ) the dwell time of the active state is proportional to the reference current magnitude. A CSI with the proposed modulation technique is able to operate without any abnormality during a black-start situation.

The experimental results in Figure 0.20 confirmed this feature by displaying the voltage and current waveforms, starting from 0A to a specified current level.



(b) Current (also magnified 13.8s<t<14s)during black-start

Figure 0.20: Experimental results confirming the capability of the proposed DRSPWM during black-start.

# 7.6. Typical application of a CSI for grid connection of PV arrays

The CSI is utilised to track the PV maximum power point and to interface the PV system to the grid. In order to achieve these requirements, three control loops are employed, namely dc-link voltage control, ac current control, and capacitor voltage control, as in Figure 0.21. The inverter dc link voltage is adjusted as the weather changes to ensure maximum power extraction all the times. This has been simulated with the aid of the lookup table that contains a pre-defined PV *I-V* characteristic. Weather condition variations are simulated by variable current from the PV arrays, which is assumed as the input signal to the lookup table.



Figure 0.21: Test system of a grid-connected CSI for photovoltaic application.

Matlab/Simulink simulations demonstrate the performance of a single-stage gridconnected CSI when tracking the maximum power point and delivering active and reactive powers into the ac grid. For performance demonstration of the grid-connected CSI under varying weather conditions, the test system shown in Figure 0.21is simulated with the following parameters: the PV array is modelled as a current source that varies its current injection ( $i_{pv}$ ) with weather, and in this case  $i_{pv}$  is varied from 50A to 60A; dc-link inductor=40mH; converter rated apparent power=100kVA; rated ac grid line voltage=380V; switching frequency=2.1kHz; and an LC-filter with 3mH and 80µF inductance and capacitance respectively.

The simulation results of the CSI under varying weather conditions are shown in Figure 0.22. As PV array output current varies from 50A to 60A, dc-link current varies accordingly as shown in Figure 0.22(a). A new dc-link reference voltage is tracked by the MPPT, and using a typical proportional-integral dc-link voltage regulator, the measured dc-link voltage able to track the reference satisfactory. This is illustrated in Figure 0.22(b). The injected output currents are in phase with the voltage at the point of common coupling as required by the grid code and shown in Figure 0.22(c)and the active and reactive power exchanged with the grid is shown in Figure 0.22(d).

From these results, it can be inferred that PV grid integration using the CSI is satisfactory and can be relatively competitive with VSI PV grid integration in terms of simpler ac-filter design, and direct control of injected current (as compared to indirect output current control in a VSI).


Figure 0.22: Simulation results of PV system behaviour in response to a step-change in PV arrays output current using three-phase current source inverter.

#### 7.7. Summary

This chapter presented the operation and control structures of a CSI in island and gridconnected modes. The control systems used in this chapter, namely proportionalresonance (PR) using phase variables and proportional-integral (PI) in the *dq*-frame, have been discussed and evaluated by simulation and experimentation. From these control systems, proportional-integral (PI) control in the *dq*-frame was selected for further investigation of the suitability of the proposed direct regular sample PWM (DRSPWM) technique in a grid-connected mode. The proposed system has been tested over the entire P-Q operating range, including lagging, leading and zero power factor. In addition, the feasibility of the proposed system to operate in a passive network with no generation where the inverter sets the voltage in the same way as a voltage source inverter and synchronous machines during black-start was demonstrated. This is one of the distinctive features of the proposed DRSPWM technique. For completeness, an additional case study was presented, illustrating typical PV CSI inverterimplemention, with maximum power point tracking.

#### References

- [7.1] F. Blaabjerg, R. Teodorescu, M. Liserre, and A. V. Timbus, "Overview of Control and Grid Synchronization for Distributed Power Generation Systems," *Industrial Electronics, IEEE Transactions on*, vol. 53, pp. 1398-1409, 2006.
- [7.2] D. Jingya, D. D. Xu, and W. Bin, "A Novel Control Scheme for Current-Source-Converter-Based PMSG Wind Energy Conversion Systems," *Power Electronics, IEEE Transactions on*, vol. 24, pp. 963-972, 2009.
- [7.3] D. Jingya, X. Dewei, W. Bin, and N. R. Zargari, "Unified DC-Link Current Control for Low-Voltage Ride-Through in Current-Source-Converter-Based Wind Energy Conversion Systems," *Power Electronics, IEEE Transactions on*, vol. 26, pp. 288-297, 2011.
- [7.4] D. Jingya, X. Dewei, and W. Bin, "A Novel Control Scheme for Current-Source-Converter-Based PMSG Wind Energy Conversion Systems," *Power Electronics, IEEE Transactions on*, vol. 24, pp. 963-972, 2009.
- [7.5] P. Tenca, A. A. Rockhill, T. A. Lipo, and P. Tricoli, "Current Source Topology for Wind Turbines With Decreased Mains Current Harmonics, Further Reducible via Functional Minimization," *Power Electronics, IEEE Transactions on*, vol. 23, pp. 1143-1155, 2008.
- [7.6] W. Jiacheng, D. Jingya, W. Bin, X. Dewei, and N. R. Zargari, "Megawatt wind energy conversion system with diode rectifier and multilevel current source inverter," in *Energy Conversion Congress and Exposition (ECCE)*, 2011 IEEE, 2011, pp. 871-876.
- [7.7] P. P. Dash and M. Kazerani, "A multilevel current-source inverter based gridconnected photovoltaic system," in *North American Power Symposium (NAPS)*, 2011, 2011, pp. 1-6.
- [7.8] M. Salo and H. Tuusa, "vector-controlled PWM current-source-inverter-fed induction motor drive with a new stator current control method," *Industrial Electronics, IEEE Transactions on,* vol. 52, pp. 523-531, 2005.
- [7.9] P. Cancelliere, V. D. Colli, R. Di Stefano, and F. Marignetti, "Modeling and Control of a Zero-Current-Switching DC/AC Current-Source Inverter," *Industrial Electronics, IEEE Transactions on*, vol. 54, pp. 2106-2119, 2007.
- [7.10] W. Bin, J. Pontt, J. Rodriguez, S. Bernet, and S. Kouro, "Current-Source Converter and Cycloconverter Topologies for Industrial Medium-Voltage Drives," *Industrial Electronics, IEEE Transactions on*, vol. 55, pp. 2786-2797, 2008.
- [7.11] R. Gonzalez, J. Lopez, P. Sanchis, and L. Marroyo, "Transformerless Inverter for Single-Phase Photovoltaic Systems," *Power Electronics, IEEE Transactions on*, vol. 22, pp. 693-697, 2007.
- [7.12] S. Jain and V. Agarwal, "A Single-Stage Grid Connected Inverter Topology for Solar PV Systems With Maximum Power Point Tracking," *Power Electronics, IEEE Transactions on*, vol. 22, pp. 1928-1940, 2007.
- [7.13] H. Patel and V. Agarwal, "A Single-Stage Single-Phase Transformer-Less Doubly Grounded Grid-Connected PV Interface," *Energy Conversion, IEEE Transactions on*, vol. 24, pp. 93-101, 2009.
- [7.14] C. Yang and K. M. Smedley, "A cost-effective single-stage inverter with maximum power point tracking," *Power Electronics, IEEE Transactions on*, vol. 19, pp. 1289-1294, 2004.

- [7.15] P. P. Dash and M. Kazerani, "Dynamic Modeling and Performance Analysis of a Grid-Connected Current-Source Inverter-Based Photovoltaic System," *Sustainable Energy, IEEE Transactions on*, vol. 2, pp. 443-450, 2011.
- [7.16] B. Sahan, A. N. Vergara, N. Henze, A. Engler, and P. Zacharias, "A Single-Stage PV Module Integrated Converter Based on a Low-Power Current-Source Inverter," *Industrial Electronics, IEEE Transactions on*, vol. 55, pp. 2602-2609, 2008.
- [7.17] R. T. H. Li, C. Henry Shu-hung, and T. K. M. Chan, "An Active Modulation Technique for Single-Phase Grid-Connected CSI," *Power Electronics, IEEE Transactions on*, vol. 22, pp. 1373-1382, 2007.
- [7.18] B. N. Alajmi, K. H. Ahmed, G. P. Adam, and B. W. Williams, "Single-Phase Single-Stage Transformer less Grid-Connected PV System," *Power Electronics, IEEE Transactions on*, vol. 28, pp. 2664-2676, 2013.
- [7.19] T. Ouchi, H. Fujikawa, S. Masukawa, and S. Iida, "A control scheme for a threephase current source inverter in utility interactive photovoltaic system," *Electrical Engineering in Japan*, vol. 135, pp. 43-55, 2001.
- [7.20] C. Yang and K. Smedley, "Three-Phase Boost-Type Grid-Connected Inverter," *Power Electronics, IEEE Transactions on*, vol. 23, pp. 2301-2309, 2008.
- [7.21] "IEEE Recommended Practice for Utility Interface of Photovoltaic (PV) Systems," *IEEE Std 929-2000*, p. i, 2000.
- [7.22] F. Blaabjerg, C. Zhe, and S. B. Kjaer, "Power electronics as efficient interface in dispersed power generation systems," *Power Electronics, IEEE Transactions on*, vol. 19, pp. 1184-1194, 2004.
- [7.23] J. M. Carrasco, L. G. Franquelo, J. T. Bialasiewicz, E. Galvan, R. C. P. Guisado, M. A. M. Prats, J. I. Leon, and N. Moreno-Alfonso, "Power-Electronic Systems for the Grid Integration of Renewable Energy Sources: A Survey," *Industrial Electronics, IEEE Transactions on,* vol. 53, pp. 1002-1016, 2006.
- [7.24] W. Tsai-Fu, C. Chih-Hao, L. Li-Chiun, and K. Chia-Ling, "Power Loss Comparison of Single- and Two-Stage Grid-Connected Photovoltaic Systems," *Energy Conversion, IEEE Transactions on*, vol. 26, pp. 707-715, 2011.
- [7.25] U. N. G. C. plc, "Grid Code Operating Code OC5.5.4-Black Start Testing," in *Issue 5 Revision 0*, ed, August 2012, pp. pp.1-42.
- [7.26] B. Zhihong, Z. Zhongchao, and R. Xinbo, "A Natural Soft-Commutation PWM Scheme for Current Source Converter and Its Logic Implementation," *Industrial Electronics, IEEE Transactions on*, vol. 58, pp. 2772-2779, 2011.

## **Chapter Eight**

## Conclusions

#### 8.0. General conclusion

This thesis discussed grid integration of the medium-scale photovoltaic (PV) system using voltage and current source inverters. Its main focus is on modulation and control strategies that may facilitate low-loss operation of a PV system at low and medium voltage levels. To achieve this objective, six-pulse voltage and current source inverters were used as interfacing options throughout this thesis. Chapter two reviewed the most commonly used voltage source inverter (VSI) topologies that are normally used for interfacing of the PV system to the grid. The suitability of each topology for PV application (power loss, partial shading and number of conversion stages, etc.) was discussed. Based on the discussions presented in chapter two, the three-phase two-level VSI is selected as a suitable interfacing option to be used for the grid integration of the PV system in this thesis. This is due to simplicity of the two-level converter power circuit, and its applicability to medium power applications, with ac rated voltage ranging from 415V to 11kV (line-to-line).

Chapter three briefly reviewed the existing pulse width modulation (PWM) strategies for VSI, and highlighted their suitability for power generation in a harsh grid mode. Based on the discussion on this chapter, traditional space vector modulation that was developed on assumption of balanced three voltage vectors is abandoned in favour of selective harmonic elimination PWM and sinusoidal PWM with 3<sup>rd</sup> harmonic injection. This is because the traditional space vector modulation technique with circular trajectory (and equal side hexagon) is valid only during balanced ac voltage, which cannot be guaranteed during operation at distribution level in grid mode. The use of modified space vector modulation with an elliptical trajectory is possible, but it increases the complexity of the implementation and dwell times calculations. The selected modulation strategies are justified on the grounds that they can match space vector modulation in terms of dc link utilization, and they offer robust performance during unbalanced network operation. As part of thesis contribution, this chapter proposed a new selective harmonic elimination (SHE) strategy that explicits the 3<sup>rd</sup> harmonic magnitude in the phase voltage to preserve the improved dc link utilization over the full modulation index range by avoiding pulse dropping problems. An additional advantage is that it simplified the solution of the SHE equations (it is simple to find an initial set of angles that can be used to calculate the switching angles for the entire modulation index range). These features can be used to simplify pratical realization of SHE at high modulation indices when a large number of harmonics are eliminated.

In chapter four, several of control structures for islanding and grid operation of the voltage source inverter were presented, including their experimental substantiations. It has been demonstrated using a scale-down prototype that the proposed SHE with a synchronous frame PI control structure allows operation of the grid connected VSI, with an effective switching frequency of less than 1kHz. The use of SHE does not compromise any grid connected inverter functionalites that are normally provided when VSI is controlled using carrier based PWM strategies. Furthermore, the output voltage and current waveforms achieved with SHE are within allowable IEEE and IEC standards, and reduced conversion losses. Simulation and experimental results for the grid connected inverter presented in this chapter have shown that SHE with 950Hz effective switching frequency is able to match SPWM with 3<sup>rd</sup> harmonic injection that uses 2kHz switching frequency, in terms of waveforms quality, and control flexibility of the active and reactive powers. In the context of PV, this showed the potential usefulness of the proposed SHE in improving efficiency of PV inverters. Similar implementation can be extended to multilevel inverter topologies. This is, however, out of the scope of this thesis.

Chapter five presented a brief review of the current source inverter (CSI) topologies that suit PV applications, with the emphasis on the six-pulse bridge. Also, it reviewed CSI modulation strategies. Additionally, this chapter proposed a new direct-regular sampled pulse width modulation (DRSPWM) strategy with the advantage of simpler implementation than the existing CSI modulation strategies. Detailed comparison that shows the superiority of the proposed DRSPWM over SPWM, SVM and SHE was presented. This comparison has shown that the proposed DRSPWM has fewer switching transitions than its counterparts, and this indicates lower switching loss. Simulation and experimental results were incorporated to assist this analysis.

Chapter six presented a dc current offset compensation technique that can be used to reduce the overall cost of the solar farm by allowing transformerless inverter operations within farm, and uses a single power transformer at the point of common coupling to the grid. The presented dc compensation technique can be implemented using voltage or current source inverters. However, it has been found that the CSI was inappropriate due to its slow dynamic response, and for this reason only the VSI option was considered in the experimental validation. It was demonstrated that the VSI incorporating with the proposed technique was able to suppress dc current offset to zero in the output phase currents that exist in grid connected parallel inverters within a solar farm before entering the main interfacing transformer to the ac grid. Hence, the risk of transformer saturation can be avoided. Furthermore, this permits transformerless inverter operation within the farm.

Grid integration of a current source inverter using the proposed modulation strategy and suitable control structures was discussed in chapter seven. It has been demonstrated that the grid connected CSI with the proposed DRSPWM was able to provide all the flexibility needed for grid operation, including provision of leading and lagging reactive power, and zero power factor. Additionally, the viability of the proposed DRSPWM for islanding mode was demonstrated using simulation and by experiment, including black start operation. From this chapter, it is possible to conclude that when a current source inverter is correctly controlled, it has the potential to match the performance of the voltage source inverter in a wide range of grid operations.

## 8.1. Author's contributions

Main contributions of this thesis are:

- (a) A new implementation of selective harmonic elimination for voltage source inverters that exploits 3<sup>rd</sup> harmonic injection is proposed that has the following advantages:
  - Switching losses are suppressed by adopting a switching frequency less than 1kHz.
  - Improved dc-link voltage utilization and avoidance of pulse dropping at low and high modulation indices (injection of the 3<sup>rd</sup> harmonic in the phase voltage does not produce narrow pulses in conventional SHE implementation. Instead it redistributes the pulse evenly over quarter fundamental cycle). This is achieved despite the use of an offline look-up table.
  - Simplified solutions of the SHE nonlinear transcendental equations, and in some cases production of a unique solution with extended modulation index range for both arrangements where the phase voltage starts for  $-\frac{1}{2}V_{dc}$  or  $\frac{1}{2}V_{dc}$ .
- (b) The proposed new direct-regular sampled pulse width modulation (DRSPWM) strategy for the current source inverter offers the following advantages:
  - Reduces complexity of CSI gate signal generation by elimination of logic mapping circuitry, hence implicit pulse pattern generation of the carrier based modulation is avoided.
  - Offers the same degree of freedom, flexibility and dc link utilization as space vector modulation, while expecting to maintain the robustness of the carrier based SPWM during unbalanced operation in a grid connected environment.
  - Some of the proposed switching sequences in DRSPWM constitute half the switching frequency per device compared to carrier-based PWM, thus lower switching losses are anticipated.
  - Feasible to operate in a passive network with no generation where the inverter sets the voltage like a voltage source inverter and synchronous machines during black-start.

- (c) The proposed technique for suppression of dc current offset injection into ac grids is applicable to medium-scale solar farms or any group of grid connected inverters confined within relative small area. The proposed technique offers the following advantages:
  - Does not rely on high-precision current measurement devices and does not use any form of complex transformer as discussed in the open literature.
  - At least one inverter must be designated to perform dc offset compensation besides its basic function of active power interfacing to the ac grid. This inverter is required to compensate for all dc offsets that may exist in the total phase current before reaching the main interfacing transformer. The risk of transformer saturation is therefore avoided.
  - Allows transformerless operation of all inverters within the solar farm, and uses one interfacing transformer at the point of common coupling to the grid to reduce the capital cost of a medium-scale solar farm.

## 8.2. Recommendations for future research

- Investigate the viability of the proposed DRSPWM during low-voltage ridethrough and unbalanced operation of the grid connected current source inverters.
- Extension of the DRSPWM strategy to multilevel current source inverters.
- Investigate the possibility of extension of the modified SHE with 3<sup>rd</sup> harmonic injection to multilevel voltage and current source inverters.

## Appendix A

## **Real-time dSPACE processor board**

## A.1. dSPACEDS1006 DSP processor board

The DS1006 uses a single-core AMD Opteron<sup>TM</sup> processor as the real-time processor (RTP) to process real-time models. The RTP calculates real-time models and accesses the I/O boards via the PHS bus. The key features of DS1006 board are:

- X86 processor technology
- Fully programmable from Simulink
- High-speed connection to all dSPACE I/O boards via PHS bus
- Multiprocessor system of several DS1006 processor boards via fiber-optic connection (gigalinks).

Figure A.1 shows the DS1006 board that is used to interface and communicate between I/O boards and the MATLAB/SIMULINK simulation environment. The DS1006 processor board can be programmed via a real-time interface (RTI) in Simulink. Other useful features of this board in the Simulink environment are it is easy to access to the entire range of I/O boards and to alter simulation parameters such as solver options or simulation time as well as to generate C-codes.



Figure A.1: DS1006 real-time processor board

## A.2. I/O board

The I/O boards that are designed in dSPACE are DS5101 Digital Waveform Output (DWO) and DS2002 ADC boards. The DS5101 Digital Waveform Output board is designated to generate complex, high-speed digital signals at high resolution. This board can generate various TTL pulse patterns on up to 16 channels with a time resolution of 25ns. The limitations of DS5101 are as follows:

- The minimum pulse length is 250ns.
- It is not possible to generate PWM signals or similar signals which are constantly low or high, for example, PWM signals with 0% or 100% duty cycle. There will always be a remaining pulse of at least 250ns length.

The DS2002 board provides an ADC unit featuring 2 parallel A/D converters (ADC1, ADC2) multiplexed to 16 channels each (signals VIN1 ... VIN16 and VIN17 ... VIN32). The A/D converters have the following characteristics:

- 4-, 8-, 12- or 16-bit resolution (selectable for each of the 2 A/D converters individually)
- ±5V or ±10V input voltage range (selectable for each of the 2 A/D converters individually)

Figure A.2 shows the I/O boards that are used for experimentation. Voltage and current feedback signals from transducer boards are connected to the DS2002 ADC channels while the DS5101 board is used for PWM generation foe the voltage and current source inverters.



Figure A.2: I/O DS5101 DWO and DS2002 ADC boards

## A.3. dSPACE Control Desk 3.7

dSPACE Control Desk is an experimental software platform that provides all the functions to control, monitor and automate experiments and makes the development of controllers more effective. Figure A.3 illustrate an example of control desk layout that display the waveforms of controlled variables in real-time. The control variables can be run and stopped in real-time, and saved in comma separated values (.csv) format.



Figure A.3: Example of dSPACE control desk layout

## **Appendix B**

#### Arbitrary custom pulse patterns code generation in dSPACEDS5101

# **B.1.** Custom pulse pattern codes generation in DS5101 Digital Waveform Output

The implementation of a custom digital waveform output (DWO) application starts with writing the source file (.src). The source file is written in a plain ascii text file. Using DWO compiler (DWOCOMP), C-source (.c) and related files such as header file (.h), list file (.lst) and DWO simulator input file (.dat) are generated. The C-file contains the encoded DWO object code and a load routine for downloading the object code to the DS5101 by a master DSP. The header file and channel mask. These definitions allow easy access to individual timing parameters by the master DSP. All results and error messages are reported in a list file. These files are added to custom DWO program to be executed on the DS5101 board. Figure B.1 summarizes data flow compilation of customized codes in the DS5101 platform.



Figure B.1: Data flow compilation of customized codes in DS5101DWO

# **B.2.** Sample of the custom source codes for PWM generation for a current source inverter

A sample of custom source code for pulse width modulation generation for a current source inverter is presented. There are six designated channels, namely channel 1, 3 and 5 for upper switches and channel 2, 4 and 6 for bottom switches. To synchronize the six output signals, a master clock is generated on an additional channel (channel 7). The high and low level durations can be adjusted independently through parameters ta1..ta4, tb1..tb4, or tc1..tc4 for the respective PWM phase. Since the DS5101is unable to generate PWM signals with duty cycle value of 0 and 1, there will be a high pulse of 250ns generated for duty cycle values near or equal to 0. For duty cycle values near or equal to 1, a high pulse of 800ns is generated.

# =======	
unit ticks	# all delays given in 25ns ticks
delay t0 0	# PWM period / 2
delay t0a 0	# (PWM period / 2) - interruptshift
delay t0b 0	# interruptshift
delay ta1 0	# t low/2, phase A
delay ta2 0	# t high, phase A
delay ta3 0	# t high/2, phase /A
delay ta4 0	# t low, phase /A
delay tb1 0	# t low/2, phase B
delay tb2 0	# t high, phase B
delay tb3 0	# t high/2, phase /B
delay tb4 0	$\#$ t_low, phase /B
delay tc1 0	# t low/2, phase C
delay tc2 0	# t high, phase C
delay tc3 0	# t high/2, phase /C
delay tc4 0	# t low, phase /C
	(channels 17)
#	
ch1	# phase A
reset, wait;	# low after reset, wait for trigger
begin:	
syncen,	# use new data from DSP
nop, ta1;	# delay t_low/2
set, ta2;	# set output high, delay t_high
reset, wait,	# set output low, wait for trigger
goto begin;	# loop
ch2	# phase /A
reset, wait;	# high after reset, wait for trigger
begin:	" mgn arter reset, wart for ungger
syncen,	# use new data from DSP
nop, ta3;	# delay t high/2
	······································

# set output low, delay t low set, ta4; reset, wait, # set output high, wait for trigger goto begin; # loop ch3 # phase B # low after reset, wait for trigger reset, wait; begin: syncen, # use new data from DSP # delay t low/2 nop, tb1; # set output high, delay t high set, tb2; # set output low, wait for trigger reset, wait, goto begin; # loop ch4 # phase /B reset, wait; # high after reset, wait for trigger begin: # use new data from DSP syncen, nop, tb3; # delay t high/2 set, tb4; # set output low, delay t low reset, wait, # set output high, wait for trigger goto begin; # loop ch5 # phase C # low after reset, wait for trigger reset, wait; begin: # use new data from DSP syncen, # delay t low/2 nop, tc1; set, tc2; # set output high, delay t high reset, wait, # set output low, wait for trigger goto begin; # loop # phase /C ch6 reset, wait; # high after reset, wait for trigger begin: # use new data from DSP syncen, # delay t\_high/2 nop, tc3; # set output low, delay t low set, tc4; # set output high, wait for trigger reset, wait, goto begin; #loop # master clock ch7 # low after reset reset, t0; begin: # use new data from DSP syncen, # trigger channels 1..6 tr1, tr2, tr3, tr4, tr5, tr6, set, t0a; # set output high, delay t0a phsint, # send interrupt to DSP set, t0b; # set output high, delay t0b reset, t0, # set output low, delay t0 goto begin; # loop updfunc all # generate update function g t0, g t0a, g t0b g ta1, g ta2, g ta3, g ta4, g tb1, g tb2, g tb3, g tb4, g tc1, g tc2 g tc3, g tc4; # for all global delays

readfunc all # generate read function g t0, g t0a, g t0b, g ta1, g ta2, g ta3, g ta4, g tb1, g tb2, g tb3, g tb4, g tc1, g tc2 g tc3, g tc4; # for all global delays

#### B.3. Simulink s-function template for DS5101DWO block

DS5101 RTI blockset provides DS5101DWO block to enable parameter updating of a custom DWO application in Simulink model. A simulink s-function template is serviced by RTI, where DWO C-source file, header file, and a number of parameters must be specified. A s-function template is edited and compiled using MATLAB command mex<file name>.c in the same working directory as C-source, header and other related files.

A sample of s-function template is presented as below.

#-------#

#ifndef MATLAB\_MEX\_FILE
# include <ds5101.h>
# include <ap5101.h>
# include <rti\_msg\_access.h>
# include <rti\_common\_msg.h>
#endif

#define S\_FUNCTION\_LEVEL 2 #define S\_FUNCTION\_NAME pwm3\_CSI #ifndef MATLAB\_MEX\_FILE #include "PWM3\_INTSHIFT.h" #include "PWM3\_INTSHIFT.c"

#define NUM\_DWOVARS 9

#define CHANNELMASK DS5101_PWM3_INTSHIFT_CHANNELS			
#define DS5101_PWM3_INTSHIFT_GW_T0	0x000F0001		
#define DS5101_PWM3_INTSHIFT_GR_T0	0x00010001		
#define DS5101_PWM3_INTSHIFT_GW_T0A	0x000F0002		
#define DS5101_PWM3_INTSHIFT_GR_T0A	0x00010002		
#define DS5101_PWM3_INTSHIFT_GW_T0B	0x000F0003		
#define DS5101_PWM3_INTSHIFT_GR_T0B	0x00010003		
#define DS5101_PWM3_INTSHIFT_GW_TA1	0x000F0004		
#define DS5101_PWM3_INTSHIFT_GR_TA1	0x00010004		
#define DS5101_PWM3_INTSHIFT_GW_TA2	0x000F0005		
#define DS5101_PWM3_INTSHIFT_GR_TA2	0x00010005		
#define DS5101_PWM3_INTSHIFT_GW_TB1	0x000F0006		
#define DS5101_PWM3_INTSHIFT_GR_TB1	0x00010006		
#define DS5101_PWM3_INTSHIFT_GW_TB2	0x000F0007		
#define DS5101_PWM3_INTSHIFT_GR_TB2	0x00010007		
#define DS5101_PWM3_INTSHIFT_GW_TC1	0x000F0008		
#define DS5101_PWM3_INTSHIFT_GR_TC1	0x00010008		
#define DS5101_PWM3_INTSHIFT_GW_TC2	0x000F0009		

#define DS5101\_PWM3\_INTSHIFT\_GR\_TC2 #define DS5101\_PWM3\_INTSHIFT\_CHANNELS 0x00010009 0x0000000F

#define DS5101\_LOAD() \
 ds5101\_pwm3\_intshift\_load(boardBase, CHANNELMASK)

#define DS5101\_INIT\_ALL() /\*\ ds5101\_pwm3\_intshift\_init\_all(boardBase, 0.0, 0.0)\*\

#define DS5101\_UPDATE\_ALL() \ ds5101\_pwm3\_intshift\_update\_all(boardBase, u(0), u(1),u(2),u(3),u(4),u(5),u(6),u(7),u(8))

#endif

## Appendix C

#### List of Figures and Tables

#### **List of Figures**

## **Chapter Two: Voltage Source Inverter Topologies**

Figure 2.1: General block diagram of (a) single-stage arrangement grid connected PV system and (b) two-stage arrangement grid connected PV system.

Figure 2.2: Topology of half-bridge single-phase VSI.

Figure 2.3: Topology of H-bridge single-phase VSI.

Figure 2.4: Topology of three phase two-level VSI.

Figure 2.5: Topology of one-phase of neutral-point clamped inverter.

Figure 2.6: One phase of three-level flying capacitor (also known as capacitor clamped) inverter.

Figure 2.7: Power circuit of one phase leg of nine-level cascaded multilevel inverter.

Figure 2.8: Power circuit of one phase leg of three-level modular inverter.

#### **Chapter Three: Modulation Technique for the Voltage Source Inverter**

Figure 3.1: Topology of three phase two-level voltage source inverter.

Figure 3.2: Modulation of the three phase VSI using SPWM and sample of open-loop experimental waveforms ( $V_{dc}$ =100V,  $m_a$ =0.8 and  $f_{sw}$ =2000Hz)

Figure 3.3: Reference modulating and output voltage waveforms when the VSI is controlled with third harmonic injection PWM.

Figure 3.4: Space vector diagram for a two-level inverter.

Figure 3.5: Line-to-line voltage,  $v_{AB}$  waveform relative to midpoint of the dc bus when the VSI is controlled by SVM.

Figure 3.6:SHE implementation using solution 1 and 2 when  $5^{th}$  and  $7^{th}$  harmonic components are eliminated.

Figure 3.7: Waveform illustrate phase voltage relative to supply mid-point when the 5<sup>th</sup> and 7<sup>th</sup> harmonic are eliminated,  $m_a=1.1$  and 3<sup>rd</sup> harmonic is  $m_3=m/5$  for both SHE solutions.

Figure 3.8: Switching angle trajectories for solution 1 and 2 when 5<sup>th</sup> and 7<sup>th</sup> harmonics are eliminated.

Figure 3.9: Experimental voltage and current waveforms for two-level VSI using SHE solution 1, with and without adjustment of the 3<sup>rd</sup> harmonic component and 5<sup>th</sup>, 7<sup>th</sup>, 11<sup>th</sup> and 13<sup>th</sup> harmonics eliminated  $(m_q=1.1)$ .

Figure 3.10: Experimental waveforms illustrating two-level inverter output voltage normalized by solutions 1 and 2, when the  $5^{th}$  and  $7^{th}$  harmonics are eliminated without the adjustment of  $3^{rd}$  harmonic component.

Figure 3.11: Experimental waveforms illustrating two-level inverter output voltage normalized by  $\frac{1}{2}V_{dc}$  for solutions 1 and 2, when the 5<sup>th</sup> and 7<sup>th</sup> harmonics are eliminated, and including adjustment of the 3<sup>rd</sup> harmonic component.

Figure 3.12: Experimental voltage and current waveforms for two-level VSI using SHE solution 1, with and without adjustment of the  $3^{rd}$  harmonic component and  $5^{th}$ ,  $7^{th}$ ,  $11^{th}$  and  $13^{th}$  harmonics eliminated ( $m_a$ =1.1).

Figure 3.13: (a) Switching angle trajectories with nine notches per quarter fundamental cycle and (b) normalized PWM voltage waveform with reference to midpoint of dc bus.

Figure 3.14: Open-loop experiment validation of VSI when SHE is used as the modulator.

Figure 3.15: Theoretical phase voltage harmonic spectrum of all PWM techniques.

Figure 3.16: Comparison of PWM strategies with varying modulation index for per-unit fundamental (a) pre-filter inverter current and (b) total harmonic distortion (%THD).

#### **Chapter Four: Grid Integration of Voltage Source Inverter**

Figure 4.1: Circuit diagram of VSI in island mode.

Figure 4.2: Schematic diagram of the voltage control loop using a PR controller.

Figure 4.3: Control system of a VSI using PI control in the *dq*-frame.

Figure 4.4: Simulation waveforms illustrating three-phase VSI is controlled by PR controller in island mode

Figure 4.5: Simulation waveforms when three-phase VSI is controlled by synchronous frame PI controller and SPWM as the modulator in island mode.

Figure 4.6:dSPACE 1106 control desk experimental results of the three-phase VSI scaled-prototype connected to balanced loads when controlled using a PR controller in an *abc* frame.

Figure 4.7: dSPACE 1106 control desk experimental results of three-phase VSI scaled-prototype connected to balanced loads when it is controlled using synchronous dq-frame PI controller.

Figure 4.8:Schematic diagram of a grid-connected three-phase VSI.

Figure 4.9: Schematic diagram consists of PR control system in *abc*coordinates that is used to control the output current of grid-connected VSI.

Figure 4.10:Experimental results showing the performance of a VSI injecting current into the grid.

Figure 4.11: Schematic diagram of a grid connected PV inverter with SHE and THIPWM using a dSPACE1106 hardware implementation.

Figure 4.12: Voltage source converter (a) generic phasor diagram and (b) theoretical capability curve.

Figure 4.13: Simulation waveforms demonstrating operation of a high-power medium-voltage grid connected PV-inverter when controlled using SHE-PWM, with a 950Hz switching frequency, 20kV dc link voltage, and connected to a 11kV ac grid (step change in active power).

Figure 4.14: Simulation waveforms demonstrating operation of a high-power medium-voltage grid connected PV-inverter when controlled using THIPWM, with a 2kHz switching frequency, 20kV dc link voltage, and connected to a 11kV ac grid (step change in active power).

Figure 4.15: Simulation waveforms demonstrating operation of grid connected PV-inverter when controlled using SHE-PWM during rated power of 20MW, 0.8 power factor lagging.

Figure 4.16: Simulation waveforms demonstrating operation of grid connected PV-inverter when controlled using THIPWM during rated power of 20MW, 0.8 power factor lagging.

Figure 4.17: Simulation waveforms demonstrating operation of grid connected PV-inverter when controlled using THIPWM during rated power of 15MVAr, zero power factor lagging.

Figure 4.18: Hardware diagram of the test-rig.

Figure 4.19: per-unit scaled-down test rig of a multi-megawatt PV inverter prototype.

Figure 4.20: Waveforms demonstrating the operation of a PV inverter using sinusoidal PWM with triplen harmonic injection, when  $I_d$  is varied from 2A to 6A.

Figure 4.21: Experimental waveforms demonstrating operation of the grid connected inverter using SHE-PWM (step change in active power).

Figure 4.22: Experimental waveforms demonstrating PV inverter using sinusoidal PWM with triplen harmonic injection, when  $I_q$  is varied from 0 to 4A while  $I_d = 2A$ .

Figure 4.23: Experimental waveforms demonstrating operation of a grid connected inverter using SHE-PWM (step change in reactive power). Figure 4.24: Schematic diagram representing the control structure which consists of a dc-link voltage regulator and an output current regulator.

Figure 4.25: Simulation results of PV system behaviour in response to a step-change in PV arrays output current using three-phase voltage source inverter.

Figure 4.26: Simulation results showing the performance of the grid-connected photovoltaic VSI when there is low or no radiation from sun.

#### **Chapter Five: Current Source Inverter for Photovoltaic Application**

Figure 5.1: Topology of a single-phase H-bridge CSI.

Figure 5.2: Topology of the two-level three-phase CSI.

Figure 5.3: Modulation process of the CSI using carrier-based SPWM

Figure 5.4: Sample CSI open-loop simulation waveforms using SPWM

Figure 5.5: (a) Diagram of possible PWM gating signal generation using selective harmonic elimination and (b) typicalinverter current waveform ( $i_i$ ) relative to dc-link current ( $\pm I_{dc}$ ).

Figure 5.6: Diagram of PWM gating signal generation using direct SVM.

Figure 5.7: Synthesis of  $\vec{I}_{ref}$  by  $\vec{I}_1$ ,  $\vec{I}_2$  and  $\vec{I}_0$ .

Figure 5.8: SVM switching sequence of a CSI over a fundamental cycle.

Figure 5.9: Diagram of VSI-SVM with logic mapping circuit used to generate CSI gating signals.

Figure 5.10: Reference modulating signal.

Figure 5.11: Examples of the possible switch sequences in sector I.

Figure 5.12: Pictorial representation of active states interchanged for sequence I.

Figure 5.13: dSPACE and generic block diagrams that summarise digital implementation of the proposed DRSPWM.

Figure 5.14: Illustration of arbitrary pulse generation.

Figure 5.15: Open loop simulation waveforms illustrating the performance of the proposed DRSPWM at  $I_{dc}$ =5A,  $m_a$ =0.8 and a 2.1kHz switching frequency.

Figure 5.16: Experimental waveforms illustrating reference modulating signal, pre-filter inverter current and load current at  $m_a=0.3$  and  $m_a=0.5$  using the proposed DRSPWM.

Figure 5.17: Experimental waveforms illustrating reference modulating signal, pre-filter inverter current and load current at  $m_a=0.8$  and  $m_a=0.95$  using the proposed DRSPWM.

Figure 5.18: Experimental waveforms illustrating the viability of the proposed RSPWM at  $m_a = 0.8$ , 2.1kHz switching frequency, and 70µF filter capacitor.

Figure 5.19: Experimental harmonic spectrum of inverter current based on the proposed sequences I, II and III.

Figure 5.20: Current waveforms for switch S<sub>1</sub> for the proposed DRSPWM technique, CSPWM and SVM, at  $I_{dc}$ =5A,  $m_a$ =0.8 and a 2.1kHz switching frequency.

Figure 5.21: Experimental waveforms illustrating the gating signal for switch  $S_1$  for the proposed DRSPWM technique, CSPWM and SVM at  $I_{dc}$ =5A and  $m_a$ =0.8.

Figure 5.22: Theoretical spectrum of inverter current before ac filters.

Figure 5.23: Comparison of PWM strategies with varying modulation index for (a) per-unit fundamental pre-filter inverter current and (b) total harmonic distortion (%THD), all at 2.1kHz switching frequency.

## Chapter Six: DC Current Offset Compensation Technique for Grid Connected Inverters

Figure 6.1: Blocking capacitor circuits for single-phase grid connected inverters.

Figure 6.2: DC offset current control loop with a voltage transformer (VT).

Figure 6.3: DC compensation technique using low power toroidal transformer as a dc offset sensor.

Figure 6.4: H-bridge inverter using dc-link current sensor to compensate dc current injection into the ac grid.

Figure 6.5: Three-level half bridge inverter using positive and negative dc current sensors to compensate dc current injection to the load.

Figure 6.6: Generic diagram of grid connected inverters with dc offset compensation inverter in a solar farm.

Figure 6.7: Possible VSI and CSI configurations for the proposed dc offset compensation method.

Figure 6.8: Block diagram that represents the proportional resonance controller.

Figure 6.9: Proposed dc current offset compensation control system.

Figure 6.10: Simulation results of configuration I showing inverter 1 is mimicked an inverter that contains dc current offset components in its output current.

Figure 6.11: Simulation results of configuration I showing a VSI as a dc compensation inverter.

Figure 6.12: Simulation results for configuration II showing CSI as a dc compensation inverter.

Figure 6.13: (a) Hardware diagram and (b) test rig of grid-connected inverters in parallel arrangement.

Figure 6.14: Experimental results illustrating the current and voltage waveforms when two gridconnected VSIs exchange active and reactive power with the grid voltage without the introduction of dc current offset. Figure 6.15: Experimental results demonstrating the practicability of the proposed dc offset compensation for grid connected inverters (Component of dc offsets injected relative to output current in each phase: a=0.59A, b=-0.08A and c=-0.8A).

Figure 6.16: Experimental results demonstrating the practicability of the proposed dc offset compensation for grid connected inverters (Component of dc offsets injected relative to the output current in each phase: a=-0.14A, b=0.07A and c=0.4A).

Figure 6.17: Experimental results demonstrating the practicability of the proposed dc offset compensation for grid connected inverters (Component of dc offsets injected relative to the output current in each phase: a=-0.56A, b=0.75A and c=-1A).

Figure 6.18: Experimental results demonstrating the practicability of the proposed dc offset compensation for grid connected inverters (Component of dc offsets injected relative to the output current in each phase: a=-0.56A, b=0.75A and c=-1A).

Figure 6.19: Experimental results demonstrating the practicability of the proposed dc offset compensation for grid connected inverters (Component of dc offsets injected relative to the output current in each phase: a=0.5A, b=-1.2A and c=0.85A).

#### **Chapter Seven: Grid Integration of a Current Source Inverter**

Figure 7.1: Circuit diagram of CSI in island mode.

Figure 7.2: Control diagram of PR voltage controller for current source inverter.

Figure 7.3: CSI control system in island mode using PI control in thedq-frame.

Figure 7.4: Simulation results illustrating a three-phase CSI in island mode using PR voltage-control in the *abc* frame.

Figure 7.5: Simulation results illustrating the three-phase CSI in island mode using *dq*-frame PI voltage-control.

Figure 7.6: CSI experimental results in island mode using PR voltage control in an *abc* frame.

Figure 7.7: CSI experimental results in island mode using PI voltage control in the dq-frame (part a).

Figure 7.8:CSI experimental results in island mode using PI voltage control in the dq-frame (part b).

Figure 7.9: Schematic of grid-connected three-phaseCSI.

Figure 7.10: Block diagram of the control systems of a grid connected CSI.

Figure 7.11: CSI simulation results operating with the proposed DRSPWM with unity power factor.

Figure 7.12: CSI simulation waveforms operating with the proposed DRSPWM with a lagging power factor.

Figure 7.13: CSI simulation results operating with the proposed DRSPWM with a leading power factor.

Figure 7.14: CSI simulation results operating with the proposed DRSPWM at zero power factor.

Figure 7.15: (a) Hardware diagram and (b) test-rig photograph.

Figure 7.16: Waveforms of CSI operating with the proposed DRSPWM with near unity power factor  $(I_{dc}=8A, i_{od}=5A, i_{oq}=0, V_{ac}=30V \text{ per phase}).$ 

Figure 7.17: Waveforms demonstrating CSI operation with the proposed DRSPWM when delivering active power at lagging VAr ( $I_{dc}$ =8A,  $i_{od}$ =4A and  $i_{oq}$ = 1A).

Figure 7.18: Waveforms demonstrating CSI operation with the proposed DRSPWM when delivering active power into grid with leading reactive power.

Figure 7.19: Waveforms demonstrating CSI operation with the proposed DRSPWM and zero power factor (typical reactive power applications).

Figure 7.20: Experimental results confirming the capability of the proposed DRSPWM during black-start.

Figure 7.21: Test system of a grid-connected CSI for photovoltaic application.

Figure 7.22: Simulation results of PV system behaviour in response to a step-change in PV arrays output current using three-phase current source inverter.

## List of Tables

#### **Chapter One: Introduction**

Table 1.1: Total electricity production from renewable sources in 2011.

Table 1.2: Summary of some important issues in IEEE1574 and IEC61727 standards.

#### **Chapter Two: Voltage Source Inverter Topologies**

Table 2.1: Switch states and the outputs.

Table 2.2: Summary of the switch states of a single-phase H-bridge inverter that can be used to generate a three-level output voltage.

Table 2.3: Summary of the switch state combinations and their corresponding output voltage.

Table 2.4: Summary of the switch state combinations and their corresponding output voltage.

Table 2.5: Switch combinations for a three-level modular inverter.

#### **Chapter Three: Modulation Technique for the Voltage Source Inverter**

Table 3.1: Eight possible switch state combinations of a three-phase VSI.

Table 3.2: (a) Harmonic content in voltage for all discussed modulation techniques ( $m_a$ =0.9) and (b) Harmonic content in voltage for all discussed modulation techniques ( $m_a$ =0.5)

Table 3.3: (a) Harmonic content in current for all discussed modulation techniques ( $m_a$ =0.9) and (b) Harmonic content in current for all discussed modulation techniques ( $m_a$ =0.5)

Table 3.4: Comparison summary of all presented modulation techniques.

#### **Chapter Four: Grid Integration of Voltage Source Inverter**

Table 4.1: System parameters.

Table 4.2: Simulation parameters of a multi-megawatt PV inverter.

Table 4.3: Converter switching device data used for switching loss estimation, including turn-on, turn-off, and diode recovery losses, based on IGBT MG1200FXF1US53.

Table 4.4: Summary of switching losses with THI-PWM and SHE modulation strategies for a 25MVA PV-inverter.

Table 4.5: Parameters of pu scaled-down test rig for prototype of megawatt PV inverter.

Table 4.6: Harmonic content in the experimental current for SHE-PWM and THI-PWM inverters.

Table 4.7: Harmonic content in the experimental voltage for SHE-PWM and THI-PWM inverters.

Table 4.8: Converter switching device data used for switching loss estimation, including turn-on, turn-off, and diode recovery losses, based on the IGBTIRG4PH50KDbF.

Table 4.9: Summary of switching losses with THI-PWM and SHE modulation strategies for a 1200VA PV-inverter.

#### **Chapter Five: Current Source Inverter for Photovoltaic Application**

Table 5.1: Switch states and space vector.

Table 5.2: Selection of zero vector sequence in sector I to VI.

Table 5.3: Dwell times calculations for all sequences and sections identification.

Table 5.4: Summary of the realizable switch sequences for the CSI.

Table 5.5: Example calculations of the gating pulses delay-time and on-time in the first sector

## Chapter Six: DC Current Offset Compensation Technique for Grid Connected Inverters

Table 6.1: Collection of general guidelines that impose limits on the allowable level of dc current injection into the grid by photovoltaic inverter systems.

Table 6.2: Parameters used in the simulations.

Table 6.3: Experimental Parameters.

#### **Chapter Seven: Grid Integration of a Current Source Inverter**

Table 7.1: Parameters used in simulation and experimental

Table 7.2: Parameters used in simulation for grid connected current source inverter.

Table 7.3: Parameters used in the experiments

## **Appendix D**

#### Summary of relevant published work by the author

## S.A.Azmi, G.P.Adam, K.H.Ahmed, S.J.Finney, and B.W.Williams, "Grid Interfacing of Multimegawatt Photovoltaic Inverters," *Power Electronics, IEEE Transactions on*, vol. 28, pp. 2770-2784, 2013.

*Abstract*— This paper investigates the suitability of selective harmonic elimination (SHE) for low loss multi-megawatt grid connected photovoltaic (PV) inverters. The proposed system is able to meet utilities regulations, IEEE and IEC standards. In an attempt substantiate the potential superiority of SHE over carrier based or space vector pulse width modulation (PWM), this paper demonstrates SHE may allow grid connected PV inverters to be controlled using a switching frequency of less than 1kHz while the inverter is still able to provide necessary operation features such as independent control of active and reactive powers and operation control simplicity. For system validation, experimental results with SHE are compared to the case when the inverter is controlled using third harmonic injection PWM, with a 2kHz switching frequency. Furthermore, the paper proposes a new implementation technique for SHE that utilizes the 3<sup>rd</sup> harmonics to spread the switching angles over 90° instead of being located in a narrow range as generated when using conventional techniques, along with increases in modulation index. The advantages of the proposed technique include simplicity in implementation and flexibility in PWM waveforms. Simulation and experimentation demonstrate agreement, which validates the practicability of the proposed system.

## [2] S.A.Azmi, G.P.Adam and B.W.Williams, "New Direct Regular Sampled PWM Applicable for Grid and Islanding Operation of Current Source Inverters," *Power Electronics, IET Transaction on*, pp.1-17,2013.(doi:10.1049/ietpel.2013.0235)

*Abstract*—This paper presents new direct regular sampled pulse width modulation for grid connected three-phase current source inverters. The theoretical basis of the presented modulation strategy is described in detail and includes dwell time calculations and switching sequence selection. The modulation strategy is simple and suited for digital implementation, and has the flexibility and degree of freedoms of space vector modulation. This is because some switch sequences ensure a minimum number of switching transitions per fundamental cycle, unlike other methods presented in the literature. It is therefore applicable for high-power medium-voltage applications. The validity of the presented direct regular sampled pulse width modulation is confirmed using simulation and experimentation of a current source inverter, in open and closed loop operation.

# [3] S.A.Azmi, K.H.Ahmed, S.J.Finney, and B.W.Williams, "Comparative Analysis between Voltage and Current Source Inverters in Grid-Connected Application", *Renewable Power Generation (RPG2011), IET International Conference on*, pp1-6, 2011.

*Abstract*—The voltage source inverter (VSI) is mainly used for grid interfacing of distributed generation systems. In order to boost the voltage of a renewable energy source to the required dc voltage level, a dc-dc converter is necessary even though the cost and complexity of the system are increased. Thus, in improving the cost and life expectancy of the power electronic interface, a current source inverter (CSI) is an alternative which offers short-circuit protection capabilities, implicit voltage boosting and a simple acside filter structure. However, it has attracted less research attention since it is purported to have drawbacks in term of high conduction losses and inefficient inductive energy storage on its dc-side. With reference to advantages and disadvantages of both inverter types, this paper presents a comprehensive comparative analysis with respect to the topological and operational features of the inverters. The comparison includes the power circuit, semiconductor constraints, cost, dc input filtering, steady-state, transient performance and power semiconductor losses. The analytical expectations of the systems are verified using MATLAB/ Simulink software package.

# [4] S.A.Azmi, G.P.Adam and B.W.Williams, "New Modulation Strategy for Three-Phase Current Source Inverters", *Power Engineering, Energy and Electrical Drives (POWERENG 2013), 4<sup>th</sup> IEEE International Conference on*, pp1110-1115, 2013.

*Abstract*— This paper proposes a new regular sampled pulse width modulation (PWM) for three-phase current source inverters. The theoretical basis of the proposed modulation strategy is described, including dwell time calculations and switching sequence selection. This strategy is well suited for digital implementation, and enjoys all the flexibilities and degree of freedoms as in space vector modulation. In addition, it ensures a minimum number of switching transitions per fundamental cycle, unlike other PWM methods. It is therefore promising for high-power medium-voltage applications. The validity of the proposed modulation strategy has been confirmed using simulation and experimentation. To furthermore substantiate the superiority of the proposed modulation strategy against other PWM methods, its performance is compared to sinusoidal PWM, selective harmonic elimination, and space vector modulation.